UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

	_		1.1.				
	For	the quarterly period en	ded June 30, 2023				
TRANSITION REPORT	PURSUANT TO	SECTION 13 OR 1	5(d) OF THE SECURITIES EXCHANGE	GE ACT OF 1934			
	For the	transition period from Commission File Numb					
I			LDINGS, INC.				
	(Exact	name of registrant as sp	ecified in its charter)				
	Delaw	35-2318913					
(State or	other jurisdiction of in	(IRS Employer Identifica	ntion No.)				
2	01 Burlington Road Bedford	01730	01730				
	(Address of principal	(Zip Code)					
		(978) 671-					
	(Regi	istrant's telephone number,	,				
	(F f	Not Applica	ble cal year, if changed since last report				
	(Former name, for	ilei address and former fisc					
es registered pursuant to Section 12(b) of	the Act:						
Title of each class		Trading Symbol(s)	Name of each exchange on which registered				
Common stock, par value \$0.01 pe	r share	LNTH	The Nasdaq Global Market				
orter period that the registrant was require	d to file such reports), and submitted electronically	d (2) has been subject to such	or 15(d) of the Securities Exchange Act of 1934 during the filing requirements for the past 90 days. Yes ☑ No equired to be submitted pursuant to Rule 405 of Regulations the files). Yes ☑ No □				
	large accelerated filer, a	n accelerated filer, a non-acce	elerated filer, smaller reporting company, or an emerging growth company" in Rule 12b-2 of the Exchange Act.	growth company. See the			
			Accelerated filer				
	7						
ons of "large accelerated filer," "accelerate			Smaller reporting company				

Indicate by check mark whether the registrant is a shell company (as defined by Rule 12b-2 of the Act) Yes □ No ☑
The registrant had 68,430,697 shares of common stock, \$0.01 par value, outstanding as of July 28, 2023.

LANTHEUS HOLDINGS, INC. TABLE OF CONTENTS

		Page
PART I. FINAN	NCIAL INFORMATION	
Item 1.	<u>Financial Statements (Unaudited)</u>	
	Condensed Consolidated Balance Sheets	1
	Condensed Consolidated Statements of Operations	2
	Condensed Consolidated Statements of Comprehensive Income	3
	Condensed Consolidated Statements of Changes in Stockholders' Equity	4
	Condensed Consolidated Statements of Cash Flows	6
	Notes to Condensed Consolidated Financial Statements	8
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	22
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	34
Item 4.	Controls and Procedures	34
PART II. OTHI	ER INFORMATION	
Item 1.	<u>Legal Proceedings</u>	36
Item 1A.	Risk Factors	37
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	38
Item 3.	<u>Defaults Upon Senior Securities</u>	38
Item 4.	Mine Safety Disclosures	38
Item 5.	Other Information	38
Item 6.	<u>Exhibits</u>	40
SIGNATURES		41

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements (Unaudited)

Lantheus Holdings, Inc. Condensed Consolidated Balance Sheets

(Unaudited) (in thousands, except par value)

	June 30, 2023]	December 31, 2022
Assets			
Current assets			
Cash and cash equivalents	\$ 414,076	\$	415,652
Accounts receivable, net	256,277		213,397
Inventory	51,801		35,475
Other current assets	22,906		13,092
Assets held for sale	7,159		_
Total current assets	752,219		677,616
Property, plant and equipment, net	129,981		122,166
Intangibles, net	221,004		315,285
Goodwill	61,189		61,189
Deferred tax assets, net	134,201		110,647
Other long-term assets	39,126		34,355
Total assets	\$ 1,337,720	\$	1,321,258
Liabilities and stockholders' equity		_	
Current liabilities			
Current portion of long-term debt and other borrowings	\$ 384	\$	354
Accounts payable	38,861		20,563
Short-term contingent liability	_		99,700
Accrued expenses and other liabilities	114,798		127,084
Total current liabilities	154,043		247,701
Asset retirement obligations	22,729		22,543
Long-term debt, net and other borrowings	559,235		557,712
Other long-term liabilities	49,068		46,155
Total liabilities	785,075		874,111
Commitments and contingencies (See Note 18)			
Stockholders' equity			
Preferred stock (\$0.01 par value, 25,000 shares authorized; no shares issued and outstanding)	_		_
Common stock ($\$0.01$ par value, $250,000$ shares authorized; $69,755$ and $68,851$ shares issued as of June $30,2023$ and December $31,2022$, respectively)	698		689
Additional paid-in capital	729,733		715,875
Treasury Stock at cost - 1,339 shares as of June 30, 2023 and December 31, 2022	(75,000)		(75,000)
Accumulated deficit	(101,834)		(193,158)
Accumulated other comprehensive loss	(952)		(1,259)
Total stockholders' equity	552,645		447,147
Total liabilities and stockholders' equity	\$ 1,337,720	\$	1,321,258

Lantheus Holdings, Inc. Condensed Consolidated Statements of Operations

(Unaudited)

(in thousands, except per share data)

	Three Mor June	nths e 30,		Six Months Ended June 30,			
	2023		2022		2023		2022
Revenues	\$ 321,700	\$	223,723	\$	622,484	\$	432,603
Cost of goods sold	119,053		85,694		342,761		165,504
Gross profit	202,647		138,029		279,723		267,099
Operating expenses							
Sales and marketing	36,456		27,492		69,073		47,846
General and administrative	26,151		32,598		49,422		70,186
Research and development	15,901		14,735		46,433		26,938
Total operating expenses	78,508		74,825		164,928		144,970
Operating income	124,139		63,204		114,795		122,129
Interest expense	4,933		1,469		9,924		2,978
Other income	(4,482)		(310)		(7,713)		(795)
Income before income taxes	123,688		62,045		112,584		119,946
Income tax expense	29,557		18,987		21,260		33,926
Net income	\$ 94,131	\$	43,058	\$	91,324	\$	86,020
Net income per common share:							
Basic	\$ 1.38	\$	0.63	\$	1.34	\$	1.26
Diluted	\$ 1.33	\$	0.61	\$	1.31	\$	1.22
Weighted-average common shares outstanding:							
Basic	 68,371		68,674		68,062		68,343
Diluted	71,014		70,796		69,957		70,412

Lantheus Holdings, Inc. Condensed Consolidated Statements of Comprehensive Income

(Unaudited) (in thousands)

	Three Months Ended June 30,					Six Months Ended June 30,			
	2023			2022		2023		2022	
Net income	\$	94,131	\$	43,058	\$	91,324	\$	86,020	
Other comprehensive income:									
Foreign currency translation		426		(224)		307		(84)	
Unrealized gain on cash flow hedges, net of tax		_		637		_		2,893	
Total other comprehensive income		426		413		307		2,809	
Comprehensive income	\$	94,557	\$	43,471	\$	91,631	\$	88,829	

Lantheus Holdings, Inc. Condensed Consolidated Statements of Changes in Stockholders' Equity

(Unaudited) (in thousands)

Six Months Ended June 30, 2023

	Commo	on Stock	Pasury Stock Pa		Additional Paid-In	Accumulated		Accumulated Other Comprehensive		Total Stockholders'				
_	Shares	Amount	Shares	Α	Amount	_	Capital		Deficit		Loss		Equity	
Balance, January 1, 2023	68,851	\$ 689	1,339	\$	(75,000)	\$	715,875	\$	(193,158)	\$	(1,259)	\$	447,147	
Net loss	_	_	_		_		_		(2,807)		_		(2,807)	
Other comprehensive loss	_	_	_		_		_		_		(119)		(119)	
Stock option exercises and employee stock plan purchases	120	1	_		_		2,781		_		_		2,782	
Vesting of restricted stock awards and units	813	8	_		_		(8)		_		_		_	
Shares withheld to cover taxes	(154)	(2)	_		_		(11,152)		_		_		(11,154)	
Stock-based compensation	_	_					9,667				_		9,667	
Balance, March 31, 2023	69,630	\$ 696	1,339	\$	(75,000)	\$	717,163	\$	(195,965)	\$	(1,378)	\$	445,516	
Net income									94,131				94,131	
Other comprehensive income	_	_	_		_		_		_		426		426	
Stock option exercises and employee stock plan purchases	73	1	_		_		1,346		_		_		1,347	
Vesting of restricted stock awards and units	68	1	_		_		(1)		_		_		_	
Shares withheld to cover taxes	(16)	_	_		_		(1,467)		_		_		(1,467)	
Stock-based compensation							12,692		_		_		12,692	
Balance, June 30, 2023	69,755	\$ 698	1,339	\$	(75,000)	\$	729,733	\$	(101,834)	\$	(952)	\$	552,645	

Six Months Ended June 30, 2022

	Six Months Ended June 30, 2022										
	Commo Shares	on Stock Amount	Treasu	ry Stock Amount	Additional Paid-In Capital	Accumulated Deficit	Accumulated Other Comprehensive Income	Total Stockholders' Equity			
Balance, January 1, 2022	67,739	\$ 677		<u> </u>	\$ 685,472	\$ (221,225)	\$ (485)				
Net income	_	_	_	_	_	42,962	_	42,962			
Other comprehensive income	_	_	_	_	_	- —	2,396	2,396			
Stock option exercises and employee stock plan purchases	296	3	_	_	5,931	. —	_	5,934			
Vesting of restricted stock awards and units	645	7	_	_	(7) —	_	_			
Shares withheld to cover taxes	(110)	(1)	_	_	(5,503	<u> </u>	_	(5,504)			
Stock-based compensation	_	_	_	_	5,623	_	_	5,623			
Balance, March 31, 2022	68,570	\$ 686		\$ —	\$ 691,516	\$ (178,263)	\$ 1,911	\$ 515,850			
Net income	_		_			43,058	_	43,058			
Other comprehensive income	_	_	_	_	_	_	413	413			
Stock option exercises and employee stock plan purchases	61	1	_	_	1,422	. —	_	1,423			
Vesting of restricted stock awards and units	108	1	_	_	(1) —	_	_			
Shares withheld to cover taxes	(13)	(1)	_	_	(823	<u> </u>	_	(824)			
Stock-based compensation	_				7,412			7,412			
Balance, June 30, 2022	68,726	\$ 687		\$	\$ 699,526	\$ (135,205)	\$ 2,324	\$ 567,332			

Cash, cash equivalents and restricted cash, end of period

Lantheus Holdings, Inc. Condensed Consolidated Statements of Cash Flows

(Unaudited) (in thousands)

(in thousands)					
		Six Months End June 30,			
		2023		2022	
Operating activities					
Net income	\$	91,324	\$	86,020	
Adjustments to reconcile net income to net cash flows from operating activities:					
Depreciation, amortization and accretion		30,375		23,893	
Impairment of long-lived assets		138,050		_	
Asset retirement obligation acceleration		_		500	
Amortization of debt related costs		2,155		491	
Changes in fair value of contingent assets and liabilities		(8,975)		26,900	
Provision for excess and obsolete inventory		3,179		3,185	
Stock-based compensation		22,359		13,035	
Deferred taxes		(39,616)		6,598	
Long-term indemnification receivable		(327)		(793)	
Long-term income tax payable and other long-term liabilities		1,386		1,154	
Other		1,735		2,172	
(Decreases) increases in cash from operating assets and liabilities:					
Accounts receivable		(39,078)		(100,994)	
Inventory		(19,255)		(2,013)	
Other current assets		(9,260)		744	
Other long-term assets		_		(533)	
Accounts payable		14,186		(861)	
Accrued expenses and other liabilities		(112,004)		23,363	
Net cash provided by operating activities		76,234		82,861	
Investing activities					
Capital expenditures		(19,865)		(7,533)	
Proceeds from sale of assets, net		_		1,800	
Acquisition of assets, net		(45,345)		_	
Net cash used in investing activities		(65,210)		(5,733)	
Financing activities					
Payments on long-term debt and other borrowings		(528)		(5,219)	
Contingent value rights settlement		(3,700)		_	
Proceeds from stock option exercises		3,189		6,780	
Proceeds from issuance of common stock		940		577	
Payments for minimum statutory tax withholding related to net share settlement of equity awards		(12,621)		(6,328)	
Net cash used in financing activities		(12,720)		(4,190)	
Effect of foreign exchange rates on cash, cash equivalents and restricted cash		139		(19)	
Net (decrease) increase in cash, cash equivalents and restricted cash		(1,557)		72,919	
Cash, cash equivalents and restricted cash, beginning of period		417,241		100,651	
Cook and a maintain and a maintain and a part of a maintain	¢.	415 604	¢	172.570	

415,684

173,570

Lantheus Holdings, Inc. Condensed Consolidated Statements of Cash Flows (Continued)

(Unaudited) (in thousands)

	Six Mont Jun	ths Ei e 30,	nded
	 2023		2022
Reconciliation to amounts within the condensed consolidated balance sheets	 		
Cash and cash equivalents	\$ 414,076	\$	171,427
Restricted cash included in other long-term assets	 1,608		2,143
Cash, cash equivalents and restricted cash at end of period	\$ 415,684	\$	173,570

Lantheus Holdings, Inc. Notes to Condensed Consolidated Financial Statements

(Unaudited)

Note Regarding Company References and Trademarks

Unless the context otherwise requires, references to the "Company" and "Lantheus" refer to Lantheus Holdings, Inc. and its direct and indirect wholly-owned subsidiaries; references to "Lantheus Holdings" refer to Lantheus Holdings, Inc. and not to any of its subsidiaries; references to "LMI" refer to Lantheus Medical Imaging, Inc., the direct subsidiary of Lantheus Holdings; references to "Cerveau," "Lantheus Real Estate," "Lantheus Two," "Lantheus Three" and "Progenics" refer to Cerveau Technologies, Inc., Lantheus MI Real Estate, LLC, Lantheus Two, LLC, Lantheus Three, LLC and Progenics Pharmaceuticals, Inc., respectively, each a wholly-owned subsidiary of LMI, and references to "EXINI" refer to EXINI Diagnostics AB, a wholly-owned subsidiary of Progenics. Solely for convenience, the Company refers to trademarks, service marks and trade names without the TM, SM and ® symbols. Those references are not intended to indicate, in any way, that the Company will not assert, to the fullest extent permitted under applicable law, its rights to its trademarks, service marks and trade names.

1. Basis of Presentation

The accompanying unaudited condensed consolidated financial statements include the accounts of Lantheus and have been prepared in accordance with generally accepted accounting principles for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, these condensed consolidated financial statements do not include all of the information and notes required by generally accepted accounting principles in the United States ("U.S. GAAP") for complete financial statements. In the opinion of management, all adjustments (consisting of normal and recurring adjustments) considered necessary for a fair statement have been included. The results of operations for the three and six months ended June 30, 2023 are not necessarily indicative of the results that may be expected for the year ended December 31, 2023 or any future period.

The condensed consolidated balance sheet at December 31, 2022 has been derived from the audited consolidated financial statements at that date but does not include all the information and notes required by U.S. GAAP for complete financial statements. These condensed consolidated financial statements and accompanying notes should be read in conjunction with the consolidated financial statements and notes thereto included in Item 8 of the Company's most recent Annual Report on Form 10-K for the year ended December 31, 2022 filed with the Securities Exchange Commission ("SEC") on February 23, 2023.

Progenics Acquisition

On June 19, 2020 (the "Closing Date"), pursuant to the Amended and Restated Agreement and Plan of Merger, dated as of February 20, 2020 (the "Merger Agreement"), by and among Lantheus Holdings, Plato Merger Sub, Inc., a wholly-owned subsidiary of Lantheus Holdings ("Merger Sub"), and Progenics, Lantheus Holdings completed the acquisition of Progenics by means of a merger of Merger Sub with and into Progenics, with Progenics becoming an indirect subsidiary of Lantheus Holdings following the completion of such merger (the "Progenics Acquisition").

In connection with the Progenics Acquisition, Lantheus Holdings issued 26,844,877 shares of Lantheus Holdings common stock and 86,630,633 contingent value rights (each a "CVR") tied to the financial performance of PYLARIFY to former Progenics stockholders and option holders. Each CVR entitled its holder to receive a pro rata share of aggregate cash payments equal to 40% of United States ("U.S.") net sales generated by PYLARIFY in 2022 and 2023 in excess of \$100.0 million and \$150.0 million, respectively. The Company's aggregate payments in respect of the CVRs, together with any other non-stock consideration treated as paid in connection with the Progenics Acquisition, was capped at 19.9% of the total consideration the Company paid in the Progenics Acquisition. Based on the Company's 2022 PYLARIFY net sales, the Company determined that the aggregate payment obligation under the CVRs was \$99.6 million, which was the maximum amount payable. The Company paid out this amount in May 2023 in full satisfaction of the CVRs.

2. Summary of Significant Accounting Policies

Recent Accounting Pronouncements

The Company has not adopted any new accounting standards during the six months ended June 30, 2023.

3. Revenue from Contracts with Customers

The following table summarizes revenue by revenue source as follows:

	Three Mo Jun	nths i e 30,	Ended		ths Ended e 30,			
Major Products/Service Lines (in thousands)	2023 2022			2023	2022			
Product revenue, net ⁽¹⁾	\$ 314,084	\$	217,782	\$ 606,340	\$	397,791		
License and royalty revenues ⁽²⁾	7,616		5,941	16,144		34,812		
Total revenues	\$ 321,700	\$	223,723	\$ 622,484	\$	432,603		

⁽¹⁾ The Company's product revenue includes PYLARIFY and DEFINITY among other products. This category represents the delivery of physical goods. The Company applies the same revenue recognition policies and judgments for all its principal products.

The Company classifies its revenues into three product categories: Radiopharmaceutical Oncology, Precision Diagnostics, and Strategic Partnerships and Other Revenue. Radiopharmaceutical Oncology includes PYLARIFY and AZEDRA. Precision Diagnostics includes DEFINITY, TechneLite and other diagnostic imaging products. Strategic Partnerships and Other Revenue includes strategic partnerships and other arrangements related to other products of the Company, including our royalty revenue from our license of RELISTOR.

On January 31, 2022, the Company entered into a global settlement agreement with Novartis Pharma AG ("Novartis"), Advanced Accelerator Applications USA, Inc. ("AAA"), Endocyte, Inc. ("Endocyte") and their affiliates (the "Novartis Agreement") to settle certain disputes between the parties. Under the Novartis Agreement, Novartis agreed to make a lump sum payment to the Company, as well as to reimburse the Company for certain fees and expenses in connection with certain litigation, and the Company agreed to license certain intellectual property to Novartis. In addition, the Company agreed to supply PYLARIFY for clinical purposes at an arms-length value which will be recorded revenue in the future as product is provided. In accordance with the Company's ASC 606, Revenue from Contracts with Customers, assessment, Novartis is considered to be a customer. The Company determined that the \$24.0 million that Novartis paid to the Company pursuant to the Novartis Agreement constituted a single element which was satisfied on the date of the execution of the Novartis Agreement. The Company determined that the license of intellectual property carried a fair value of \$24.0 million. As such, the Company assigned the value of the license to be \$24.0 million, which constitutes the entire transaction price and does not require further allocation. The Company determined that the \$24.0 million represented the point at which Novartis, as the licensee, was able to use and benefit from the license and recognized revenue when the license was granted to Novartis upon execution of the Novartis Agreement. The Company recognized the \$24.0 million fee as revenue on its consolidated statement of operations for the quarter ended March 31, 2022.

Revenue by product category on a net basis is as follows:

		Three Mor Jun	nths e 30,		Six Months Ended June 30,				
(in thousands)	2023			2022	2023			2022	
PYLARIFY	\$	210,522	\$	130,232	\$	405,992	\$	223,009	
Other radiopharmaceutical oncology		818		928		1,535		2,255	
Total radiopharmaceutical oncology		211,340		131,160		407,527		225,264	
DEFINITY		70,529		62,306		139,353		120,634	
TechneLite		21,594		19,440		42,580		42,045	
Other precision diagnostics		5,454		5,363		11,261		10,628	
Total precision diagnostics		97,577		87,109		193,194		173,307	
Strategic partnerships and other revenue		12,783		5,454		21,763		34,032	
Total revenues	\$	321,700	\$	223,723	\$	622,484	\$	432,603	

The Company would be required to allocate a portion of its revenue received from commercial contracts to future reporting periods to the extent the Company had performance obligations that extended beyond one year. However, the Company's performance obligations are typically part of contracts that have an original expected duration of one year or less. As such, the Company is not disclosing the aggregate amount of the transaction price allocated to performance obligations that are unsatisfied (or partially satisfied) as of the end of the reporting period.

⁽²⁾ The Company recognized \$24.0 million license revenue in the first quarter of 2022 related to an agreement with Novartis Pharma AG.

4. Fair Value of Financial Instruments

Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. In order to increase consistency and comparability of fair value measurements, financial instruments are categorized based on a hierarchy that prioritizes observable and unobservable inputs used to measure fair value into three broad levels, which are described below:

- Level 1 Inputs are unadjusted quoted prices in active markets for identical assets or liabilities that the Company has the ability to access at the measurement date.
- Level 2 Inputs include quoted prices for similar assets and liabilities in active markets, quoted prices for identical or similar assets or liabilities in markets that are not active, inputs other than quoted prices that are observable for the asset or liability (i.e., interest rates, yield curves, etc.) and inputs that are derived principally from or corroborated by observable market data by correlation or other means (market corroborated inputs).
- *Level 3* Unobservable inputs that reflect a Company's estimates about the assumptions that market participants would use in pricing the asset or liability. The Company develops these inputs based on the best information available, including its own data.

The Company's financial assets and liabilities measured at fair value on a recurring basis currently consist of money market funds and contingent consideration liabilities. The Company invests excess cash from its operating cash accounts in overnight investments and reflects these amounts in cash and cash equivalents in the condensed consolidated balance sheets at fair value using quoted prices in active markets for identical assets.

The tables below present information about the Company's assets and liabilities measured at fair value on a recurring basis:

	June 30, 2023							
(in thousands)	7	Total Fair Value		Level 1		Level 2		Level 3
Assets:								
Money market	\$	302,605	\$	302,605	\$	_	\$	_
Total assets	\$	302,605	\$	302,605	\$	_	\$	
Liabilities:								•
Contingent consideration liabilities	\$	3,000	\$	_	\$	_	\$	3,000
Total liabilities	\$	3,000	\$	_	\$		\$	3,000
							_	

		December 31, 2022							
(in thousands)	Т	otal Fair Value		Level 1		Level 2		Level 3	
Assets:						_			
Money market	\$	342,646	\$	342,646	\$	_	\$	_	
Total assets	\$	342,646	\$	342,646	\$	_	\$	_	
Liabilities:	·								
Contingent consideration liabilities	\$	111,600	\$	_	\$	_	\$	111,600	
Total liabilities	\$	111,600	\$	_	\$		\$	111,600	
			_				_		

During the three and six months ended June 30, 2023, there were no transfers into or out of Level 3.

As part of the Progenics Acquisition, the Company issued CVRs and recorded the fair value as part of consideration transferred. Each CVR entitled its holder to receive a pro rata share of aggregate cash payments equal to 40% of U.S. net sales generated by PYLARIFY in 2022 and 2023 in excess of \$100.0 million and \$150.0 million, respectively, subject to a maximum cap. Refer to Note 1, "Basis of Presentation" for further details on the CVRs. Based on the U.S. net sales generated by PYLARIFY in 2022, the Company paid out the maximum amount payable under the CVRs from available cash in May 2023 in full satisfaction of the CVR obligation.

The Company also assumed contingent consideration liabilities related to a previous acquisition completed by Progenics in 2013 ("2013 Acquisition"). These contingent consideration liabilities include potential payments of up to \$70.0 million if the Company attains certain net sales targets primarily for AZEDRA and 1095 (also known as 131 I-MIP-1095) and a \$5.0 million 1095 commercialization milestone. Additionally, there is a potential payment of up to \$10.0 million related to a 1404 commercialization

milestone. The Company's total potential payments related to the 2013 Acquisition are approximately \$85.0 million. The Company considers the contingent consideration liabilities relating to the 2013 Acquisition each a Level 3 instrument (one with significant unobservable inputs) in the fair value hierarchy. The estimated fair value of these was determined based on probability adjusted discounted cash flows and Monte Carlo simulation models that included significant estimates and assumptions pertaining to commercialization events and sales targets. The most significant unobservable inputs with respect to 1095 and 1404 are the probabilities of achieving regulatory approval of those development projects and subsequent commercial success.

Significant changes in any of the probabilities of success, the probabilities as to the periods in which sales targets and milestones will be achieved, discount rates or underlying revenue forecasts would result in a significantly higher or lower fair value measurement. The Company records the contingent consideration liability at fair value with changes in estimated fair values recorded in general and administrative expenses in the condensed consolidated statements of operations. The Company can give no assurance that the actual amounts paid, if any, in connection with the contingent consideration liabilities, will be consistent with any recurring fair value estimate of such contingent consideration liabilities.

The following tables summarize quantitative information and assumptions pertaining to the fair value measurement of liabilities using Level 3 inputs at June 30, 2023.

	Fair	Value at			Assu	mptions
<u>(in thousands)</u> Contingent consideration	June 30, 2023	December 31, 2022	Valuation Technique	Unobservable Input	June 30, 2023	December 31, 2022
liability: Net sales targets - PYLARIFY (CVRs)	\$ —	\$ 99,700	Probability adjusted discounted cash flow model	Period of expected milestone achievement and sales targets	May 2023	2022 - 2023
	Ψ	\$		Probability of success	N/A	100 %
1095 commercialization milestone	1,700	1,700	Probability adjusted discounted cash flow model			
				Period of expected milestone achievement	2026	2026
				Probability of success	40 %	40 %
				Discount rate	4.1 %	3.8 %
Net sales targets - AZEDRA and 1095	1,300	10,200	Monte Carlo simulation			
				Probability of success and sales targets	40% - 100%	20% - 100%
				Discount rate	16% - 17%	16% - 17%
Total	\$ 3,000	\$ 111,600				

For those financial instruments with significant Level 3 inputs, the following table summarizes the activities for the periods indicated:

<u>(in thousands)</u>		Financi Six Mon Jun		Financial Liabilities Six Months Ended June 30,			
		2023	2022	2023	2023 2022		
Fair value, beginning of period	\$	_	\$ 9,300	\$ 111,600	\$	86,200	
Changes in fair value included in net (loss) income		_	(1,600)	(8,975)		25,300	
Cash payments		_	_	(99,625)		_	
Fair value, end of period	\$	_	\$ 7,700	\$ 3,000	\$	111,500	

The change in fair value of the contingent financial liabilities resulted in a reduction of general and administrative expense of \$8.9 million for the six months ended June 30, 2023 and was primarily due to changes in revenue forecasts and the passage of time (excluding the CVRs). The Company made the applicable cash payment related to the CVRs in May 2023.

5. Income Taxes

The Company calculates income taxes at the end of each reporting period based on the estimated effective tax rate for the full year, adjusted for any discrete events which are recorded in the period they occur. Cumulative adjustments to the tax provision are recorded in the reporting period in which a change in the estimated annual effective tax rate is determined. The Company's income tax expense is presented below:

						ths Ended e 30,		
<u>(in thousands)</u>		2023		2022		2023		2022
Income tax expense	\$	29,557	\$	18,987	\$	21,260	\$	33,926

The Company regularly assesses its ability to realize its deferred tax assets. Assessing the realizability of deferred tax assets requires significant management judgment. In determining whether its deferred tax assets are realizable, the Company evaluated all available positive and negative evidence, and weighed the objective evidence and expected impact. The Company continues to retain immaterial valuation allowances against the net deferred tax assets of certain of its foreign subsidiaries.

In connection with the Company's acquisition of the medical imaging business from Bristol-Myers Squibb ("BMS") in 2008, the Company recorded a liability for uncertain tax positions related to the acquired business and simultaneously entered into a tax indemnification agreement with BMS under which BMS agreed to indemnify the Company for any payments made to settle those uncertain tax positions with the state taxing authorities. Accordingly, a long-term receivable is recorded to account for the expected value to the Company of future indemnification payments to be paid on behalf of the Company by BMS, net of actual tax benefits received by the Company. The tax indemnification receivable is recorded within other long-term assets.

In accordance with the Company's accounting policy, the change in the tax liabilities, penalties and interest associated with these obligations (net of any offsetting federal or state benefit) is recognized within income tax expense. As these reserves change, adjustments are included in income tax expense while the offsetting adjustment is included in other income. Assuming that the remaining receivable from BMS continues to be considered recoverable by the Company, there will be no effect on net income and no net cash outflows related to these liabilities.

6. Inventory

Inventory consisted of the following:

(in thousands)	June 30, 2023	December 31, 2022
Raw materials	\$ 25,312	\$ 19,987
Work in process	11,403	8,234
Finished goods	 15,086	7,254
Total inventory	\$ 51,801	\$ 35,475

Inventory costs associated with products that have not yet received regulatory approval are capitalized if the Company believes there is probable future commercial use of the product and future economic benefit of the asset. If future commercial use of the

product is not probable, then inventory costs associated with such product are expensed during the period the costs are incurred. The Company has no inventory pending regulatory approval as of June 30, 2023.

7. Property, Plant and Equipment, Net

Property, plant and equipment, net, consisted of the following:

(in thousands)		June 30, 2023	December 31, 2022		
Land	\$	9,480	\$ 13,450		
Buildings		70,467	76,329		
Machinery, equipment and fixtures		94,670	92,604		
Computer software		26,373	25,864		
Construction in progress		29,931	14,047		
	·	230,921	 222,294		
Less: accumulated depreciation and amortization		(100,940)	(100,128)		
Total property, plant and equipment, net	\$	129,981	\$ 122,166		

Depreciation and amortization expense related to property, plant and equipment, net, was \$3.3 million for the three months ended June 30, 2023 and 2022, respectively, and \$6.7 million and \$6.4 million for the six months ended June 30, 2023 and 2022, respectively.

During the three months ended June 30, 2023, as a result of a decline in expected future cash flows related to a certain asset group, the Company determined certain impairment triggers had occurred. The Company reviewed revised undiscounted cash flows that were estimated to be generated by the asset group as of June 30, 2023. Based on the undiscounted cash flow analysis, the Company determined that the asset group had net carrying values that exceeded their estimated undiscounted future cash flows. The Company then estimated the fair value of the asset group based on their discounted cash flows. The carrying value exceeded the fair value and as a result, the Company recorded a noncash impairment of \$6.0 million for the six months ended June 30, 2023 in cost of goods sold in the condensed consolidated statements of operations.

Long-Lived Assets Held for Sale

During the first quarter of 2023, the Company committed to a plan to sell a portion of its land and buildings associated with its Billerica, Massachusetts campus. Effective March 16, 2023, the Company entered into a purchase and sale agreement (the "P&S") with a prospective buyer. The assets were classified as held for sale and comprised entirely of property, plant and equipment, net. The Company determined that the fair value of the net assets being sold exceeded the carrying value as of June 30, 2023. The purchase price for the campus sale is \$10.0 million in cash. The transaction is expected to close during the second half of 2023.

8. Accrued Expenses and Other Liabilities and Other Long-Term Liabilities

Accrued expenses and other liabilities and other long-term liabilities are comprised of the following:

(in thousands)	 June 30, December 3 2023 2022			
Compensation and benefits	\$ 25,059	\$	30,425	
Freight, distribution and operations	60,206		49,067	
Accrued rebates, discounts and chargebacks	11,087		13,399	
Accrued professional fees	7,288		8,668	
Other	11,158		25,525	
Total accrued expenses and other liabilities	\$ 114,798	\$	127,084	
Operating lease liabilities (Note 16)	\$ 31,252	\$	25,442	
Long-term contingent liability (Note 4)	3,000		11,900	
Other long-term liabilities	14,816		8,813	
Total other long-term liabilities	\$ 49,068	\$	46,155	

9. Asset Retirement Obligations

The Company considers its legal obligation to remediate its facilities upon a potential decommissioning of its radioactive-related operations as an asset retirement obligation. The Company has production facilities which manufacture and process radioactive materials at its North Billerica, Massachusetts and Somerset, New Jersey sites. As of June 30, 2023, the liability is measured at the present value of the obligation expected to be incurred and is approximately \$25.1 million.

The following table provides a summary of the changes in the Company's asset retirement obligations:

<u>(in thousands)</u>	A	Amount		
Balance at January 1, 2023	\$	22,543		
Accretion expense		186		
Balance at June 30, 2023	\$	22,729		

The Company is required to provide the Massachusetts Department of Public Health and New Jersey Department of Environmental Protection financial assurance demonstrating the Company's ability to fund the decommissioning of its North Billerica, Massachusetts and Somerset, New Jersey production facilities, respectively, upon closure. The Company has provided this financial assurance in the form of a \$30.3 million surety bond.

10. Intangibles, Net

Intangibles, net, consisted of the following:

			J	une 30, 2023		
<u>(in thousands)</u>	Useful Lives (in years)	Amortization Method		Cost	Accumulated Amortization	Net
Trademarks	15 - 25	Straight-Line	\$	13,540	\$ (12,143)	\$ 1,397
Customer relationships	5 - 25	Accelerated		158,017	(104,827)	53,190
Currently marketed products	9	Straight-Line		132,800	(30,899)	101,901
Licenses	11 - 16	Straight-Line		85,800	(22,874)	62,926
Developed technology	9	Straight-Line		2,400	(810)	1,590
Total			\$	392,557	\$ (171,553)	\$ 221,004

December	31,	20	22
----------	-----	----	----

<u>(in thousands)</u>	Useful Lives (in years)	Amortization Method	Cost	cumulated nortization	 Net
Trademarks	15 - 25	Straight-Line	\$ 13,540	\$ (12,061)	\$ 1,479
Customer relationships	15 - 25	Accelerated	96,681	(95,009)	1,672
Currently marketed products	9 - 15	Straight-Line	275,700	(47,628)	228,072
Licenses	11 - 16	Straight-Line	85,800	(19,101)	66,699
Developed technology	9	Straight-Line	2,400	(677)	1,723
IPR&D	N/A	N/A	15,640	_	15,640
Total			\$ 489,761	\$ (174,476)	\$ 315,285

The Company recorded amortization expense for its intangible assets of \$12.4 million and \$8.3 million for the three months ended June 30, 2023 and 2022, respectively and \$23.5 million and \$16.6 million for the six months ended June 30, 2023 and 2022, respectively.

In March 2023, the Company stopped all development activities in relation to a future indication associated with AZEDRA, which is classified as an in process research and development ("IPR&D") intangible asset. The Company did not identify any alternative future uses or development programs for the asset, therefore the asset group, which consists of the IPR&D asset and a currently marketed product, was assessed for impairment as of March 31, 2023. The Company considered several factors including market share, price and competitive product offerings in evaluating the quantitative impact of the future cash flows. The Company

concluded that the carrying amount exceeded the fair value of the asset group, which had no value. Accordingly, in the three months ended March 31, 2023, the Company recorded a non-cash impairment charge associated with the IPR&D asset of \$15.6 million in research and development expenses and a non-cash impairment charge of \$116.4 million in cost of goods sold in the condensed consolidated statements of operations.

In February 2023, the Company entered into an agreement with Cerveau to purchase all of the outstanding capital stock of Cerveau for approximately \$35.3 million. In May 2023, upon successful completion of a technology transfer, the Company paid \$10.0 million to the selling stockholders of Cerveau. This additional contingent payment was capitalized as part of the asset cost and increased the Company's customer relationship intangible assets. See Note 19, "Acquisition of Assets" for further discussion of the Cerveau acquisition.

The below table summarizes the estimated aggregate amortization expense expected to be recognized on the above intangible assets:

(in thousands)	Amou	nt
Remainder of 2023	\$	24,118
2024		42,045
2025		30,991
2026		31,790
2027		26,263
2028 and thereafter		65,797
Total	\$	221,004

11. Long-Term Debt, Net, and Other Borrowings

As of June 30, 2023, the Company's maturities of principal obligations under its long-term debt and other borrowings are as follows:

(in thousands)	Amount				
Remainder of 2023	\$	_			
2024		_			
2025		_			
2026		_			
2027		575,000			
Total principal outstanding		575,000			
Unamortized debt issuance costs		(15,736)			
Finance lease liabilities		355			
Total		559,619			
Less: current portion		(384)			
Total long-term debt, net and other borrowings	\$	559,235			

In December 2022, the Company refinanced its existing credit facility, consisting of (i) a \$200.0 million five-year term loan facility (the "2019 Term Facility") and (ii) a \$200.0 million five-year revolving credit facility (the "2019 Revolving Facility" and, together with the 2019 Term Facility, the "2019 Facility"), with a new \$100.0 million delayed draw term loan facility (the "2022 Term Facility" and, the loans thereunder, the "Term Loans") and a new \$350.0 million five-year revolving credit facility (the "2022 Revolving Facility" and, together with the 2022 Term Facility, the "2022 Facility").

The Company used approximately \$7.8 million of cash on hand to primarily repay the principal amount of the loans outstanding related to the 2019 Facility through the nine months ended September 30, 2022. In addition, in December 2022, the Company used approximately \$167.6 million of cash on hand to repay in full the aggregate remaining principal amount of the loans outstanding under the 2019 Facility and to pay related interest, transaction fees and expenses.

The Company paid off the 2019 Term Facility using available cash and did not utilize another term loan to fund the payoff. While the 2022 Term Facility allowed for a delayed draw term loan, the loan was not drawn upon. The Company recorded a loss on extinguishment of debt of \$0.6 million related to the write-off of unamortized debt issuance costs and debt discounts associated with

the 2019 Term Facility. In addition, the Company incurred and capitalized \$2.7 million of new deferred financing costs related to the refinancing.

2022 Term Facility

The Company expected to draw from the 2022 Term Facility only if a proposed offering of notes by the Company was not consummated. The commitment of the lenders to provide the 2022 Term Facility was terminated upon the completion of such note offering in December 2022. The 2022 Term Facility included a commitment fee equal to 0.20% per annum on the average daily unused amount of the 2022 Term Facility, which would have been payable commencing on January 16, 2023 and would have ended on the earliest of (i) the day the Term Loans under the 2022 Term Facility were funded, (ii) the last day of the delayed draw availability period and (iii) the day the commitments under the 2022 Term Facility were reduced to zero. The Company did not draw from the 2022 Term Facility and, as such, no interest is due under this instrument.

2022 Revolving Facility

Under the terms of the 2022 Revolving Facility, the lenders are committed to extending credit to the Company from time to time until December 2, 2027 consisting of revolving loans (the "Revolving Loans") in an aggregate principal amount not to exceed \$350.0 million (the "Revolving Commitment") at any time, including a \$20.0 million sub-facility for the issuance of letters of credit (the "Letters of Credit") and a \$10.0 million sub-facility for swingline loans (the "Swingline Loans"). The Letters of Credit, Swingline Loans and the Revolving Loans, if used, are expected to be used for working capital and for other general corporate purposes.

The Revolving Loans bear interest, with pricing based from time to time at the Company's election, at (i) the secured overnight financing rate as published by the Federal Reserve Bank of New York on its website plus an applicable margin that ranges from 1.50% to 2.50% based on the Company's total net leverage ratio or (ii) the alternative base rate plus an applicable margin that ranges from 0.50% to 1.50% based on the Company's total net leverage ratio. The 2022 Revolving Facility also includes an unused commitment fee at a rate ranging from 0.15% to 0.35% per annum based on the Company's total net leverage ratio.

The Company is permitted to voluntarily prepay the Revolving Loans, in whole or in part, or reduce or terminate the Revolving Commitment, in each case, without premium or penalty. On any business day on which the total amount of outstanding Revolving Loans, Letters of Credit and Swingline Loans exceeds the total Revolving Commitment, the Company must prepay the Revolving Loans in an amount equal to such excess. The Company is not required to make mandatory prepayments under the 2022 Revolving Facility. As of June 30, 2023, there were no outstanding borrowings under the 2022 Revolving Facility.

The Company has the right to request an increase to the Revolving Commitment in an aggregate principal amount of up to the sum of \$335.0 million or consolidated EBITDA for the four consecutive fiscal quarters most recently ended, plus additional amounts in certain circumstances (collectively, the "Incremental Cap"), minus certain incremental term loans made pursuant to specified incremental term loan commitments ("Incremental Term Loans"). The Company has the right to request Incremental Term Loans in an aggregate principal amount of up to the Incremental Cap less any incremental increases to the Revolving Commitment. Proceeds of Incremental Term Loans may be used for working capital and for other general corporate purposes and will bear interest at rates agreed between the Company and the lenders providing the Incremental Term Loans.

2022 Facility Covenants

The 2022 Facility contains a number of affirmative, negative and reporting covenants, as well as financial maintenance covenants pursuant to which the Company is required to be in quarterly compliance, measured on a trailing four quarter basis, with two financial covenants. The minimum interest coverage ratio, commencing with the fiscal quarter ended December 31, 2022, must be at least 3.00 to 1.00. The maximum total net leverage ratio permitted by the financial covenant is displayed in the table below:

2022 Credit Agreement

Period	Total Net Leverage Ratio
Q2 2023 to Q4 2023	4.00 to 1.00
Q1 2024 and thereafter	3.50 to 1.00

The 2022 Facility contains usual and customary restrictions on the ability of the Company and its subsidiaries to: (i) incur additional indebtedness (ii) create liens; (iii) consolidate, merge, sell or otherwise dispose of all or substantially all of its assets; (iv) sell certain assets; (v) pay dividends on, repurchase or make distributions in respect of capital stock or make other restricted payments; (vi) make certain investments; (vii) repay subordinated indebtedness prior to stated maturity; and (viii) enter into certain transactions with its affiliates.

Upon an event of default, the Administrative Agent will have the right to declare the loans and other obligations outstanding under the 2022 Facility immediately due and payable and all commitments immediately terminated.

The 2022 Facility is guaranteed by Lantheus Holdings, and certain subsidiaries of LMI, including Progenics and Lantheus Real Estate, and obligations under the 2022 Facility are generally secured by first priority liens over substantially all of the assets of each of LMI, Lantheus Holdings, and certain subsidiaries of LMI, including Progenics and Lantheus Real Estate (subject to customary exclusions set forth in the transaction documents) owned as of December 2, 2022 or thereafter acquired.

Convertible Notes

On December 8, 2022, the Company issued \$575.0 million in aggregate principal amount of 2.625% Convertible Senior Notes due 2027 (the "Notes"), which includes \$75.0 million in aggregate principal amount of Notes sold pursuant to the full exercise of the initial purchasers' option to purchase additional Notes. The Notes were issued under an indenture, dated as of December 8, 2022 (the "Indenture"), among the Company, LMI (the "Guarantor"), as Guarantor, and U.S. Bank Trust Company, National Association, as Trustee. The net proceeds from the issuance of the Notes were approximately \$557.8 million after deducting the initial purchasers' discounts and offering expenses payable by the Company.

The Notes are senior unsecured obligations of the Company. The Notes are fully and unconditionally guaranteed on a senior unsecured basis by the Guarantor. The Notes bear interest at a rate of 2.625% per year, payable semi-annually in arrears on June 15 and December 15 of each year, beginning on June 15, 2023, and will mature on December 15, 2027 unless earlier redeemed, repurchased or converted in accordance with their terms. The initial conversion rate for the Notes is 12.5291 shares of the Company's common stock per \$1,000 in principal amount of Notes (which is equivalent to an initial conversion price of approximately \$79.81 per share of the Company's common stock, representing an initial conversion premium of approximately 42.5% above the closing price of \$56.01 per share of the Company's common stock on December 5, 2022). In no event shall the conversion rate per \$1,000 in principal amount of notes exceed 17.8539 shares of the Company's common stock. Prior to the close of business on the business day immediately preceding September 15, 2027, the Notes may be converted at the option of the holders only upon occurrence of specified events and during certain periods, and thereafter until the close of business on the business day immediately preceding the maturity date, the Notes may be converted at any time. The Company will satisfy any conversion by paying cash up to the aggregate principal amount of the Notes to be converted and by paying or delivering, as the case may be, cash, shares of the Company's common stock, or a combination of cash and shares of the Company's common stock, at its election, in respect of the remainder, if any, of its conversion obligation in excess of the aggregate principal amount of the Notes being converted. The Company may redeem for cash all or any portion of the Notes, at its option, on or after December 22, 2025 if the closing sale price per share of the Company's common stock exceeds 130% of the conversion price of the Notes for a specified period of time. The redemption da

The Company evaluated the Notes upon completion of the sale and concluded on the following features:

- *Conversion Feature:* The Company determined that the conversion feature qualifies for the classification of equity. As a result, the conversion feature should not be bifurcated as a derivative instrument and the Notes were accounted for as a single liability.
- *Redemption Features:* The redemption features were reviewed within the Notes and the Company determined that the redemption features are closely related to the Notes and as such should not be separately accounted for as a bifurcated derivative instrument.
- Additional Interest Features: The Notes may result in additional interest if the Company fails to timely file any document that the Company is required to file with the SEC pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934. The Company will pay additional interest on the notes at a rate equal to 0.25% to 0.50% per annum based on the principal amount of notes outstanding for each day the Company failure to file has occurred or the notes are not otherwise freely tradable. Further, if the notes are assigned a restricted CUSIP number or the notes are not otherwise freely tradable pursuant to Rule 144 under the Securities Act by holders other than our affiliates or holders that were our affiliates at any time during the three months immediately preceding as of the 385th day after the last date of original issuance of the notes offered hereby, the Company will pay additional interest on the notes at a rate equal to (i) 0.25% to 0.50% per annum based on the principal amount of notes outstanding for each day until the restrictive legend has been removed from the notes, the notes are assigned an unrestricted CUSIP and the notes are freely tradable. The Company concluded that the interest feature is unrelated to the credit risk and should be bifurcated from the Notes, however, the Company assessed the probabilities of triggering events occurring under these features and does not expect to trigger the aforementioned events. These events will continue to be monitored to determine whether the interest feature will be bifurcated if it has value.

As of June 30, 2023, the carrying value of the Notes was \$575.0 million and the fair value of the liability was \$575.0 million. The Company recorded interest expense of approximately \$3.9 million and \$7.8 million related to the Notes for the three and six months ended June 30, 2023, respectively. There were no conversions of Notes during the six months ended June 30, 2023.

12. Derivative Instruments

The Company has used, but does not currently use, interest rate swaps to reduce the variability in cash flows associated with portions of the Company's interest payments on variable rate debt. In March 2020, the Company entered into interest rate swap contracts to fix the LIBOR rate on a notional amount of \$100.0 million through May 31, 2024. The average fixed LIBOR rate on the interest rate swaps was approximately 0.82%. This agreement involved the receipt of floating rate amounts in exchange for fixed rate interest payments over the life of the agreement without an exchange of the underlying principal amount. The interest rate swaps were designated as cash flow hedges. In accordance with hedge accounting, the interest rate swaps were recorded on the Company's consolidated balance sheets at fair value, and changes in the fair value of the swap agreements were recorded to other comprehensive loss and reclassified to interest expense in the period during which the hedged transaction affected earnings or it will become probable that the forecasted transaction would not occur.

On December 2, 2022, the Company voluntarily terminated the interest rate swap contracts in connection with the refinancing of debt. Upon termination, the Company received approximately \$5.6 million in cash and the remaining balance of approximately \$5.5 million in accumulated other comprehensive income related to the interest rate swap contracts were reclassified into earnings.

13. Accumulated Other Comprehensive (Loss) Income

The components of accumulated other comprehensive (loss) income, net of tax of zero and \$1.1 million for the six months ended June 30, 2023 and 2022, respectively, consisted of the following:

(in thousands)	reign currency translation	1	Unrealized loss on cash flow hedges	Accumulated other nprehensive income (loss)
Balance at January 1, 2023	\$ (1,259)	\$	_	\$ (1,259)
Other comprehensive loss before reclassifications	307		_	307
Amounts reclassified to earnings	 <u> </u>		<u> </u>	_
Balance at June 30, 2023	\$ (952)	\$	_	\$ (952)
Balance at January 1, 2022	\$ (754)	\$	269	\$ (485)
Other comprehensive (loss) income before reclassifications	(84)		2,710	2,626
Amounts reclassified to earnings	_		183	183
Balance at June 30, 2022	\$ (838)	\$	3,162	\$ 2,324

14. Stock-Based Compensation

The following table presents stock-based compensation expense recognized in the Company's accompanying condensed consolidated statements of operations:

	 Three Moi Jun	nths I e 30,	Ended		hs Ended e 30,			
(in thousands)	2023		2022	2023		2022		
Cost of goods sold	\$ 2,231	\$	1,083	\$ 3,873	\$	1,995		
Sales and marketing	1,959		1,675	4,221		2,688		
General and administrative	6,670		3,784	11,072		6,786		
Research and development	1,832		870	3,193		1,566		
Total stock-based compensation expense	\$ 12,692	\$	7,412	\$ 22,359	\$	13,035		

15. Leases

Operating and finance lease assets and liabilities are as follows:

(in thousands)	Classification		June 30, 2023	December 31, 2022
Assets				
Operating	Other long-term assets	\$	24,205	\$ 19,033
Finance	Property, plant and equipment, net		400	582
Total leased assets		\$	24,605	\$ 19,615
Liabilities		·		
Current				
Operating	Accrued expenses and other liabilities	\$	1,601	\$ 2,177
Finance	Current portion of long-term debt and other borrowings		384	354
Noncurrent				
Operating	Other long-term liabilities		31,252	25,442
Finance	Long-term debt, net and other borrowings		(29)	231
Total leased liabilities		\$	33,208	\$ 28,204

On May 4, 2023, the Company entered into a modification to the operating lease for office space in Bedford, Massachusetts, (the "Existing Premises") that was executed in February 2022. The lease commenced and was recorded in December 2022 for \$11.0 million and the initial term was set to expire in June 2031. The lease modification includes a lease of additional office and laboratory space at the Bedford location (the "Additional Premises") for a term of 15 years and 4 months and extends the term of the lease for the Existing Premises to be coterminous with the term of the lease for the Additional Premises. As a result of the extended term for the Existing Premises, the Company recorded an additional right-of-use asset and liability of \$6.0 million. The modification also contains a provision to convert the rent schedule of the Existing Premises from gross to triple net in 2024, which may result in an additional adjustment to the right of use asset and liability. The Additional Premises are currently being renovated and the associated right-of-use asset and liability will be recorded once renovations are substantially complete. The lease payments for the Additional Premises are currently expected to commence in the second half of 2024 when renovation of the Additional Premises is expected to be completed.

16. Net Income Per Common Share

A summary of net income per common share is presented below:

	Three Mor		Six Months Ended June 30,				
(in thousands, except per share amounts)	 2023		2022		2023		2022
Net income	\$ 94,131	\$	43,058	\$	91,324	\$	86,020
Basic weighted-average common shares outstanding	68,371		68,674		68,062		68,343
Effect of dilutive stock options	399		443		391		441
Effect of dilutive restricted stock	1,447		1,679	1,504			1,628
Effect of convertible debt instrument	797		_		_		_
Diluted weighted-average common shares outstanding	71,014		70,796		69,957		70,412
	 <u>:</u>				;		
Basic income per common share	\$ 1.38	\$	0.63	\$	1.34	\$	1.26
Diluted income per common share	\$ 1.33	\$	0.61	\$	1.31	\$	1.22
Antidilutive securities excluded from diluted net income per common share	349		268		389		196

Impact of the Notes

The Company considered whether the Notes are participating securities through the two-class method. The Company determined that if a cash dividend is paid that is greater than the then stock price, the holder of Notes will receive cash on an if-converted basis.

While this feature is considered to be a participating right; basic earnings per share is only impacted if the Company's earnings exceeds the current share price, regardless of whether such dividend is declared. During the six months ended June 30, 2023 and 2022, no such dividend was declared. In addition, the Company is required to settle the principal amount of the Notes in cash upon conversion, and therefore, the Company uses the if-converted method for calculating any potential dilutive effect of the conversion option on diluted net income per share, if applicable, unless the application of the two-class method is dilutive. The conversion option will have a dilutive impact on net income per share of Common Stock when the average price per share of the Company's common stock for a given period exceeds the conversion price of the Notes of \$79.81 per share. During the three months ended June 30, 2023, the average price per share of the Company's Common Stock exceeded the conversion price of the Notes. During the six months ended June 30, 2023, the average price per share of the Company's Common Stock was below the conversion price of the Notes.

17. Other Income

Other income consisted of the following:

	Three Months Ended June 30,				Six Mont Jun	hs Ended e 30,			
(in thousands)		2023		2022	2023		2022		
Foreign currency (gains) losses	\$	(224)	\$	125	\$ 22	\$	44		
Tax indemnification income, net		(232)		(396)	(328)		(792)		
Interest income		(4,027)		(39)	(7,550)		(47)		
Other		1		_	143		_		
Total other income	\$	(4,482)	\$	(310)	\$ (7,713)	\$	(795)		

18. Commitments and Contingencies

Legal Proceedings

From time to time, the Company is a party to various legal proceedings arising in the ordinary course of business. In addition, the Company has in the past been, and may in the future be, subject to investigations by governmental and regulatory authorities, which expose it to greater risks associated with litigation, regulatory or other proceedings, as a result of which the Company could be required to pay significant fines or penalties. The costs and outcome of litigation, regulatory or other proceedings cannot be predicted with certainty, and some lawsuits, claims, actions or proceedings may be disposed of unfavorably to the Company and could have a material adverse effect on the Company's results of operations or financial condition. In addition, intellectual property disputes often have a risk of injunctive relief which, if imposed against the Company, could materially and adversely affect its financial condition or results of operations. If a matter is both probable to result in material liability and the amount of loss can be reasonably estimated, the Company estimates and discloses the possible material loss or range of loss. If such loss is not probable or cannot be reasonably estimated, a liability is not recorded in its condensed consolidated financial statements.

As of June 30, 2023, the Company did not have any material ongoing litigation to which the Company was a party.

19. Acquisition of Assets

On February 6, 2023, the Company acquired Cerveau. Cerveau's primary asset is MK-6240, a second-generation F 18-labeled positron emission tomography ("PET") imaging agent that targets Tau tangles in Alzheimer's disease. The Company determined that upon review of the Cerveau acquisition, the transaction did not meet the definition of a business combination and is therefore treated as an asset acquisition.

In February 2023, the Company made an upfront payment of approximately \$35.3 million to the stockholders of Cerveau (the "Selling Stockholders") and paid an additional \$10.0 million in May 2023 upon the successful completion of a technology transfer. The Company could pay up to an additional \$51.0 million in milestone payments upon achievement of specified U.S. regulatory milestones related to MK-6240. The Selling Stockholders are also eligible to receive up to \$1.2 billion in sales milestone payments upon the achievement of specified annual commercial sales thresholds of MK-6240 in the event the Company pursues commercialization, as well as up to \$13.5 million in research revenue milestones upon achievement of specified annual research revenue thresholds. Additionally, the Company will pay to the Selling Stockholders up to double-digit royalty payments for research revenue and commercial sales. Research revenue is derived from existing partnerships with pharmaceutical companies that use MK-6240 in clinical trials and includes milestone and dose-related payments. The purchase agreement pursuant to which the Company purchased Cerveau specifies, among other things, that certain members of the Selling Stockholders will also provide transition and clinical development services for a prescribed time following the closing of the transaction.

In December 2022, the Company made upfront payments of \$260.0 million to POINT Biopharma Global Inc. ("POINT") as a part of an asset acquisition with the potential for additional milestone payments of approximately \$1.8 billion for the two licensed assets based on U.S. Food and Drug Administration ("FDA") approval and net sales and commercial milestones.

Under the terms of the license agreement between Lantheus Two and POINT for PNT2002, Lantheus Two paid POINT an upfront cash payment of \$250.0 million, and could pay up to an additional \$281.0 million in milestone payments upon the achievement of specified U.S. and ex-U.S. regulatory milestones related to PNT2002. POINT is also eligible to receive up to \$1.3 billion in sales milestone payments upon the achievement of specified annual sales thresholds of PNT2002.

Under the terms of the license agreement between Lantheus Three and POINT for PNT2003, Lantheus Three paid POINT an upfront cash payment of \$10.0 million, and could pay up to an additional \$34.5 million in milestone payments upon the achievement of specified U.S. and ex-U.S. regulatory milestones related to PNT2003. POINT is also eligible to receive up to \$275.0 million in sales milestone payments upon the achievement of specified annual sales thresholds of PNT2003.

Additionally, the Company will pay POINT royalties on net sales, beyond certain financial thresholds and subject to conditions, of 20% for PNT2002 and 15% for PNT2003. Costs of IPR&D projects acquired as part of an asset acquisition that have no alternative future use are expensed when incurred, and therefore, a charge of \$260.0 million was recognized in research and development expenses during the year ended December 31, 2022.

20. Segment Information

The Company operates as one business segment: the development, manufacture and sale of innovative imaging diagnostics, radiotherapeutics, and artificial intelligence solutions designed to enable clinicians to Find, Fight and Follow disease to deliver better patient outcomes. This conclusion reflects the Company's focus on the performance of the business on a consolidated worldwide basis. The results of this operating segment are regularly reviewed by the Company's chief operating decision maker, the Chief Executive Officer. The Company's chief operating decision maker does not manage any part of the Company separately, and the allocation of resources and assessment of performance are based on the Company's consolidated operating results.

21. Subsequent Events

On August 2, 2023, the Company entered into an agreement to sell the right to all of its RELISTOR net sales royalties under its license agreement with Bausch, relating to royalties payable in respect of the second quarter of 2023 and subsequent quarters. The Company received an initial payment of approximately \$98.0 million in connection with the sale and has the right to receive an additional payment from the buyer of \$5.0 million if worldwide net sales of RELISTOR in 2025 exceed a specified threshold. Proceeds from this transaction will be used in the Company's core businesses and product pipeline.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Cautionary Note Regarding Forward-Looking Statements

Some of the statements contained in this Quarterly Report on Form 10-O are forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Securities Act"), and Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"). These forward-looking statements, including, in particular, statements about our plans, strategies, prospects and industry estimates are subject to risks and uncertainties. These statements identify prospective information and can generally be identified by words such as "anticipates," "believes," "can," "could," "designed," "estimates," "expects," "hopes," "intends," "launch," "may," "pipeline," "plans," "predicts," "seeks," "should," "target," "will," "would" and similar expressions, or by express or implied discussions regarding potential marketing approvals or new indications for the collaborations, products candidates or approved products described in this Quarterly Report on Form 10-Q, or regarding potential future revenues from such collaborations, product candidates and products. Examples of forward-looking statements include statements we make relating to our outlook and expectations including, without limitation, in connection with: (i) continued market expansion and penetration for our established commercial products, particularly PYLARIFY and DEFINITY, in a competitive environment in which other imaging agents have been approved and are being commercialized, and our ability to clinically and commercially differentiate our products from other products; (ii) our ability to have third parties manufacture our products and our ability to manufacture DEFINITY in our in-house manufacturing facility; (iii) the global availability of Molybdenum-99 ("Mo-99") and other raw material and key components; (iv) the efforts and timing for clinical development, regulatory approval and successful commercialization of our product candidates and new clinical applications and territories for our products, in each case, that we or our strategic partners may undertake; (v) our strategies, future prospects, and our projected growth, including revenue related to our collaboration agreements with POINT Biopharma Global Inc. (vi) our ability to successfully continue existing clinical development partnerships using MK-6240 as a research tool and to further develop and commercialize such research tool; and (vii) our ability to identify and acquire or in-license additional diagnostic and therapeutic product opportunities in oncology and other strategic areas. Forward-looking statements are based on our current expectations and assumptions regarding our business, the economy and other future conditions. Because forward-looking statements relate to the future, such statements are subject to inherent uncertainties, risks and changes in circumstances that are difficult to predict. Our actual results may differ materially from those contemplated by the forward-looking statements. These statements are neither statements of historical fact nor guarantees or assurances of future performance. The matters referred to in the forward-looking statements contained in this Quarterly Report on Form 10-Q may not in fact occur. We caution you, therefore, against relying on any of these forward-looking statements. These forward-looking statements are subject to a number of risks, uncertainties and assumptions, including those described in Part I, Item 1A, "Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2022, and in Part II, Item 1A. "Risk Factors" in this Quarterly Report on Form 10-Q.

Any forward-looking statement made by us in this Quarterly Report on Form 10-Q speaks only as of the date on which it is made. Factors or events that could cause our actual results to differ may emerge from time to time, and it is not possible for us to predict all of them. We undertake no obligation to publicly update any forward-looking statement, whether as a result of new information, future developments or otherwise, except as may be required by law.

Available Information

Our global Internet site is www.lantheus.com. We routinely make available important information, including copies of our Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K and amendments to those reports filed or furnished pursuant to Section 13(a) or 15(d) of the Exchange Act, as soon as reasonably practicable after those reports are electronically filed with, or furnished to, the SEC, free of charge on our website at investor.lantheus.com. We recognize our website as a key channel of distribution to reach public investors and as a means of disclosing material non-public information to comply with our disclosure obligations under SEC Regulation FD. Information contained on our website shall not be deemed incorporated into, or to be part of this Quarterly Report on Form 10-Q, and any website references are not intended to be made through active hyperlinks.

Our reports filed with, or furnished to, the SEC are also available on the SEC's website at www.sec.gov, and for Annual Reports on Form 10-K and Quarterly Reports on Form 10-Q, in an iXBRL (Inline Extensible Business Reporting Language) format. iXBRL is an electronic coding language used to create interactive financial statement data over the Internet. The information on our website is neither part of nor incorporated by reference into this Quarterly Report on Form 10-Q.

The following discussion and analysis of our financial condition and results of operations should be read together with the condensed consolidated financial statements and the related notes included in Item 1 of this Quarterly Report on Form 10-Q as well as the other factors described in Part I, Item 1A. "Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2022, and in Part II, Item 1A. "Risk Factors" in this Quarterly Report on Form 10-Q.

Overview

Our Business

With more than 65 years of experience in delivering life-changing science, we are committed to delivering better patient outcomes and improving lives through diagnostics, radiotherapeutics and artificial intelligence solutions that enable clinicians to Find, Fight and Follow disease. We classify our products in three categories: Radiopharmaceutical Oncology, Precision Diagnostics, and Strategic Partnerships and Other Revenue. Our Radiopharmaceutical Oncology diagnostics and therapeutics help healthcare professionals ("HCPs") Find, Fight and Follow cancer. Our leading Precision Diagnostic products assist HCPs to Find and Follow diseases, with a focus in cardiology. Our Strategic Partnerships focus on enabling precision medicine through the use of biomarkers, digital solutions and pharma services platforms, and includes our license of RELISTOR to Bausch Health Companies, Inc. ("Bausch").

Our commercial products are used by oncologists, urologists, nuclear medicine physicians, cardiologists, sonographers, technologists, radiologists, and internal medicine physicians working in a variety of clinical settings. We believe that our diagnostic products provide improved clinical insight that enables HCPs to better detect and characterize, or rule out, disease, with the potential to achieve better patient outcomes, reduce patient risk and limit overall costs throughout the healthcare system.

We produce and market our products throughout the U.S., selling primarily to hospitals, independent diagnostic testing facilities, government facilities, integrated delivery networks, radiopharmacies, clinics, and group practices. We sell our products outside the U.S. through a combination of direct distribution in Canada and third party distribution relationships in Europe, Canada, Australia, Asia-Pacific, Central America and South America.

Our executive offices are located in Bedford, Massachusetts, with additional offices in North Billerica, Massachusetts; Somerset, New Jersey; Montreal, Canada and Lund, Sweden.

Recent Developments

Exclusive License for PNT2002 and PNT2003

On December 20, 2022, we announced the closing of a set of strategic collaborations with POINT Biopharma Global Inc. ("POINT"), in which we were granted a license to exclusive worldwide rights (excluding Japan, South Korea, China (including Hong Kong, Macau and Taiwan), Singapore and Indonesia) to co-develop and commercialize POINT's PNT2002 and PNT2003 product candidates.

With respect to PNT2002, POINT is generally responsible for funding and development activities required for FDA approval, including generating all clinical and nonclinical data, analysis and other information, and we are responsible for preparing for and seeking regulatory approval, as well as performing and funding all future development and commercialization following such approval. POINT will be responsible for all manufacturing of PNT2002, subject to certain exceptions described in the PNT2002 License Agreement.

In April 2023, we announced with POINT that the FDA had granted Fast Track designation for PNT2002. Fast Track is a process designed to facilitate the development and expedite the review of drugs to treat serious conditions and address unmet needs.

With respect to PNT2003, POINT is responsible for curating all data, analysis and other information necessary for regulatory approval, and supporting us in the preparation of regulatory filings, and we are responsible for preparing for and seeking regulatory approval of all such applications, as well as performing and funding all future development and commercialization following such approval. POINT will be responsible for all manufacturing of PNT2003, subject to certain exceptions described in the PNT2003 License Agreement.

For more information, see Note 19, "Acquisition of Assets" in our consolidated financial statements included herein.

Acquisition of Cerveau Technologies, Inc.

On February 6, 2023, we announced that we acquired Cerveau. Cerveau's asset is MK-6240, a second-generation F 18-labeled PET imaging agent that targets Tau tangles in Alzheimer's disease. Under the terms of the agreement, we paid the Selling Stockholders an upfront payment of \$35.0 million in February 2023 and paid an additional \$10.0 million in May 2023 upon the successful completion of a technology transfer. The Selling Stockholders are also eligible to receive additional development and commercial milestone payments. Additionally, we will pay double-digit royalty payments for research revenue and commercial sales. Research revenue is derived from existing partnerships with pharmaceutical companies that use MK-6240 in clinical trials and includes milestone and dose-related payments. Pursuant to the terms of the stock purchase agreement for Cerveau, certain members of the Selling Stockholders will also provide transition and clinical development services for a prescribed time following the closing of the transaction.

For more information, see Note 19, "Acquisition of Assets" in our consolidated financial statements included herein.

Key Factors Affecting Our Results

Our business and financial performance have been, and continue to be, impacted by the following:

Continued Growth of PYLARIFY

PYLARIFY, an F 18-labeled PET imaging agent targeting prostate-specific membrane antigen ("PSMA"), was approved by the FDA in May 2021 and commercially launched in the U.S. in June 2021.

In March 2022, we announced a strategic collaboration with Novartis to include PYLARIFY in prostate cancer trials with PLUVICTO, Novartis' PSMA-targeted therapeutic. In addition, in 2022, we entered into an agreement with Curium to add PYLARIFY to its PSMA-targeted therapeutic clinical trial referred to by Curium as ECLIPSE. Both of these collaborations, as well as other collaborations using PYLARIFY are described further under Part I, Item 1. "Business - Strategic Partnerships and Other Revenue – Oncology." in our Annual Report on Form 10-K for the year ended December 31, 2022.

Also, during 2022, the National Comprehensive Cancer Center updated its guidelines and the Society for Nuclear Medicine and Molecular Imaging updated its appropriate use criteria, both noting that PSMA PET imaging agents, including PYLARIFY, can be used for patient selection for PSMA-targeted radioligand therapy.

During 2021, 2022 and the first half of 2023, we hired additional employees to assist us with the commercialization of PYLARIFY, including in Sales, Marketing, Reimbursement, Quality and Medical Affairs.

In addition to our network of PET manufacturing facilities ("PMFs"), we partnered with academic medical centers in the U.S. that have radioisotope-producing cyclotrons who have expressed an interest in manufacturing PYLARIFY.

Our commercial launch required obtaining necessary market access which included, but was not limited to, adequate coding, coverage and payment for PYLARIFY, including coverage from Medicare, Medicaid and other government payors, as well as private payors. We received notification that our Healthcare Procedure Coding System ("HCPCS") code, which enables streamlined billing, went into effect as of January 1, 2022. In addition, effective January 1, 2022, the Centers for Medicare and Medicaid Services ("CMS") granted Transitional Pass-Through Payment Status ("TPT Status") in the hospital outpatient setting for PYLARIFY, enabling traditional Medicare to provide an incremental payment for PET/CT scans performed with PYLARIFY in that setting. TPT Status for PYLARIFY could expire on December 31, 2024.

The successful growth of PYLARIFY is dependent on our ability to sustain PYLARIFY as the leading PSMA PET imaging agent in an increasingly competitive marketplace. PYLARIFY's competition includes two commercially available gallium-68-based PSMA imaging agents, a recently approved fluorine-18-based PSMA imaging agent, and other non-PSMA-based imaging agents, commonly referred to as conventional imaging.

In connection with the acquisition of Progenics in June 2020, we issued CVRs tied to the financial performance of PYLARIFY. Each CVR entitled its holder to receive a pro rata share of aggregate cash payments equal to 40% of U.S. net sales generated by PYLARIFY in 2022 and 2023 in excess of \$100.0 million and \$150.0 million, respectively, but not to exceed, in the aggregate, 19.9% of the total consideration we paid in the acquisition of Progenics. Based on our 2022 PYLARIFY net sales, we determined that the aggregate payment obligation under the CVRs was \$99.6 million, which was the maximum amount payable under the CVRs. We paid \$99.6 million to the CVR holders during May 2023 in full satisfaction of our obligations under the CVRs.

PYLARIFY AI Use

During 2021, we announced that EXINI was granted 510(k) clearance by the FDA in the U.S. and received European Conformity Marking ("CE marking") in Europe for aPROMISE. We commercially launched aPROMISE under the name PYLARIFY AI in the U.S. in November 2021. During the second quarter of 2022, we received a new 510(k) clearance for an updated version of our PYLARIFY AI platform.

PYLARIFY AI is artificial intelligence medical device software developed to assist with the reading and quantification of PYLARIFY scans. The technology automatically analyzes a PSMA PET/CT image to segment anatomical regions – 51 bones and 12 soft tissue organs. This image segmentation enables automated localization, detection and quantification of potential PSMA-avid lesions in a PSMA PET/CT image, which data is then incorporated into the reporting system used by physicians.

Continued Growth of DEFINITY

We believe we will be able to increase use of DEFINITY through continued education of physicians and healthcare providers about the benefits of ultrasound enhancing agents in suboptimal echocardiograms. The U.S. market currently has three echocardiography ultrasound enhancing agents approved by the FDA; we estimate that DEFINITY will continue to hold at least an 80% market share.

As we continue to expand our microbubble franchise, our activities include:

- *Patents* We continue to actively pursue additional patents in connection with DEFINITY and DEFINITY RT, both in the U.S. and internationally. In the U.S. for DEFINITY, we have six Orange Book-listed method of use patents, one of which expires in 2035 and five of which expire in 2037, as well as additional manufacturing patents that are not Orange Book-listed expiring in 2037.
- DEFINITY RT The formulation of DEFINITY that we have branded as DEFINITY RT became commercially available in the fourth quarter of 2021. DEFINITY RT allows both storage and shipment at room temperature and provides clinicians an additional choice and allows for greater utility of this formulation in broader clinical settings. Given its physical characteristics, we believe DEFINITY RT is also well-suited for inclusion in kits requiring microbubbles for other indications and applications (including in kits developed by third parties of the type described in the paragraph entitled Microbubble Franchise below).
- VIALMIX RFID DEFINITY is activated through the use of medical devices branded as VIALMIX and VIALMIX RFID. The activation rate and
 time are controlled by VIALMIX RFID through the use of radio-frequency identification technology ("RFID") to ensure reproducible activation
 of DEFINITY. The RFID tag, which is affixed to the vial label, enables the DEFINITY vial to be appropriately activated with the VIALMIX RFID
 activation device.

Expansion of Strategic Partnerships and Other Revenue

We continue to seek ways to further increase the overall value of our portfolio of products and product candidates. We are evaluating a number of different opportunities to collaborate, in-license or acquire additional products, product candidates, businesses and technologies to drive our future growth. In particular, we are focused on late-stage radiopharmaceutical diagnostic and therapeutic product opportunities in oncology and other strategic areas that will complement our existing portfolio.

Our Strategic Partnerships and Other Revenue category includes our Strategic Partnerships, Pharma Services and Digital Solutions businesses and is focused on enabling precision medicine with biomarkers and digital solutions.

- Strategic Partnerships We seek to monetize our assets through our Strategic Partnerships business, by optimizing core assets geographically and by driving value through non-core assets. For example, with respect to PYLARIFY, we have licensed the development and commercialization rights of that imaging agent in Europe to Curium. Similarly, we licensed the commercialization rights for RELISTOR to Bausch and the commercialization rights for flurpiridaz fluorine-18 to GE Healthcare Limited.
- *Pharma Services* We use our Pharma Services business to offer our Biomarkers and Microbubble Platforms to pharmaceutical and start-up companies to support their research and development of therapeutic drugs and devices. The strategic goal of our Pharma Services business is to gain early access to innovation, de-risk the development, data generation and co-funding of our pipeline through collaborations, embed our technologies in the clinical ecosystem and establish the clinical utility of product candidates and research tools in our pipeline. Our Biomarkers are intended to support patient selection and the monitoring of disease progression, primarily with respect to different oncological diseases. For example, piflufolastat F 18 is currently being used by Curium, Novartis, POINT and Regeneron in those companies' prostate cancer therapeutic drug development programs. Our acquisition of Cerveau in February 2023 added MK-6240 to our biomarker portfolio. MK-6240 is currently being used in more than sixty academic and industry clinical trials for several Alzheimer's disease therapeutic candidates being developed by more than sixteen pharmaceutical companies. Most recently, in collaboration with Ratio Therapeutics, we initiated a Phase I study for LNTH-1363S to evaluate the pharmokinetics, biodistribution and radiation dosimetry in adult health volunteers. LNTH-1363S is our novel fibroblast activation protein, alpha targeted, copper-64 labeled PET imaging agent candidate.

In addition, we continue to expand our Microbubble Platform. In December 2022, we announced a strategic collaboration with SonoThera, Inc. ("SonoThera"), which will use our microbubbles in combination with SonoThera's ultrasound-guided, non-viral, gene therapy platform and treatments. We also have additional collaborations with other partners that are generally designed to include our microbubble as part of a kit used with our partner's medical device for therapeutic applications. In these collaborations, our microbubble is intended to be used as a vehicle to deliver a therapeutic drug.

• Digital Solutions – Our Digital Solutions are designed to enhance imaging value and the throughput, reproducibility and reliability of image analysis, as well as to inform treatment selection and response to therapy. Our digital solutions include aPROMISE and aBSI (as defined below). aPROMISE, which is currently sold as PYLARIFY AI in the U.S., is artificial intelligence medical device software that is designed to allow healthcare professionals and researchers to perform standardized quantitative assessment of PSMA PET/CT images in prostate cancer, including those images obtained by using

PYLARIFY. Automated Bone Scan Index ("aBSI") automatically calculates the disease burden of prostate cancer by detecting and classifying bone scan tracer uptakes as metastatic or benign lesions using an artificial neural network. aBSI is FDA cleared and CE marked. The software is currently used as one of the correlative objectives of the DORA trial, an open-labeled, randomized, Phase III study of docetaxel versus docetaxel in combination with radium-223 (Ra-223) in subjects with metastatic castration-resistant prostate cancer. We offer our Digital Solutions to HCPs for clinical use and to pharmaceutical companies for development purposes, and in some cases, we also obtain clinical imaging data that we may use to further develop artificial intelligence solutions.

Global Mo-99 Supply

We currently have Mo-99 supply agreements with Institute for Radioelements ("IRE"), running through December 31, 2023, with auto-renewal provisions that are terminable upon notice of non-renewal, and with NTP Radioisotopes ("NTP"), acting for itself and on behalf of its subcontractor, the Australian Nuclear Science and Technology Organisation ("ANSTO"), running through December 31, 2024.

Although we have the most globally diverse Mo-99 supply chain with IRE in Belgium, NTP in South Africa, and ANSTO in Australia, we still face supplier and logistical challenges in our Mo-99 supply chain. When one supplier experiences outages, we generally rely on Mo-99 supply from the other suppliers to limit the impact of the outages. We believe we effectively manage these various supply chain challenges, but depending on reactor and processor schedules and operations, at times we have not been able to fill some or all of the demand for our TechneLite generators on certain manufacturing days. A prolonged disruption of service from one of our three Mo-99 processing sites or one of their main Mo-99-producing reactors could have a negative effect on our business, results of operations, financial condition and cash flows.

To augment our current supply of Mo-99, we have a strategic arrangement with SHINE Medical Technologies LLC ("SHINE") for the future supply of Mo-99. Under the terms of the supply agreement, entered into in November 2014, SHINE will provide Mo-99 produced using its proprietary LEU-solution technology for use in our TechneLite generators once SHINE's facility becomes operational and receives all necessary regulatory approvals, which SHINE now estimates will occur in 2024. The term of this arrangement provides for three years of supply of Mo-99. However, we cannot provide assurance that SHINE will be able to produce commercial quantities of Mo-99 for our business, or that SHINE, together with our current suppliers, will be able to deliver a sufficient quantity of Mo-99 to meet our needs.

Inventory Supply

We obtain a substantial portion of our Precision Diagnostics from a third party supplier. Jubilant HollisterStier ("JHS") is currently a significant supplier of DEFINITY and our sole source manufacturer of select other products. On February 23, 2022, LMI entered into a Manufacturing and Supply Agreement (the "MSA") with JHS, effective as of February 23, 2022, pursuant to which JHS will manufacture, and LMI will purchase, DEFINITY and select other products. The new MSA supersedes all of the prior agreements of the parties. The initial term of the MSA runs through December 31, 2027 and can be further extended by mutual agreement of the parties. The MSA requires LMI to purchase from JHS specified percentages of its total requirements for DEFINITY, as well as specified quantities of select other products, each year during the contract term, except to the extent that LMI has the right to withhold payment for defective product and cancel orders for untimely delivery without penalty. Either party can terminate the MSA upon the occurrence of certain events, including the material breach or bankruptcy of the other party. In addition to JHS, we rely on Samsung BioLogics as our sole source manufacturer of DEFINITY RT.

In 2021, we completed the construction of a specialized in-house manufacturing facility at our North Billerica campus to produce DEFINITY. On February 22, 2022, we received FDA approval of our supplemental new drug application authorizing commercial manufacturing of DEFINITY at our new facility. We believe this facility will allow us to better manage DEFINITY manufacturing and inventory, reduce our costs in a potentially more price competitive environment, and provide us with supply chain redundancy.

Radiopharmaceuticals are decaying radioisotopes with half-lives ranging from a few hours to several days. Radiopharmaceutical finished goods, such as doses of PYLARIFY, AZEDRA and TechneLite generators cannot be kept in inventory because of their limited shelf lives and are subject to just-in-time manufacturing, processing and distribution, which takes place at our facilities in North Billerica, Massachusetts and Somerset, New Jersey, as well as at our PMF partner manufacturing facilities across the U.S.

Research and Development Expenses

To ensure we remain the leading radiopharmaceutical-focused company in our industry, we have historically made and will continue to make substantial investments in new product development and lifecycle management for existing products, including:

• For PYLARIFY, our development of PYLARIFY resulted in approval by the FDA in May 2021.

- For PYLARIFY AI, our development of PYLARIFY AI resulted in a 510(k) clearance granted by the FDA in the third quarter of 2021 and an additional 510(k) clearance granted during the second quarter of 2022.
- For PNT2002 and PNT2003, we were granted a license to exclusive worldwide rights (excluding certain countries) for \$260.0 million in upfront payments during the fourth quarter of 2022 and will potentially make additional payments as described below.
- For 1095, our PSMA-targeted iodine-131-labeled small molecule product candidate, we enrolled the last patient in our ARROW Phase 2 study during the second quarter of 2022. Patients in this study will be followed for one year after their first treatment for all efficacy endpoints and survival and safety data will be collected for an additional year.
- We are also developing additional lifecycle management opportunities for PYLARIFY, including preparing a clinical study in favorable intermediate-risk prostate cancer patients, and exploring other possibilities beyond prostate cancer.

PNT2002

Under the terms of the PNT2002 License Agreement, Lantheus Two, LLC paid POINT an upfront cash payment of \$250.0 million, and could pay up to an additional \$281.0 million in milestone payments upon the achievement of specified U.S. and ex-U.S. regulatory milestones related to PNT2002. POINT is also eligible to receive up to \$1.3 billion in sales milestone payments upon the achievement of specified annual sales thresholds of PNT2002. In addition, after Lantheus Two achieves \$500.0 million in cumulative gross profit, POINT is eligible to receive royalty payments of twenty percent of net sales of PNT2002. Prior to achieving that financial recoupment threshold, POINT is eligible to receive royalty payments of twenty percent on that portion of annual net sales of PNT2002 that generate annual gross profit in excess of specified levels.

PNT2003

Under the terms of the PNT2003 License Agreement, Lantheus Three, LLC paid POINT an upfront cash payment of \$10.0 million, and could pay up to an additional \$34.5 million in milestone payments upon the achievement of specified U.S. and ex-U.S. regulatory milestones related to PNT2003. POINT is also eligible to receive up to \$275.0 million in sales milestone payments upon the achievement of specified annual sales thresholds of PNT2003. In addition, POINT is eligible to receive royalty payments of fifteen percent of net sales of PNT2003.

Our investments in these additional clinical activities and lifecycle management opportunities will increase our operating expenses and impact our results of operations and cash flow, and we can give no assurances as to whether any of our clinical development candidates or lifecycle management opportunities will be successful.

Results of Operations

The following is a summary of our consolidated results of operations:

		Three Months Ended June 30,								Six Months Ended June 30,						
(in thousands)	2023		2022	C	Change \$	Change %		2023		2022	(Change \$	Change %			
Revenues	\$ 321,700	\$	223,723	\$	97,977	43.8 %	\$	622,484	\$	432,603	\$	189,881	43.9 %			
Cost of goods sold	119,053		85,694		33,359	38.9 %		342,761		165,504		177,257	107.1 %			
Gross profit	202,647		138,029		64,618	46.8 %		279,723		267,099		12,624	4.7 %			
Operating expenses												,				
Sales and marketing	36,456		27,492		8,964	32.6 %		69,073		47,846		21,227	44.4 %			
General and administrative	26,151		32,598		(6,447)	(19.8)%		49,422		70,186		(20,764)	(29.6)%			
Research and development	15,901		14,735		1,166	7.9 %		46,433		26,938		19,495	72.4 %			
Total operating expenses	78,508		74,825		3,683	4.9 %		164,928		144,970		19,958	13.8 %			
Operating income	124,139		63,204		60,935	96.4 %		114,795		122,129		(7,334)	(6.0)%			
Interest expense	4,933		1,469		3,464	235.8 %		9,924		2,978		6,946	233.2 %			
Other income	(4,482))	(310)		(4,172)	1345.8 %		(7,713)		(795)		(6,918)	870.2 %			
Income before income taxes	123,688		62,045		61,643	99.4 %		112,584		119,946		(7,362)	(6.1)%			
Income tax expense	29,557		18,987		10,570	55.7 %		21,260		33,926		(12,666)	(37.3)%			
Net income	\$ 94,131	\$	43,058	\$	51,073	118.6 %	\$	91,324	\$	86,020	\$	5,304	6.2 %			

Comparison of the Periods Ended June 30, 2023 and 2022

Revenues

We classify our revenues into three product categories: Radiopharmaceutical Oncology, Precision Diagnostics, and Strategic Partnerships and Other Revenue. Radiopharmaceutical Oncology includes PYLARIFY and AZEDRA. Precision Diagnostics includes DEFINITY, TechneLite and other diagnostic imaging products. Strategic Partnerships and Other Revenue includes out-licensing arrangements, partnerships that focus on facilitating precision medicine through the use of biomarkers, digital solutions and radiotherapeutic platforms, and on our other products, such as RELISTOR.

Revenues are summarized by product category on a net basis as follows:

	Three Months Ended June 30,							Six Months Ended June 30,						
(in thousands)		2023		2022	C	hange \$	Change %	2023		2022	(Change \$	Change %	
PYLARIFY	\$	210,522	\$	130,232	\$	80,290	61.7 %	\$ 405,992	\$	223,009	\$	182,983	82.1 %	
Other radiopharmaceutical oncology		818		928		(110)	(11.9)%	1,535		2,255		(720)	(31.9)%	
Total radiopharmaceutical oncology		211,340		131,160		80,180	61.1 %	407,527		225,264		182,263	80.9 %	
DEFINITY		70,529		62,306		8,223	13.2 %	139,353		120,634		18,719	15.5 %	
TechneLite		21,594		19,440		2,154	11.1 %	42,580		42,045		535	1.3 %	
Other precision diagnostics		5,454		5,363		91	1.7 %	11,261		10,628		633	6.0 %	
Total precision diagnostics		97,577		87,109		10,468	12.0 %	193,194		173,307		19,887	11.5 %	
Strategic partnerships and other revenue		12,783		5,454		7,329	134.4 %	21,763		34,032		(12,269)	(36.1)%	
Total revenues	\$	321,700	\$	223,723	\$	97,977	43.8 %	\$ 622,484	\$	432,603	\$	189,881	43.9 %	

The increase in revenues for the three months ended June 30, 2023, is primarily driven by an increase in PYLARIFY, DEFINITY and TechneLite sales volume, as well as revenue generated by Cerveau, as recorded in Strategic Partnerships and Other Revenue.

The increase in revenues for the six months ended June 30, 2023, is primarily driven by an increase in PYLARIFY and DEFINITY sales volume, as well as revenue generated by Cerveau, as recorded in Strategic Partnerships and Other Revenue. The increase is offset, in part, by a decrease in Strategic Partnerships and Other Revenue due to the revenue recognized from the Novartis licensing agreement in the prior year.

Rebates and Allowances

Estimates for rebates and allowances represent our estimated obligations under contractual arrangements with customers and other third parties. Rebate accruals and allowances are recorded in the same period the related revenue is recognized, resulting in a reduction to revenue and the establishment of a liability which is included in accrued expenses. These rebates and allowances result from performance-based offers that are primarily based on attaining contractually specified sales volumes and growth, Medicaid rebate programs for our products, administrative fees of group purchasing organizations and certain distributor related commissions. The calculation of the accrual for these rebates and allowances is based on an estimate of the third-party's expected purchases and the resulting applicable contractual rebate to be earned over a contractual period.

An analysis of the amount of, and change in, reserves is summarized as follows:

(in thousands)	oates and lowances
Balance, January 1, 2023	\$ 13,399
Provision related to current period revenues	13,606
Payments or credits made during the period	(15,918)
Balance, June 30, 2023	\$ 11,087

Gross Profit

The increase in gross profit for the three months ended June 30, 2023, as compared to the prior year period, is primarily due to increased PYLARIFY and DEFINITY sales volumes, partially offset by the impairment related to a certain asset group.

The increase in gross profit for the six months ended June 30, 2023, as compared to the prior year period, is primarily due to the increased PYLARIFY and DEFINITY sales volumes, partially offset by the impairment of the AZEDRA currently marketed intangible asset, the Novartis licensing settlement in the prior year, amortization of Cerveau intangible assets and the impairment related to a certain asset group.

Sales and Marketing

Sales and marketing expenses consist primarily of salaries and other related costs for personnel in field sales, marketing and customer service functions. Other costs in sales and marketing expenses include the development and printing of advertising and promotional material, professional services, market research and sales meetings.

Sales and marketing expenses increased \$9.0 million and \$21.2 million for the three and six months ended June 30, 2023, respectively, as compared to the prior year period. This was primarily driven by the continued promotion activities of PYLARIFY and increased employee-related costs.

General and Administrative

General and administrative expenses consist of salaries and other related costs for personnel in executive, finance, legal, information technology and human resource functions. Other costs included in general and administrative expenses are professional fees for information technology services, external legal fees, consulting and accounting services as well as bad debt expense, certain facility and insurance costs, including director and officer liability insurance.

General and administrative expenses decreased \$6.4 million for the three months ended June 30, 2023 compared to the prior period. This was primarily driven by a \$16.0 million net reduction for the fair value adjustments to the contingent asset and liabilities in the second quarter of 2023 (refer to Note 4, "Fair Value of Financial Instruments", for further details on contingent consideration liabilities, including CVRs). This decrease is partially offset by increased employee-related costs, professional fees and investment in technology.

General and administrative expenses decreased \$20.8 million for the six months ended June 30, 2023 compared to the prior period. This was primarily driven by a \$35.8 million net reduction for the fair value adjustments to the contingent asset and liabilities in the first half of 2023 (refer to Note 4, "Fair Value of Financial Instruments", for further details on contingent consideration liabilities, including CVRs). This decrease is partially offset by increased employee-related costs, professional fees and investment in technology.

Research and Development

Research and development expenses relate primarily to the development of new products to add to our portfolio and costs related to our medical affairs, medical information and regulatory functions.

Research and development expenses increased \$1.2 million for the three months ended June 30, 2023 as compared to the prior year period. This was primarily driven by increased employee-related costs.

Research and development expenses increased \$19.5 million for the six months ended June 30, 2023 as compared to the prior year period. This was primarily driven by the AZEDRA IPR&D asset impairment loss of \$15.6 million and higher employee-related costs driven by an increased headcount. These increases were offset, in part, by lower clinical expenses related to our ARROW Phase 2 study.

Interest Expense

Interest expense increased by approximately \$6.9 million for the six months ended June 30, 2023 as compared to the prior year period due to the issuance of the Notes on December 8, 2022, which was partially offset by the interest rate swap agreements from the prior year.

Income Tax Expense

Our effective tax rate for each reporting period is presented as follows:

Six Months En	ded
Iuma 20	

	Jun	June 30,		
	2023	2022		
Effective tax rate	18.9%	28.3%		

Our effective tax rate for the six months ended June 30, 2023 differs from the U.S. statutory rate of 21% primarily due to the income tax benefit associated with stock compensation deductions, partially offset by state income taxes.

The decrease in the effective income tax rate for the six months ended June 30, 2023 is primarily due to the decrease in non-deductible contingency reserve expense and the increase in our stock compensation deductions.

Liquidity and Capital Resources

Cash Flows

The following table provides information regarding our cash flows:

	Six Months Ended June 30,			
<u>(in thousands)</u>		2023		2022
Net cash provided by operating activities	\$	76,234	\$	82,861
Net cash used in investing activities	\$	(65,210)	\$	(5,733)
Net cash used in financing activities	\$	(12,720)	\$	(4,190)

Net Cash Provided by Operating Activities

Net cash provided by operating activities of \$76.2 million in the six months ended June 30, 2023 was primarily comprised of net income adjusted for the net effect of non-cash items such as impairment of long-lived assets, depreciation, amortization and accretion expense and stock-based compensation expense. The primary working capital sources of cash were the timing of payments to large vendors. The primary working capital uses of cash were a decrease to accruals related to the CVR payment, an increase in trade receivables associated primarily with the increase in PYLARIFY revenues, and an increase in inventory related to the timing of batch processes.

Net cash provided by operating activities of \$82.9 million in the six months ended June 30, 2022 was primarily comprised of net income adjusted for the net effect of non-cash items such as the change in fair value of contingent assets and liabilities of \$26.9 million (refer to Note 4, "Fair Value of Financial Instruments", for further details on contingent consideration liabilities, including CVRs). The primary working capital sources of cash were the timing of payments to large vendors. The primary working capital uses of cash were an increase in trade receivables associated primarily with the increase in PYLARIFY revenues as well as the timing of inventory purchases.

Net Cash Used in Investing Activities

Net cash used in investing activities during the six months ended June 30, 2023 was due to \$45.3 million for our asset acquisition of Cerveau and \$19.9 million of capital expenditures.

Net cash used in investing activities during the six months ended June 30, 2022 was primarily due to \$7.5 million of capital expenditures offset by cash proceeds of \$1.8 million received from the sale of our Puerto Rico subsidiary.

Net Cash Used in Financing Activities

Net cash used in financing activities during the six months ended June 30, 2023 is primarily attributable to the payments for minimum statutory tax withholding related to net share settlement of equity awards of \$12.6 million and the CVR initial valuation as of June 30, 2020 of \$3.7 million offset by proceeds of \$3.2 million from stock option exercises.

Net cash used in financing activities during the six months ended June 30, 2022 is primarily attributable to the payments on long-term debt and other borrowings of \$5.2 million related to the 2019 Term Facility and payments for minimum statutory tax withholding related to net share settlement of equity awards of \$6.3 million offset by proceeds of \$6.8 million from stock option exercises.

External Sources of Liquidity

In December 2022, we voluntarily repaid our 2019 \$200.0 million five-year term loan facility. In addition, we replaced our \$200.0 million revolving facility with the 2022 Revolving Facility. The terms of the 2022 Revolving Facility are set forth in the Credit Agreement, dated as of December 2, 2022, by and among us, the lenders from time to time party thereto and Citizens Bank, N.A., as administrative agent and collateral agent (the "2022 Credit Agreement"). We have the right to request an increase to the 2022 Revolver Facility or request the establishment of one or more new incremental term loan facilities, in an aggregate principal amount of up to \$335 million or consolidated EBITDA for the four consecutive fiscal quarters most recently ended, plus additional amounts, in certain circumstances.

Under the terms of the 2022 Revolving Facility, the lenders thereunder agreed to extend credit to us from time to time until December 2, 2027 consisting of revolving loans in an aggregate principal amount not to exceed \$350.0 million at any time. The 2022 Revolving Facility includes a \$20.0 million sub-facility for the issuance of Letters of Credit. The 2022 Revolving Facility includes a

\$10.0 million sub-facility for Swingline Loans. The Letters of Credit, Swingline Loans and the borrowings under the 2022 Revolving Facility are expected to be used for working capital and other general corporate purposes.

Please refer to Note 13, "Long-Term Debt, Net, and Other Borrowings" for further details on the 2022 Facility.

As of June 30, 2023, we were in compliance with all financial and other covenants under the 2022 Credit Agreement.

On December 8, 2022, we issued \$575.0 million in aggregate principal amount of the Notes, which includes \$75.0 million in aggregate principal amount of Notes sold pursuant to the full exercise of the initial purchasers' option to purchase additional Notes. The Notes were issued under the Indenture. The net proceeds from the issuance of the Notes were approximately \$557.8 million, after deducting the initial purchasers' discounts and offering expenses payable by us.

Our ability to fund our future capital needs will be affected by our ability to continue to generate cash from operations and may be affected by our ability to access the capital markets, money markets or other sources of funding, as well as the capacity and terms of our financing arrangements.

We may from time to time repurchase or otherwise retire our debt and take other steps to reduce our debt or otherwise improve our balance sheet. These actions may include prepayments of our term loans or other retirements or refinancing of outstanding debt, privately negotiated transactions or otherwise. The amount of debt that may be retired, if any, could be material and would be decided at the sole discretion of our Board of Directors and will depend on market conditions, our cash position and other considerations.

Funding Requirements

Our future capital requirements will depend on many factors, including:

- The level of product sales and the pricing environment of our currently marketed products, particularly PYLARIFY and DEFINITY, as well as any additional products that we may market in the future;
- Revenue mix shifts and associated volume and selling price changes that could result from additional competition or changes in customers' product demand;
- The continued costs of the ongoing commercialization of PYLARIFY and PYLARIFY AI;
- Our investment in the further clinical development and commercialization of products and development candidates, including PNT2002, PNT2003, 1095 and MK-6240;
- The costs of acquiring or in-licensing, developing, obtaining regulatory approval for, and commercializing, new products, businesses or technologies, including any potential related milestone or royalty payments, together with the costs of pursuing opportunities that are not eventually consummated;
- The costs of investing in our facilities, equipment and technology infrastructure;
- The costs and timing of establishing or amending manufacturing and supply arrangements for commercial supplies of our products and raw materials and components;
- Our ability to have products manufactured and released from manufacturing sites in a timely manner in the future, or to manufacture products at our in-house manufacturing facilities in amounts sufficient to meet our supply needs;
- The costs of further commercialization of our existing products, particularly in international markets, including product marketing, sales and distribution and whether we obtain local partners to help share such commercialization costs;
- The legal costs relating to maintaining, expanding and enforcing our intellectual property portfolio, pursuing insurance or other claims and defending against product liability, regulatory compliance, intellectual property or other claims;
- · The cost of interest on any additional borrowings which we may incur under our financing arrangements; and
- The impact of sustained inflation on our costs of goods sold and operating expenses.

Disruption in our financial performance could occur if we experience significant adverse changes in product or customer mix, broad economic downturns, sustained inflation, adverse industry or company conditions or catastrophic external events, including pandemics such as COVID-19, natural disasters and political or military conflict. If we experience one or more of these events in the future, we may be required to further implement expense reductions, such as a delay or elimination of discretionary spending in all functional areas, as well as scaling back select operating and strategic initiatives.

If our capital resources become insufficient to meet our future capital requirements, we would need to finance our cash needs through public or private equity offerings, debt financings, assets securitizations, sale-leasebacks or other financing or strategic alternatives, to the extent such transactions are permissible under the covenants of our 2022 Credit Agreement. Additional equity or debt financing, or other transactions, may not be available on acceptable terms, if at all. If any of these transactions require an amendment or waiver under the covenants in our 2022 Credit Agreement, which could result in additional expenses associated with obtaining the amendment or waiver, we will seek to obtain such a waiver to remain in compliance with those covenants. However, we cannot provide assurance that such an amendment or waiver would be granted, or that additional capital will be available on acceptable terms, if at all.

At June 30, 2023, our only current committed external source of funds is our borrowing availability under our 2022 Revolving Facility. We had \$414.1 million of cash and cash equivalents as of June 30, 2023. Our 2022 Revolving Facility contains a number of affirmative, negative, reporting and financial covenants, in each case subject to certain exceptions and materiality thresholds. Incremental borrowings under the 2022 Revolving Facility may affect our ability to comply with the covenants including the financial covenants restricting consolidated net leverage and interest coverage. Accordingly, we may be limited in utilizing the full amount of our 2022 Revolving Facility as a source of liquidity.

Based on our current operating plans, we believe our balance of cash and cash equivalents, which totaled \$414.1 million as of June 30, 2023, along with cash generated by ongoing operations and continued access to our 2022 Revolving Facility, will be sufficient to satisfy our cash requirements over the next twelve months and beyond.

Critical Accounting Policies and Estimates

The discussion and analysis of our financial condition and results of operations are based on our condensed consolidated financial statements, which have been prepared in accordance with U.S. GAAP. The preparation of these condensed consolidated financial statements requires us to make estimates and judgments that affect our reported assets and liabilities, revenues and expenses, and other financial information. Actual results may differ materially from these estimates under different assumptions and conditions. In addition, our reported financial condition and results of operations could vary due to a change in the application of a particular accounting standard.

There have been no significant changes to our critical accounting policies or in the underlying accounting assumptions and estimates used in such policies in the six months ended June 30, 2023. For further information, refer to our summary of significant accounting policies and estimates in our Annual Report on Form 10-K filed for the year ended December 31, 2022.

Off-Balance Sheet Arrangements

We are required to provide the Massachusetts Department of Public Health and New Jersey Department of Environmental Protection financial assurance demonstrating our ability to fund the decommissioning of our North Billerica, Massachusetts and Somerset, New Jersey production facilities, respectively, upon closure. We have provided this financial assurance in the form of a \$30.3 million surety bond.

Since inception, we have not engaged in any other off-balance sheet arrangements, including structured finance, special purpose entities or variable interest entities.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

For quantitative and qualitative disclosures about market risk, see Part II, Item 7A. "Quantitative and Qualitative Disclosures About Market Risk," of our Annual Report on Form 10-K for the year ended December 31, 2022. Our exposures to market risk have not changed materially since December 31, 2022.

Item 4. Controls and Procedures

Disclosure Controls and Procedures

The Company's management, with the participation of the Company's Chief Executive Officer ("CEO") and Chief Financial Officer ("CFO"), its principal executive officer and principal financial officer, respectively, has evaluated the effectiveness of the Company's disclosure controls and procedures as defined in Rule 13a-15(e) and 15d-15(e) of the Exchange Act. Based on that evaluation, the Company's CEO and CFO concluded that the Company's disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) were effective as of the period covered by this report.

Changes in Internal Controls Over Financial Reporting

There were no changes in our internal control over financial reporting during the quarter ended June 30, 2023 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

We are continually monitoring and assessing the pandemic status and geopolitical environment to determine any potential impact on the design and operating effectiveness of our internal controls over financial reporting.

PART II. OTHER INFORMATION

Item 1. Legal Proceedings

Information with respect to certain legal proceedings is included in Note 18, "Commitments and Contingencies", to the condensed consolidated financial statements contained in Part I, Item 1. Financial Statements of this Quarterly Report on Form 10-Q and is incorporated herein by reference.

Item 1A. Risk Factors

There have been no material changes to the risk factors set forth in our Annual Report on Form 10-K for the year ended December 31, 2022.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Repurchases

The following table presents information with respect to purchases of common stock we made during the three months ended June 30, 2023. In December 2022, in connection with the issuance of the Notes, our Board of Directors authorized the repurchase of up to \$150.0 million in aggregate amount of our common stock under certain circumstances, of which \$75.0 million were repurchased prior to the three months ended June 30, 2023. The 2015 Equity Incentive Plan, adopted by us on June 24, 2015, as amended on April 26, 2016 and as further amended on April 27, 2017, April 24, 2019, April 28, 2021 and April 28, 2022 (the "2015 Plan"), provides for the withholding of shares to satisfy minimum statutory tax withholding obligations. It does not specify a maximum number of shares that can be withheld for this purpose. The shares of common stock withheld to satisfy minimum tax withholding obligations may be deemed to be "issuer purchases" of shares that are required to be disclosed pursuant to this Item 2.

Period	Total Number of Shares Purchased	Avo	erage Price Paid per Share	Total Number of Shares Purchased as Part of Publicly Announced Programs	Approximate Dollar Value of Shares that May Yet Be Purchased Under the Program
April 2023*	11,215	\$	89.16	0	\$75.0 million
May 2023*	3,997	\$	98.11	0	\$75.0 million
June 2023*	1,272	\$	89.81	0	\$75.0 million
Total	16,484			0	\$75.0 million

Reflects shares withheld to satisfy minimum statutory tax withholding amounts due from employees related to the receipt of stock which
resulted from the exercise or vesting of equity awards.

Dividend Policy

We did not declare or pay any dividends, and we do not currently intend to pay dividends in the foreseeable future. We currently expect to retain future earnings, if any, for the foreseeable future, to finance the growth and development of our business and to repay indebtedness. Our ability to pay dividends is restricted by our financing arrangements. See Part I, Item 2. "Management's Discussion and Analysis of Financial Condition and Results of Operations-Liquidity and Capital Resources-External Sources of Liquidity" for further information.

Item 3. Defaults Upon Senior Securities

None.

Item 4. Mine Safety Disclosures

Not applicable.

Item 5. Other Information

Rule 10b5-1 Trading Plans

On June 15, 2023, Brian Markison, a member of our Board of Directors, entered into a trading plan intended to satisfy the affirmative defense conditions of Rule 10b5-1(c) under the Exchange Act (a "10b5-1 Plan"), providing for the potential sale of up to 10,000 shares of our common stock between September 25, 2023 and December 29, 2023.

On June 2, 2023, Gérard Ber, a member of our Board of Directors, entered into a 10b5-1 Plan providing for the potential sale of up to 15,500 shares of our common stock between September 5, 2023 and February 29, 2024.

On May 17, 2023, Gary Pruden, a member of our Board of Directors, entered into a 10b5-1 Plan providing for the potential sale of up to 12,530 shares of our common stock between August 16, 2023 and July 31, 2024.

On May 16, 2023, Heinz Mäusli, a member of our Board of Directors, entered into a 10b5-1 Plan providing for the potential sale of up to 21,549 shares of our common stock, including shares obtained from the exercise of vested stock options covered by the 10b5-1 Plan, between August 15, 2023 and December 29, 2023.

On May 9, 2023, Julie McHugh, a member of our Board of Directors, entered into a 10b5-1 Plan providing for the potential sale of up to 2,180 shares of our common stock between August 8, 2023 and December 29, 2023.

On May 8, 2023, Samuel Leno, a member of our Board of Directors, entered into a 10b5-1 Plan providing for the potential sale of up to 26,479 shares of our common stock, including shares obtained from the exercise of vested stock options covered by the 10b5-1 Plan, between August 9, 2023 and August 31. 2023.

On May 22, 2023, Paul Blanchfield, our President, entered into a 10b5-1 Plan providing for the potential sale of up to 1,908 shares of our common stock between August 21, 2023 and November 16, 2023, plus up to 33,820 shares of our common stock that may be issued pursuant to the vesting of an equity award on March 7, 2024. Pursuant to the 10b5-1 Plan, Mr. Blanchfield will also make a gift of 500 shares of common stock on March 7, 2024.

Item 6. Exhibits

		INCORPORATED BY REFERENCE			
EXHIBIT NUMBER	DESCRIPTION OF EXHIBITS	FORM	FILE NUMBER	EXHIBIT	FILING DATE
10.1*†	Office Lease by and between LMI and 201 Burlington Road Owner, LLC dated February 14, 2022 (the "Lease"); as amended by the First Amendment To Lease dated May 4, 2023				
10.2+	Lantheus Holdings, Inc. 2023 Employee Stock Purchase Plan	8-K	001-36569	10.1	May 1, 2023
31.1*	Certification of Chief Executive Officer pursuant to Exchange Act Rule 13a-14(a).				
31.2*	Certification of Chief Financial Officer pursuant to Exchange Act Rule 13a-14(a).				
32.1**	Certification pursuant to 18 U.S.C. Section 1350.				
101.INS*	Inline XBRL Instance Document				
101.SCH*	Inline XBRL Taxonomy Extension Schema Document				
101.CAL*	Inline XBRL Taxonomy Extension Calculation Linkbase Document				
101.DEF*	Inline XBRL Taxonomy Extension Definition Linkbase Document				
101.LAB*	Inline XBRL Taxonomy Extension Label Linkbase Document				
101.PRE*	Inline XBRL Taxonomy Extension Presentation Linkbase Document				
104*	Cover Page Interactive Data File (embedded within the Inline XBRL document)				

^{*} Filed herewith.

^{**} Furnished herewith.

[†] Portions of this exhibit have been omitted for confidential treatment pursuant to Item 601(b)(10)(iv) of Regulation S-K.

⁺ Indicates management contract or compensatory plan or arrangements.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

LANTHEUS HOLDINGS, INC.

By: /s/ MARY ANNE HEINO

Mary Anne Heino Name: Chief Executive Officer (Principal Executive Officer) Title:

Date: August 3, 2023

LANTHEUS HOLDINGS, INC.

By: /s/ ROBERT J. MARSHALL, JR.

Name: Robert J. Marshall, Jr.

Chief Financial Officer and Treasurer (Principal Financial Officer) Title:

Date: August 3, 2023 Certain identified information has been omitted from this exhibit because it is (i) not material and (ii) of the type that the registrant treats as private or confidential. [***] indicates the information has been omitted.

FINAL

OFFICE LEASE BY AND BETWEEN

201 BURLINGTON ROAD OWNER, LLC (AS LANDLORD) AND LANTHEUS MEDICAL IMAGING, INC. (AS TENANT)

FOR PREMISES AT 201 BURLINGTON

ROAD

BEDFORD MA

TABLE OF CONTENTS

1.1 Subject 1.2 Exhib	ct Referred To 1 its 4
ARTICLE 2	PREMISES AND TERM 4
2.1 Premi 2.2 TERM 2.3 Extens 2.4 Landle	ses 4 5 sion Option 6 ord's Termination Right 9
	IMPROVEMENTS 9
3.2 Plan <i>A</i> 3.3 Tenan 3.4 Early	mance of Landlord's Work 9 Approval; Final Plans 10 t Delay 10 Access 11 ed Delivery 11
ARTICLE 4	RENT 12
4.2 Additi 4.2.1 R 4.2.2 Po 4.2.3 O 4.2.4 In 4.2.5 U 4.3 Late F	ixed Rent 12 ional Rent 12 eal Estate Taxes. 12 ersonal Property Taxes 14 operating Costs 14 osurance 19 ftilities 20 orange of Rent 21 ity and Restoration Deposit 21
ARTICLE 5	LANDLORD'S COVENANTS 22
5.1.1 H 5.1.2 E 5.1.3 C 5.1.4 E 5.1.5 R 5.1.6 A 5.1.7 C 5.2 INTER 5.3 Outsic 5.4 Landle	native Covenants 22 leat and Air-Conditioning. 22 lectricity 22 leaning; Water. 23 levator; Fire Alarm. 23 epairs. 23 .ccess. 23 ompliance with Law 24 RUPTION 24 de Services 25 ord's Insurance 25 MNIFICATION 25

ARTICLE 6 TENANT'S ADDITIONAL COVENANTS 26

6.1 Affirmative Covenants 26 6.1.1 Perform Obligations 26

ARTICLE 1 REFERENCE DATA 1

6.1.2 Use. 26 6.1.3 Repair and Maintenance. 26 6.1.4 Compliance with Law 26 6.1.5 Indemnification 27 6.1.6 Landlord's Right to Enter 27 6.1.7 Personal Property at Tenant's Risk 27 6.1.8 Payment of Landlord's Cost of Enforcement. 28 6.1.9 Yield Up 28 6.1.10 Rules and Regulations 30 6.1.11 Estoppel Certificate 30 6.1.12 Landlord's Expenses Re Consents 31 6.1.13 Guaranty. 31 6.2 Negative Covenants 31 6.2.1 Assignment and Subletting. 31 6.2.2 Nuisance. 34 6.2.3 Hazardous Wastes and Materials. 34 6.2.4 Floor Load; Heavy Equipment 36 6.2.5 Installation, Alterations or Additions 36 6.2.6 Abandonment 38 6.2.7 Signs 38 6.2.8 Parking and Storage 38
ARTICLE 7 CASUALTY OR TAKING 38
7.1 Termination 387.2 Restoration 397.3 Award 39
ARTICLE 8 DEFAULTS 39
 8.1 Events of Default 39 8.2 Remedies 40 8.3 Remedies Cumulative 41 8.4 Landlord's Right to Cure Defaults 42 8.5 Effect of Waivers of Default 42 8.6 No Waiver, etc 42 8.7 No Accord and Satisfaction 42
ARTICLE 9 RIGHTS OF MORTGAGE HOLDERS 42
9.1 Rights of Mortgage Holders 429.2 Lease Superior or Subordinate to Mortgages 43
ARTICLE 10 MISCELLANEOUS PROVISIONS 44
 10.1 Notices from One Party to the Other 44 10.2 Quiet Enjoyment 44 10.3 Lease not to be Recorded 44 10.4 Limitation of Landlord's Liability 44 10.5 Force Majeure 45 10.6 Landlord's Default 45

- 10.7 Brokerage 45 10.8 Applicable Law and Construction; Merger; Jury Trial 45 10.9 Consents 46 10.10 Authority 46

- 10.11 Execution; Counterparts 46 10.12 OFAC Compliance 46
- 10.13 Confidentiality; No Public Statement 48

OFFICE LEASE

ARTICLE 1 Reference Data

1.1 Subject Referred To

Each reference in this Lease to any of the following subjects shall be construed to incorporate the data stated for that subject in this Section 1.1.

February 14, 2022 Effective Date: Property: The building located at 201 Burlington Road, Bedford, MA (the "Building"), together with the parking areas, landscaping, walkways and other improvements related to the Building, located on the parcel of land described in Exhibit A attached hereto. The Building is a single building comprised of a south portion (the "South Building") and a north portion the "North Building") connected by an enclosed bridge. Premises: A portion of the first floor and the entire second and third floors of the South Building, substantially as shown on Exhibit B attached hereto. Rentable Floor Area of Premises: Approximately 46,526 rentable square feet, consisting of (i) approximately 2,991 rentable square feet on the 1st floor, and (ii) 43,535 rentable square feet on the 2nd and 3rd floors in total.. [***] Original Notice Address of Landlord: Landlord: 201 Burlington Road Owner LLC, a Delaware limited liability company Tenant: Lantheus Medical Imaging, Inc., a Delaware corporation [***] **Original Notice Address of Tenant:** Commencement Date: See Section 2.2 The last day of the Partial Lease Year (as hereinafter **Expiration Date:** defined).

One hundred twenty (120) days after the Effective Date and Landlord's receipt of a building permit for Landlord's Work (as defined below) from the Town of Bedford. Anticipated Delivery Date:

Initial Rent Commencement Date: The Commencement Date.

 1^{st} day of the 10^{th} calendar month of Lease Year 1 (Fixed Rent calculation based on 35,000 rsf). First Rent Adjustment Date:

Second Rent Adjustment Date: 1st day of Lease Year 2 (Fixed Rent calculation based on

35,000 rsf).

Full Rent Commencement Date:

Annual Fixed Rent Rate:

 1^{st} day of the 7^{th} calendar month of Lease Year 2 (Fixed Rent calculation based on the entire Premises).

Initial Rent Commencement Date – end of the 9th calendar

month of Lease Year 1: \$1,001,731.68 (calculation based on 30,000 rsf)

First Rent Adjustment Date – end of Lease Year 1: \$1,168,686.96 (calculation

based on 35,000 rsf)

Second Rent Adjustment Date – end of 6th

month of

Lease Year 2: \$1,203,747.60 (calculation based on

35,000 rsf)

Full Rent Commencement Date -

the end of Lease Year 2: \$1,600,158.84 (calculation

based on 46,526 rsf)

Lease Year 3: \$1,648,163.64

Lease Year 4: \$1,697,608.44

Lease Year 5: \$1,748,536.80

Lease Year 6: \$1,800,992.88

(Annual Fixed Rent Rate continued)

Lease Year 7: \$1,855,022.64

Lease Year 8: \$1,910,673.36

Partial Lease Year: \$1,967,993.52

Initial Rent Commencement Date – end of the

9th calendar

month of Lease Year 1:: \$83,477.64

First Rent Adjustment

Date – end of Lease Year 1: \$97,390.58

Second Rent Adjustment Date – end of the

6th month of

Lease Year 2 \$100,312.30

Full Rent Commencement

Date - end of Lease Year 2: \$133,346.57 Lease Year

3: \$137,346.97

Lease Year 4: \$141,467.37

Lease Year 5: \$145,711.40

Lease Year 6: \$150,082.24

Lease Year 7: \$154,585.22

Lease Year 8: \$159,222.78

Partial Lease Year: \$163,999.46

[***]

Operating Costs for calendar year 2022 (subject to the

terms of Section 4.2.3)

Monthly Fixed Rent Rate:

Security and Restoration Deposit:

Base Operating Costs:

Base Taxes: Taxes for fiscal tax year 2022 (i.e., July 1, 2022 - June 30,

2023) (subject to the terms of Section 4.2.1)

Tenant's Percentage: The ratio of the Rentable Floor Area of the Premises to the

total rentable area of the Building (which is approximately

131,568

square feet), which shall initially be deemed to be

35.34%.

Permitted Uses: General office and light research and development uses.

Commercial General Liability Insurance Limits: [***] per occurrence

[***] general aggregate

Commercial Excess Liability and/or Umbrella Insurance

Limits:

rella Insurance [***] general aggregate
[***] per occurrence

Guarantor:

Lantheus Holdings, Inc.

1.2 Exhibits.

The Exhibits listed below in this section are incorporated in this Lease by reference and are to be construed as a part of this Lease.

EXHIBIT A Legal Description of the Property

EXHIBIT B Plan showing the Premises

EXHIBIT C Commencement Date Notification

EXHIBIT D Preliminary Fit Plan
EXHIBIT D-1 Landlord's Work Letter
EXHIBIT E Work Change Order
EXHIBIT F Rules and Regulations

EXHIBIT F-1 Construction Rules and Regulations

EXHIBIT G Form Tenant Estoppel Certificate

EXHIBIT H Form of Landlord's Consent and Waiver

ARTICLE 2 Premises and Term

2.1 <u>Premises.</u> Landlord hereby leases the Premises to Tenant and Tenant hereby leases the Premises from Landlord, subject to and with the benefit of the terms, covenants, conditions and provisions of this Lease and any agreements, easements, and restrictions of record, as the same may be amended from time to time, applicable to the Property. Excluded from the Premises are the roof, exterior faces of exterior walls, the common

stairways, stairwells, elevators and elevator shafts, and pipes, ducts, conduits, wires, and appurtenant fixtures serving exclusively or in common other parts of the Building (and any areas, such as the

space above the ceiling or in the walls, that may contain such pipes, ducts, conduits, wires or appurtenant fixtures), and if Tenant's space includes less than the entire rentable area of any floor, excluding the central core area of such floor.

Tenant shall have, as appurtenant to the Premises, rights to use in common, subject to reasonable rules of general applicability to tenants of the Building from time to time made by Landlord of which Tenant is given notice: (a) the common lobbies, hallways, stairways, and elevators of the Building, (b) the Common Areas (defined in Section 5.1.5), and (c) if the Premises include less than the entire rentable area of any floor, the common toilets and other common facilities in the central core area of such floor.

Tenant shall be permitted to use 158 unreserved parking spaces in the surface parking area serving the Building on a first come, first served basis.

Landlord reserves the right from time to time, upon reasonable prior notice (except in the case of emergency) and without unreasonable interference with Tenant's use of the Premises: (a) to install, use, maintain, repair, replace and relocate for service to the Premises and other parts of the Building, or either, pipes, ducts, conduits, wires and appurtenant fixtures, wherever located in the Premises or Building, (b) to alter or relocate any other common facility, (c) to make any repairs and replacements to the Premises which Landlord may deem necessary, and (d) in connection with any excavation made upon adjacent land of Landlord or others, to enter, and to license others to enter, upon the Premises to do such work as the person causing such excavation deems necessary to preserve the wall of the Building from injury or damage and to support the same. Landlord shall use commercially reasonable efforts to perform all repairs, improvements and additions in such a manner as to minimize interference with Tenant's use of the Premises.

2.2 <u>Term.</u> TO HAVE AND TO HOLD for an original term (the "Original Term") beginning on the Commencement Date, which shall be the earlier of (a) the date on which the work to be performed by Landlord pursuant to Exhibits D and D-1 has been Substantially Completed or (b) the opening by Tenant of its business in the Premises, and ending on the Expiration Date, unless sooner terminated as hereinafter provided. The term "Substantially Completed" as used herein shall mean that the work to be performed by Landlord pursuant to Exhibits D and D-1 has been completed with the exception of minor items which can be fully completed without material interference with Tenant and other items which because of the season or weather or the nature of the item are not practicable to do at the time (collectively, "Punchlist Items"), provided that none of said items is necessary to make the Premises tenantable for the Permitted Uses, and a Certificate of Occupancy (which may be a temporary Certificate of Occupancy) is issued by the Town of Bedford. For the purposes of this paragraph, a temporary Certificate of Occupancy shall mean a temporary Certificate of Occupancy issued by the Town of Bedford, provided that such temporary Certificate of Occupancy allows Tenant's use and occupancy of, and access to a substantial portion of the Premises for the Permitted Uses. Promptly after receipt of a temporary Certificate of Occupancy, Landlord shall diligently pursue the receipt of a final Certificate of Occupancy. Within ten (10) days after Tenant commences business operations in the Premises,

Landlord and Tenant, each acting reasonably and in good faith, shall conduct a joint walk- through of the Premises in order to prepare a document identifying the Punchlist Items. When the Commencement Date, the Initial Rent Commencement Date, the First Rent Adjustment Date, the Second Rent Adjustment Date, the Full Rent Adjustment Date, the date of the 1st Lease Year, and the Expiration Date have been determined, such dates shall be evidenced by a document, in the form attached hereto as Exhibit C, which Landlord shall complete and deliver to Tenant, and which shall be deemed conclusive unless Tenant shall notify Landlord of any disagreement therewith within ten (10) business days of receipt.

The term "Lease Year" as used herein shall mean a period of twelve (12) consecutive full calendar months. The 1st Lease Year shall begin on the Commencement Date if the Commencement Date is the first day of a calendar month; if not, then the 1st Lease Year shall commence upon the first day of the calendar month immediately following the Commencement Date. Each succeeding Lease Year shall commence upon the anniversary date of the 1st Lease Year. The "Phase-in Periods" specified in Section 4.1 of this Lease serve to establish specific periods for phased-in rent payments and shall have no bearing on the determination of the beginning and end of the 1st Lease Year and the 2nd Lease Year, respectively. The term "Partial Lease Year" as used herein shall mean the six (6) month period commencing on the first day immediately following the expiration of the 8th Lease Year. For the avoidance of doubt, the Original Term shall be comprised of 102 full calendar months (plus the portion of any calendar month in which the Commencement Date occurs).

2.3 Extension Option.

- (a) Tenant shall have the option (the "Extension Option") to extend the Original Term of this Lease for one additional period of five (5) years to begin immediately upon the expiration of the Original Term (the "Extended Term"), provided that each of the following conditions has been satisfied:
 - (i) As of the date of the Extension Notice (defined below) and as of the commencement of the Extended Term, Tenant shall not (x) be in monetary default and/or (y) be in non-monetary default beyond notice and cure period and/or (z) have been in default of its obligations under this Lease beyond any applicable grace period more than once during the immediately preceding 12-month period in each instance, and is occupying 100% of the Premises;
 - (ii) Tenant and/or Guarantor shall have had a net income for the 12-month period immediately preceding the date of the Extension Notice and for the 12-month period immediately preceding the commencement of the Extended Term; and
 - (iii) Simultaneously with the delivery of the Extension Notice and also at the commencement of the Extended Term, Tenant shall have delivered to Landlord (x) consolidated audited statements (collectively, an "Audited Statement") for Tenant

and/or Guarantor for the preceding year, prepared using generally accepted accounting principles, and (y) unaudited consolidated statements for Tenant and/or Guarantor for the most recently completed quarter prepared using generally accepted accounting principles, all of which are evidencing such net income during each of the periods specified in clause (ii) hereinabove; it being agreed that if and when an additional Audited Statement is completed between the delivery of the Extension Notice and the commencement of the Extension Term (an "Updated Audited Statement"), Tenant will deliver such Updated Audited Statement to Landlord (and such Updated Audited Statement must satisfy the requirements set forth herein in all respects).

If the above conditions have not been satisfied, then the Extension Option shall be void and the Extension Notice previously sent by Tenant shall be nullified and of no effect. Notwithstanding anything to the contrary contained herein, Landlord hereby reserves the right, in its sole and absolute discretion, to waive any condition set forth in clauses (i), (ii), and (iii) of this Section 2.3. No such waiver shall be effective unless communicated by Landlord to Tenant in writing and shall be used for the sole purpose of giving effect to the Extension Option and no other purpose.

- (b) All of the terms, covenants and provisions of this Lease shall apply to the Extended Term except that the Annual Fixed Rent Rate for Extended Term shall be the fair-market rental rate for comparable space in comparable Class A office buildings in the Burlington/Bedford area as of the commencement of the Extended Term ("Market Rate"), as designated by Landlord. If Tenant shall elect to exercise this Extension Option, it shall do so by giving Landlord written notice (the "Extension Notice") of its election not later than twelve (12) months prior to the expiration of the Original Term of this Lease, time being of the essence thereof. If timely and properly Tenant gives such notice and satisfies the conditions specified above, the extension of this Lease shall be automatically effected without the necessity of the execution of any additional lease or amendment documents. Tenant's Extension Option is personal to the initial named Tenant, Lantheus Medical Imaging, Inc. (and any assignee or subtenant pursuant to a Permitted Transfer, as defined in Section 6.2.1 below), and may not otherwise be assigned under any circumstances. The Original Term and the Extended Term shall be collectively referred to as the "Term" or the "term".
- (c) Not later than thirty (30) days following the giving of Tenant's Extension Notice, Landlord shall notify Tenant of Landlord's determination of the Market Rate for the Extended Term. Within fifteen (15) business days after Landlord gives Tenant Landlord's determination of the Market Rate, Tenant shall notify Landlord whether Tenant accepts or disputes such rate. If Tenant disagrees with Landlord's determination, then Landlord and Tenant shall commence negotiations to agree upon the Market Rate. In any event, the Annual Fixed Rent Rate for the Extended Term shall not be less than the Annual Fixed Rent Rate in effect immediately prior to the commencement of the Extended Term, inclusive of any escalations. If Landlord and Tenant are unable to reach agreement on the Market Rate within thirty (30) days after the date on which Landlord first gave Tenant

Landlord's proposal for the Market Rate, then the Market Rate shall be determined as provided below.

- (d) If Landlord and Tenant are unable to agree on the Market Rate by the end of said thirty (30)-day period, then within five (5) business days thereafter, Landlord and Tenant shall each simultaneously submit to the other in a sealed envelope its good faith estimate of the Market Rate. If the higher of such estimates is not more than [***] of the other estimate, then the Market Rate shall be the average of the two estimates. If the matter is not resolved by the exchange of estimates, then Market Rate shall be determined by an independent arbitrator as set forth below.
- E. Within seven (7) days after the exchange of estimates, the parties shall select, as an arbitrator, a mutually acceptable commercial real estate broker or appraiser licensed in the Commonwealth of Massachusetts specializing in the field of commercial office leasing in the Burlington/Bedford market area, having no less than ten (10) years' experience (an "Approved Arbitrator"). If the parties cannot agree on such person, then within a second period of seven (7) days, each shall select one Approved Arbitrator and the two appointed Arbitrators shall, within five (5) days, select a third Approved Arbitrator who shall be the final decision-maker (the "Final Arbitrator"). If one party shall fail to timely make such appointment, then the person chosen by the other party shall be the sole arbitrator. Once the Final Arbitrator has been selected as provided for above, then, as soon thereafter as practicable, but in any case within fourteen (14) days after his or her appointment, the arbitrator shall determine the Market Rate by selecting either the Landlord's estimate of Market Rate or the Tenant's estimate of Market Rate. Such arbitrator must choose the proposed Market Rate that he/she determines is closest to the actual market rental rate for the Premises. There shall be no discovery or similar proceedings. The arbitrator's decision as to which estimate shall be the Market Rate for the Extended Term shall be rendered in writing to both Landlord and Tenant and shall be final and binding upon them and shall be the Annual Fixed Rent for the Extended Term. The costs of the Final Arbitrator will be equally divided between Landlord and Tenant. Any fees of any Approved Arbitrator or counsel engaged by Landlord or Tenant, however, shall be borne by the party that retained such Approved Arbitrator or counsel. If the dispute between the parties as to a market rate has not been resolved before the commencement of the Extended Term, then Tenant shall continue to pay Fixed Rent under the Lease based upon the Fixed Rent Rate in effect immediately prior to the commencement of the Extended Term until either the agreement of the parties as to the market rate, or the decision of the Final Arbitrator, as the case may be, at which time Tenant shall pay any underpayment of Fixed Rent to Landlord.
- F. Once the Market Rate has been determined, the parties shall promptly execute an amendment to this Lease setting forth the Fixed Rent for the Premises during the Extended Term.
- G. With respect to any assignment or subletting during the Original Term of this Lease, such assignment shall not include the Extension Option granted to Tenant, and such sublease shall be for a term expiring no later than the Expiration Date.

- 2.4 Landlord's Termination Right.(a) Landlord shall have the one-time right to terminate this Lease (the "Early Termination Right"), effective as of the last day of the 5th Lease Year (the "Early Termination Date"). If Landlord elects to exercise the Early Termination Right, it shall do so by giving Tenant written notice of termination (the "Early Termination Notice") no later than eighteen (18) calendar months prior to the Early Termination Date. Upon the Early Termination Date Tenant shall surrender and yield up the Premises in accordance with the provisions of Section 6.1.9 of this Lease and this Lease shall terminate without any further accruing liability or obligation to either party, except with respect to the provisions of this Lease intended to survive termination. The return of the Security and Restoration Deposit following the Early Termination Date shall be governed by the terms of Section 4.4 below. Landlord shall provide Tenant with an allowance in the amount of [***] for the costs of Tenant's furniture, fixtures and equipment, which shall be paid to Tenant on the Early Termination Date. In the event that Landlord does not give the Early Termination Notice to Tenant within the time period set forth above then Landlord shall be deemed to have waived its Early Termination Right and the provisions of this Section
 - 2.4 shall have no further force and effect. The period of time beginning when Landlord gives the Early Termination Notice and ending on the Early Termination Date is referred to in this Section 2.4 as the "Flat Fixed Rent Period".
 - (b) Fixed Rent for the Flat Fixed Rent Period shall be at the rate in effect for the third (3rd) Lease Year. It is the intention of the parties that during the Flat Fixed Rent Period Tenant shall receive the financial benefit of a full 24-months of Fixed Rent at the rate in effect for the 3rd Lease Year even though there will only be eighteen (18) months of term remaining until the Early Termination Date. Tenant shall receive a \$24,722.00 credit applied against the Fixed Rent owing during the Flat Fixed Rent Period to offset the excess Fixed Rent previously paid by Tenant during the 6 months immediately preceding the Flat Fixed Rent Period (the excess being the difference between the Fixed Rent rates in effect during the 3rd and 4th Lease Years.

ARTICLE 3 **Improvements**

3.1 <u>Performance of Landlord's Work.</u> Landlord shall, at its expense, cause to be performed the work ("Landlord's Work") to provide a turnkey buildout substantially as shown on preliminary fit plan attached hereto as Exhibit D and Landlord's Work Letter attached hereto as Exhibit D-1 and in accordance with the Final Plans (as defined below). All Landlord's Work shall be done in a good and workmanlike manner employing Building- standard materials and finishes (as specified in Exhibit D-1), and so as to conform to all applicable building laws. Tenant understands there is significant value in the existing first- class improvements, finishes and infrastructure in the Premises and agrees that Landlord may utilize existing improvements, fixtures and equipment to reduce costs. Tenant agrees that Landlord may make any changes in Landlord's Work which may become reasonably necessary or advisable, other than substantial changes, without approval of Tenant, provided written notice is promptly given to Tenant; and Landlord may make substantial changes in such work, subject to the written approval of Tenant, which shall not be unreasonably withheld or delayed. If Tenant wishes to make any change to Landlord's

Work at any time (including changes to finishes), whether prior to or during the performance of Landlord's Work, and if that change increases the cost and scope of work provided for in Exhibits D, and subsequently the Final Plans, then, provided Landlord has approved such change, Tenant shall execute a Work Change Order in the form attached hereto as Exhibit E, and shall fund 100% of that specific Work Change Order prior to Landlord's commencement of the work in such Change Order. Landlord shall use diligence to cause Landlord's Work to be Substantially Completed by the Anticipated Delivery Date, subject to the provisions of Section 10.5 hereof, and any Tenant Delay (defined in Section 3.3 of this Lease). In no event shall Landlord's consent to the alterations and improvements shown on Exhibit D or the Final Plans or approval of any other plans, nor Landlord or Landlord's agents performing work on Tenant's behalf, be construed as a waiver of Landlord's right, pursuant to Section 6.1.9 of this Lease, to require removal of any installations, alterations or improvements in connection with Landlord's Work at end of the Term, subject to all of the provisions of Section 6.1.9 of this Lease. Notwithstanding anything to the contrary contained herein, Tenant shall be solely responsible, at Tenant's sole cost and expense for the installation of Tenant's furniture, fixtures and equipment, Tenant's specialized installations and equipment, and telephone/data cabling within the Premises. All such installations shall be done in accordance with, and subject to, all of the terms and conditions of this Lease, including but not limited to Section 6.2.5 of this Lease. The Premises shall be delivered to Tenant in compliance with all applicable building codes, including those relating to life safety, and with all base building systems in good working order.

- 3.2 <u>Plan Approval; Final Plans.</u> Based on the preliminary fit plan attached hereto as Exhibit D, Landlord's architect shall prepare a full set of construction drawings and specifications sufficient to obtain a building permit for Landlord's Work (collectively, the "Construction Plans"). The Construction Plans shall emanate from and be consistent with Exhibit D. The Construction Plans will be submitted to Tenant for its approval. Tenant agrees to approve or disapprove the Construction Plans within five (5) business days following its receipt thereof, specifying the reasons for any disapproval. Landlord shall then cause its architect to redesign the Construction Plans incorporating those revisions reasonably requested by Tenant. Tenant's approval of the Construction Plans shall not be unreasonably withheld, conditioned or delayed. During the plan approval process, should Tenant require changes to the Construction Plans that increase the cost and scope of Landlord's Work, as reasonably determined by Landlord, then Tenant shall execute a Work Change Order in the form of Exhibit E attached hereto and the increased cost for any such changes shall be paid by Tenant in accordance the Work Change Order. Tenant shall pay the full amount of the Work Change Order in accordance with Exhibit E attached hereto. If Tenant fails to timely respond to any request for approval, the submission in question shall be deemed approved. The Construction Plans that are finally approved by Tenant shall be called the "Final Plans".
- 3.3 <u>Tenant Delay.</u> The phrase, "Tenant Delay", shall be defined as any delay in the completion of Landlord's Work actually caused by (i) special work, upgrades or long lead-time items for which Landlord identifies in writing (it being agreed such writing may be in the form of an e-mail), promptly upon submission of a Work Change Order, a specified period of delay, and in either instance Tenant does not withdraw or alter such special work, upgrade, long lead-time item which avoids such delay, (ii) any changes to any plans, including the

Final Plans, made by Tenant, or any Work Change Order requested by Tenant, in any case for which Landlord identifies a specified period of delay at the time of its approval and for which Tenant does not withdraw such change to avoid delay, (iii) any change in the design of the Landlord's Work or any component thereof made by Tenant that results in a delay in the Landlord's construction start date beyond the date that is sixty (60) days after Landlord's receipt of the construction drawings, (iv) the delay of Tenant or its architects and engineers in approving the Construction Plans, or supplying, submitting or approving any other plans, specifications, pricing or estimates or giving authorizations or supplying information reasonably required by Landlord or its general contractor or the Town of Bedford by the dates stipulated in this Article 3, or in any other instance if no time period is specified, then within five (5) business days after request therefor, (v) any failure by any contractors employed by Tenant including, without limitation, contractors furnishing telecommunications, data processing or other service or equipment directly to Tenant (and not via Landlord's contractors) to comply with the agreed upon timetables for coordination of the parties' respective components of work, as established at on-site or virtual progress meetings between Landlord's representative and Tenant's representative, each acting reasonably and in good faith, (vi) any failure to comply with this Article 3 or any material interference with the performance of Landlord's Work by Tenant or any of its agents, employees, architects, engineers or contractors, or (vii) Tenant's delay in paying a Work Change Order when due or in delivering the Security and Restoration Deposit required pursuant to Section 4.4.

- 3.4 <u>Early Access.</u> Upon full execution of the Lease, Tenant may, prior to the Commencement Date (but following coordination with Landlord as to the timing), enter the Premises without payment of rent, but otherwise subject to all the terms and conditions of this Lease, to facilitate Tenant's preparation of the Premises (e.g., taking measurements, installing trade fixtures and furniture, etc.), provided that (i) Tenant shall not interfere with any work then being performed by or for Landlord in the Premises or Building, (ii) Tenant shall immediately cease its activities in the Premises in the event that Landlord notifies Tenant (which notice may be given orally) that Tenant is interfering with Landlord, and (iii) provided Tenant shall reimburse Landlord for Landlord's actual costs incurred in connection with Tenant's pre-commencement entry and/or work. All such preparation by Tenant shall be done in accordance with the provisions of this Lease, including but not limited to Section 6.2.5 hereof.
- 3.5 <u>Delayed Delivery.</u> In the event that Landlord's Work is not Substantially Completed by the date that is one hundred and eighty (180) days after the Anticipated Delivery Date (the "Outside Delivery Date"), and provided such delay is not due to either a Tenant Delay and/or Force Majeure event, then the Fixed Rent first coming due as of the Initial Rent Commencement Date shall be abated by one (1) day for each day of delay during the period beginning on the Outside Delivery Date and ending on the day the Landlord's Work is Substantially Completed. Further, in the event that Landlord's Work is not Substantially Completed by the date that is three hundred and sixty (360) days after the Anticipated Delivery Date (the "Final Outside Date"), and provided such delay is not due to either a Tenant Delay and/or Force Majeure event, then Tenant shall have the right, upon written notice to Landlord given no later than ten (10) business days following the Final Outside Date (such notice the "Failure to Deliver Termination Notice"), to terminate this Lease

upon thirty (30) days prior notice; *provided*, *however*, that such termination will be rendered ineffective if, prior to the date that is thirty (30) days following the date Tenant gives the Failure to Deliver Termination Notice, Landlord shall have Substantially Completed Landlord's Work. The foregoing rights to abatement of Fixed Rent and terminate the Lease shall be Tenant's sole remedy at law or in equity for Landlord's failure to Substantially Complete the Landlord's Work by the timelines set forth herein.

ARTICLE 4 Rent

- 4.1 <u>The Fixed Rent.</u> (a) Tenant covenants and agrees to pay rent("Fixed Rent") to Landlord, by electronic fund transfer (or to such other person or entity as Landlord may by notice in writing to Tenant from time to time direct), at the Annual Fixed Rent Rate, in equal installments at the Monthly Fixed Rent Rate (which is 1/12th of the Annual Fixed Rent Rate), in advance, without notice or demand, and without setoff, abatement, suspension, deferment, reduction or deduction, except as otherwise expressly provided herein, on the first day of each calendar month included in the Term, commencing as of the Commencement Date; and for any portion of a calendar month at beginning of the Term, at the rate for the 1st Lease Year, payable in advance for such portion. The term "Additional Rent" shall mean all sums other than Fixed Rent that are payable to Landlord under this Lease, including, without limitation all Operating Costs, Taxes, late charges, and interest. It is intended that Tenant's initial obligation to pay Fixed Rent shall be phased-in during the first eighteen (18) months of the Original Term. The Fixed Rent stated in Section 1.1 is calculated based on specific phase-in periods with stepped-up square footage amounts during each phase-in period, as stated in Section 1.1.
 - (b) It is the intention of the parties hereto that the obligations of Tenant hereunder shall be separate and independent covenants and agreements, that the Fixed Rent, the Additional Rent and all other sums payable by Tenant to Landlord shall continue to be payable in all events and that the obligations of Tenant hereunder shall continue unaffected, unless the requirement to pay or perform the same shall have been abated or terminated pursuant to an express provision of this Lease.
 - (c) If Landlord shall give written notice (which for the purposes hereof may be in the form of an e-mail) to Tenant that all rent and/or other payments due hereunder are to be made to Landlord by wire transfer or any other commercially reasonable means, Tenant shall make all such payments as shall be due after receipt of said notice by means as designated by Landlord, with such payments to be made to such address and to such person or entity as is specified by Landlord.
- 4.2 <u>Additional Rent.</u> Tenant covenants and agrees to pay its insurance costs and, as Additional Rent, utility charges, personal property taxes and its pro rata share of increases in Operating Costs and Taxes with respect to the Premises as provided in this Section 4.2 as follows:
 - 4.2.1 <u>Real Estate Taxes.</u> If Taxes (as hereinafter defined) for any Tax Year during the Term shall exceed Base Taxes, Tenant shall reimburse Landlord. as

Additional Rent, for Tenant's Percentage of such excess (such amount hereinafter referred to as "Tax Excess"). The term "Tax Year" means the fiscal tax period of the Town of Bedford, which presently is a twelve-month period that begins on July 1 of each calendar year. Tenant shall remit to Landlord, on the first day of each calendar month, estimated payments on account of Tax Excess, such monthly amounts to be sufficient to provide Landlord, by the time real estate tax payments are due and payable to any governmental authority responsible for collection of same, a sum equal to the Tax Excess, as reasonably estimated by Landlord from time to time on the basis of the most recent tax data available. If the total of such monthly remittances for any Tax Year is greater than the actual Tax Excess for such Tax year, Landlord shall promptly pay to Tenant, or credit against the next accruing payments to be made by Tenant pursuant to this subsection 4.2.1, the difference; if the total of such remittances is less than the actual Tax Excess for such Tax Year, Tenant shall pay the difference to Landlord at least ten (10) days prior to the date or dates within such Tax Year that any Taxes become due and payable to the governmental authority (but in any event no earlier than ten (10) days following a written notice to Tenant, which notice shall set forth the manner of computation of Tax Excess).

If, after Tenant shall have made reimbursement to Landlord pursuant to this subsection 4.2.1, Landlord shall receive a refund of any portion of Taxes paid by Tenant with respect to any Tax Year during the term hereof as a result of an abatement of such Taxes by legal proceedings, settlement or otherwise (without Landlord having any obligation to undertake any such proceedings), Landlord shall promptly pay to Tenant, or credit against the next accruing payments to be made by Tenant pursuant to this subsection 4.2.1, the Tenant's Percentage of the refund (less the proportional, pro rata expenses, including commercially reasonable attorneys' fees and appraisers' fees, incurred in connection with obtaining any such refund), as relates to Tax Excess paid by Tenant to Landlord with respect to any Tax Year for which such refund is obtained.

In the event this Lease shall commence, or shall end (by reason of expiration of the Term or earlier termination pursuant to the provisions hereof), on any date other than the first or last day of the Tax Year, or should the Tax Year or period of assessment of real estate taxes be changed or be more or less than one (1) year, as the case may be, then the amount of Tax Excess which may be payable by Tenant as provided in this subsection 4.2.1 shall be appropriately apportioned and adjusted.

The term "Taxes" shall mean all taxes, assessments, betterments and other charges and impositions (including, but not limited to, fire protection service fees and similar charges) levied, assessed or imposed at any time during the term by any governmental authority upon or against the Property, or taxes in lieu thereof, and additional types of taxes to supplement real estate taxes due to legal limits imposed thereon. If, at any time during the term of this Lease, any tax or excise on rents or other taxes, however described, are levied or assessed against Landlord with respect to the rent reserved hereunder, either wholly or partially in substitution for, or in

addition to, real estate taxes assessed or levied on the Property, such tax or excise on rents shall be included in Taxes; however, Taxes shall not include late fees and penalties (unless incurred as a result of Tenant's failure to pay Tenant's Percentage of Tax Excess pursuant to the terms hereof), unpaid Taxes due from another tenant, franchise, estate, inheritance, succession, capital levy, transfer, income or excess profits taxes assessed on Landlord. Taxes shall include any estimated payment made by Landlord on account of a fiscal tax period for which the actual and final amount of taxes for such period has not been determined by the governmental authority as of the date of any such estimated payment.

In the event that the assessment for Taxes for the Property during the Base Taxes year (or any comparison year) specifically includes in writing a reduction for vacancy in the Building or a vacancy reduction is otherwise reasonably noted in writing on the assessor's tax field card, then the Taxes for the applicable Base Taxes year or any comparison year, as the case may be, during which such discount continues to be applied in whole or in part shall be reasonably grossed up by the amount of the discount or reduction actually applied by the assessing department of the Town of Bedford on account of such vacancy in the Building.

- 4.2.2 <u>Personal Property Taxes.</u> Tenant shall pay all taxes charged, assessed or imposed upon the personal property of Tenant in or upon the Premises.
- 4.2.3 Operating Costs. If, during the Term hereof, Operating Costs (as hereinafter defined) incurred by Landlord in any calendar year shall exceed Base Operating Costs, Tenant shall reimburse Landlord, as Additional Rent, for Tenant's Percentage of any such excess (such amount being hereinafter referred to as the "Operating Costs Excess"). Tenant shall remit to Landlord, on the first day of each calendar month, estimated payments on account of Operating Costs Excess, such monthly amounts to be sufficient to provide Landlord, by the end of the calendar year, a sum equal to the Operating Costs Excess, as reasonably estimated by Landlord from time to time. If, at the expiration of the year in respect of which monthly installments of Operating Costs Excess shall have been made as aforesaid, the total of such monthly remittances is greater than the actual Operating Costs Excess for such year, Landlord shall promptly pay to Tenant, or credit against the next accruing payments to be made by Tenant pursuant to this subsection 4.2.3, the difference; if the total of such remittances is less than the Operating Costs Excess for such year, Tenant shall pay the difference to Landlord within twenty (20) days from the date Landlord shall furnish to Tenant an itemized statement of the Operating Costs Excess, prepared, allocated and computed in accordance with generally accepted accounting principles. Any reimbursement for Operating Costs due and payable by Tenant with respect to periods of less than twelve (12) months shall be equitably prorated. Landlord shall use reasonable efforts to provide Tenant with any such reconciliation statement within one hundred and twenty (120) days following the expiration of any calendar year during the Term, such statement prepared, allocated and computed in accordance with generally

accepted accounting principles. This section shall survive the expiration or earlier termination of the Lease.

The term "Operating Costs" shall mean all costs and expenses incurred for the operation, cleaning, maintenance, repair and upkeep of the Property, including, without limitation, all costs of maintaining and repairing the Property, including snow removal; landscaping and grounds maintenance; operation, repair and maintenance of parking lots (including lighting), sidewalks, walking paths, access roads and driveways); all costs of Property signage, repair and maintenance of the roof of the Building (including drains, flashings, gutters, waterproofing); painting; security; operation, repair and replacement of heating and air-conditioning equipment, elevators, lighting and any other Building equipment or systems; all repairs and replacements (other than repairs or replacements for which Landlord has received full reimbursement from contractors, other tenants of the Building or from others) necessary to keep the Property in good working order, repair, appearance and condition; all costs, including material and equipment costs, for cleaning and janitorial services to the Building (including window cleaning of the Building); all costs of any reasonable insurance carried by Landlord relating to the Property; utilities attributable to the Common Areas; all costs related to provision of heat (including oil, electric, and/or gas), cooling, water (including sewer charges) refuse disposal and other utilities to the Premises, if not separately metered, and to the Building (exclusive of reimbursement to Landlord for any of same received as a result of direct billing to any tenant of the Building or costs for utilities that are separately metered); payments under all service contracts relating to the foregoing; all compensation, fringe benefits, payroll taxes and workmen's compensation insurance premiums related thereto with respect to any employees of Landlord or its affiliates at the level of property manager or below who are engaged in the operation, security and maintenance of the Property; costs of maintaining, repairing and operating amenities available for the general use of all tenants that are now or hereafter located in the Common Areas, including an on-site café (including a monthly cafeteria subsidy, if any), and a fitness facility; charges under any declarations, covenants, cross-easements, or other like documents of record applicable to the Property; attorneys' fees and disbursements (exclusive of any such fees and disbursements incurred in tax abatement proceedings or the preparation of leases) and auditing and other professional fees and expenses allocable to the operation, maintenance, repair and upkeep of the Property; and a property management fee which shall not exceed [***] of the gross receipts for the Building.

There shall not be included in such Operating Costs:

- (l) expenses for repairs, replacement or other work necessitated by fire or other casualty (including the action of any public authority in consequence thereof) or by any taking by eminent domain, but excluding such expenses to the extent of any insurance deductible;
- (2) salaries of partners, officers and executives of Landlord;

- (3) depreciation or amortization;
- (4) costs relating to alterations for other tenants in the Building and other expenses incurred in leasing or procuring any tenants or relocating tenants;
- (5) principal and interest of indebtedness;
- (6) costs for which Landlord separately charges Tenant or other tenants;
- (7) advertising, marketing costs, and leasing commissions associated with leasing space in the Building or selling the Property;
- (8) legal and accounting expenses not directly related to the operation, maintenance, repair and upkeep of the Property including, but not limited to lease negotiations and enforcement of leases;
- (9) costs or expenses associated with or incurred in connection with the removal, enclosure, encapsulation or other handling or of hazardous materials or hazardous substances (as referenced in Section 6.2.3 below) at the Building or the Property existing as of the Effective Date;
- (10) costs of financing or refinancing or costs in incurred in connection with any other financing, sale or syndication of the Property;
- (11) costs relating to maintaining Landlord's existence as a corporation, limited partnership or other entity or other corporate overhead of Landlord;
- (12) increased Taxes assessed specifically to any tenant of the Property for which Landlord is entitled to reimbursement from any other tenant;
- (13) reserves for bad debts, repairs or capital improvements;
- (14) any expenses for repairs or maintenance to the extent recovered under warranties, guarantees and service contracts;
- (15) costs of purchasing or renting paintings or other works of art;
- (16) charitable or political contributions;
- (17) costs of capital improvements except to the extent permitted in the paragraph below;
- (18) rental on ground leases or other underlying leases and the costs of providing the same;
- (19) cost of any HVAC, janitorial or other services provided to tenants on an extra cost basis after regular business hours;

- (20) cost of any work or service performed on an extra cost basis for any tenant in the Building or the Property to a materially greater extent or in a materially more favorable manner than furnished generally to the tenants and other occupants;
- (21) any cost representing an amount paid to a person firm, corporation or other entity related to Landlord that is in excess of the amount which would have been paid in the absence of such relationship;
- (22) Landlord's general overhead and any other expenses not directly attributable to the operation, cleaning, maintenance, repair, upkeep and management of the Building and the Property (e.g. the activities of Landlord's officers and executives or professional development expenditures), except to the extent included in the management fee permitted hereby;
- (23) costs and expenses incurred in connection with compliance with or contesting or settlement of any claimed violation of law or requirements of law on the part of Landlord; and
- (26) expenditures already paid by Tenant with respect to (a) cleaning, maintenance and upkeep of the Premises, and (b) the provision of electricity to the Premises (such that such items shall not be charged to Tenant more than once).
- If, during the Term, Landlord shall replace any capital items or make any capital expenditures which (a) are intended to reduce Operating Costs or (b) are required to comply with laws enacted after the Effective Date, or (c) are required to replace worn-out items as may be necessary to maintain the Building in good working order, repair and condition and not to enhance the Building over and above its current appearance and condition (collectively called "capital expenditures") the total amount of which is not properly included in Operating Costs for the calendar year in which they were made, there shall nevertheless be included in Operating Costs for each calendar year in which and after such capital expenditure is made the annual charge-off of such capital expenditure. (Annual charge-off shall be determined by (i) dividing the original cost of the capital expenditure by the number of years of useful life thereof [The useful life shall be reasonably determined by Landlord in accordance with generally accepted accounting principles and practices in effect at the time of acquisition of the capital item.]; and
- (ii) adding to such quotient an interest factor computed on the unamortized balance of such capital expenditure based upon an interest rate reasonably determined by Landlord as being the interest rate then being charged for long-term mortgages by institutional lenders on like properties within the locality in which the Building is located.) Provided, further, that if Landlord reasonably concludes on the basis of engineering estimates that a particular capital expenditure will effect savings in Operating Costs and that such annual projected savings will exceed the annual charge-off of capital expenditure computed as aforesaid, then and in such events, the annual charge-off shall be determined by dividing the amount of such capital expenditure by the number of years over which the projected amount of such

savings shall fully amortize the cost of such capital item or the amount of such capital expenditure; and by adding the interest factor, as aforesaid.

If during any portion of any year for which Operating Costs are being computed (including the calendar year for Base Operating Costs), the Building was not fully occupied by tenants or if not all of such tenants were paying fixed rent or if Landlord was not supplying all tenants with the services being supplied hereunder, actual Operating Costs incurred shall be reasonably extrapolated by Landlord to the estimated Operating Costs that would have been incurred if the Building were fully occupied by tenants and all such tenants were then paying fixed rent or if such services were being supplied to all tenants, and such extrapolated amount shall, for the purposes of this Section 4.2.3, be deemed to be the Operating Costs for such year. If Operating Costs in any year increase due to a new service being introduced in operating the Property, such increase shall be included in Operating Costs only to the extent of the increase in cost over the projected costs that would have been included in the calendar year for Base Operating Costs Year if such service had been in effect during the calendar year for Base Operating Costs. Conversely if the Operating Costs in any year decrease due to an elimination of any service, then the cost of such service shall be deleted from the calendar year for Base Operating Costs for purposes of determining the Operating Costs payable for subsequent years that the service is not included.

Provided that Tenant is not in default beyond notice and cure periods of its obligations under this Lease (it being agreed that if Landlord has given notice of default for which the applicable cure period has yet to lapse, Tenant must cure such default within such cure period for Tenant to be entitled to exercise the right set forth herein), Tenant shall have the right to examine, copy and audit Landlord's books and records establishing the Operating Costs set forth in this Section 4.2.3, for the calendar year immediately preceding Landlord's yearend reconciliation statement of Operating Costs for the specific calendar year in question. Tenant shall give Landlord at least thirty (30) days prior written notice (the "Audit Notice") of its intention to examine and audit such books and records, and such examination and audit shall take place at Landlord's or Landlord's building manager's office no later than on hundred twenty (120) days following Tenant's receipt of any year-end reconciliation statement and shall be completed no later than ninety (90) days following the date Tenant was first given access to Landlord's books and records. Before conducting any audit, Tenant must pay the full amount of Tenant's Percentage of Operating Costs Excess billed. No subtenant shall have any right to conduct an audit. As a condition to performing any such inspection, Tenant and its examiners may be required to execute and deliver to Landlord an agreement to keep confidential any information which Tenant and the examining party discover about the Building in connection with such examination, except for disclosures required by law, court order or regulatory authorities, or to Tenant's attorneys, accountants, auditors, or potential purchasers of the Tenant company. Tenant agrees to use for such audit either (i) a certified public accountant or (ii) an accounting firm or (iii) or a lease administration firm that is subject to Landlord's prior approval (not to be unreasonably withheld, conditioned or delayed), provided that any such party listed

- in (i), (ii) and/or (iii) is not being paid on a contingency fee basis. All costs of the examination and audit shall be borne by Tenant provided, however, that if such examination and audit establishes that the Operating Costs Excess for the year in question is less than the amount set forth in Landlord's statements delivered to Tenant by more than five percent (5%), then Landlord shall pay to Tenant the reasonable costs of such examination and audit (not to exceed a maximum amount of [***]). If, pursuant to the audit, the payments made for such year by Tenant exceed Tenant's required payment on account thereof for such year, such overpayment shall be credited against Rent next due, or refunded to Tenant if the Term of this Lease has then expired and Tenant has no further obligation to Landlord; but, if the payments made by Tenant for such year are less than Tenant's required payment as established by the examination and audit, Tenant shall pay the deficiency to Landlord within thirty (30) days after conclusion of the examination and audit. If there is any dispute over the results of the audit, Landlord shall have ninety (90) days following receipt of the audit results to obtain an audit from an accountant of Landlord's choice, at Landlord's cost and expense. In the event that Landlord's and Tenant's accountants shall be unable to reconcile the results which thirty (30) days following completion of Landlord's accountants' review, then, both accountants shall mutually agree upon a third accountant who shall make a final determination within sixty (60) days after being appointed by the parties and whose determination shall be conclusive. The cost of any such third accountant shall be shared equally between Landlord and Tenant.
- 4.2.4 <u>Insurance.</u> Tenant shall, at its expense, as Additional Rent, take out and maintain from the time Tenant first occupies the Premises for any purpose and throughout the Term the following insurance protecting Landlord:
 - 4.2.4.1 Commercial general liability insurance and commercial excess liability insurance on "follow form" basis and/or umbrella naming Landlord, Tenant, and Landlord's managing agent and any mortgagee of which Tenant has been given notice as insureds or additional insureds, on a primary and noncontributory basis, in amounts which shall, at the beginning of the term, be at least equal to the limits set forth in Section 1.1; and, which, from time to time during the Term upon written notice from Landlord, shall be for such higher limits, if any, as are customarily carried in the area in which the Premises are located on property similar to the Premises and used for similar purposes; and workmen's compensation insurance with statutory limits covering all of Tenant's employees working on the Premises, covering the state in which the employee was hired, works and resides; and Employers liability insurance with minimum limits of [***] each accident, bodily injury by disease.
 - 4.2.4.2 Special Risk property insurance with the usual extended coverage endorsements covering all Tenant's furniture, furnishings, fixtures and equipment, and any alterations or improvements installed by or on behalf of Tenant with the exception of the initial Landlord's Work, and business

- interruption insurance with limits not less than the equivalent of 12 months of rent, with extra expense coverage, and shall list Landlord as loss payee as their interests may appear.
- 4.2.4.3 Automobile liability insurance for all owned, leased, non-owned and hired vehicles. The minimum limit of liability shall be [***] each accident, combined single limit for bodily injury and property damage.
- 4.2.4.4 All such policies shall be obtained from responsible companies having a policy rating of A-; X or better, as set forth in the most current issue of the Best's Key Rating Guide and which are qualified to do business and in good standing in Massachusetts. Tenant agrees to furnish Landlord with certificates evidencing all such insurance prior to the beginning of the Term hereof and evidencing renewal thereof at least thirty (30) days prior to the expiration of any such policy. Tenant shall provide at least thirty (30) days prior written notice to Landlord should any of the policies required herein be cancelled. In the event provision for any such insurance is to be by a blanket insurance policy, the policy shall allocate a specific and sufficient amount of coverage to the Premises.
- All insurance which is carried by either party with respect to the Building, Premises or to furniture, furnishings, fixtures, or equipment therein or alterations or improvements thereto, workmen's compensation insurance, and all liability insurance, whether or not required, shall include provisions which either designate the other party as one of the insured or deny to the insurer acquisition by subrogation of rights of recovery against the other party to the extent such rights have been waived by the insured party prior to occurrence of loss or injury, insofar as, and to the extent that, such provisions may be effective without making it impossible to obtain insurance coverage from responsible companies qualified to do business in the state in which the Premises are located. Each party shall be entitled to have certificates of any policies containing such provisions. Each party hereby waives all rights of recovery against the other for loss or injury against which the waiving party is protected by insurance containing said provisions, reserving, however, any rights with respect to any excess of loss or injury over the amount recovered by such insurance. Tenant shall not acquire as insured under any insurance carried on the Premises which includes all tenant improvements any right to participate in the adjustment of loss or to receive insurance proceeds and agrees upon request promptly to endorse and deliver to Landlord any checks or other instruments in payment of loss in which Tenant is named as payee.
- 4.2.5 <u>Utilities.</u> (a) Commencing on the Commencement Date, Tenant shall reimburse Landlord through Operating Costs, pursuant to Section 4.2.3, for all charges for the cost of electricity for base building HVAC, gas, and water (including sewer charges) for ordinary cleaning and lavatory use during the Term. Tenant shall pay to Landlord by means of monthly escrow payments (as set

forth in Section 4.2.5(b)), for all charges for the cost of electricity for Tenant's lights, outlets, VAV boxes/Water Source Heat Pumps, and any special requirements, such as supplemental HVAC units, for example, and for any above-normal office usage of electricity at the Premises ("Tenant Electric").

- (b) Commencing as of the Commencement Date], Tenant shall pay directly to Landlord, as Additional Rent, estimated charges (the "Utilities Charges") on account of Tenant Electric. Landlord shall reasonably estimate the amount of Utilities Charges payable by Tenant per month and shall notify Tenant prior to the Commencement Date of the initial estimate of Utilities Charges to be paid by Tenant. Tenant shall pay the Utilities Charges on the first day of each calendar month included in the Term, in the same manner as Tenant pays Fixed Rent pursuant to Section 4.1 above. Following Landlord's receipt of utility provider invoices, Landlord will reconcile the estimated Utilities Charges paid by Tenant with the actual amounts owing from Tenant based on the number of kilowatt hours of electricity used in the Premises for the preceding month as registered on the sub- meters for the Premises. If it is determined Tenant has been overcharged, then such overpayment will be credited against Tenant's account for the following month. If Tenant has underpaid, then Landlord will invoice Tenant for the amount owed and Tenant shall pay such amount within twenty (20) days after billing. Landlord reserves the right to adjust the monthly Utilities Charges from time to time based on the most current data available for Tenant's electrical, and Tenant shall thereafter pay the adjusted Utilities Charges to Landlord until further notice.
- (c) Tenant shall pay all charges for telephone and other utilities or services not supplied by Landlord, whether designated as a charge, tax, assessment, fee or otherwise, all such charges to be paid as the same from time to time become due. Except as otherwise provided in Article 5, it is understood and agreed that Tenant shall make its own arrangements for the installation or provision of all such utilities and that Landlord shall be under no obligation to furnish any utilities to the Premises and shall not be liable for any interruption or failure in the supply of any such utilities to the Premises.
- 4.3 <u>Late Payment of Rent.</u> If any installment of Fixed Rent or other amount due Landlord is paid after the date the same was due, and if on a prior occasion in the twelve (12) month period prior to the date such installment was due an installment of rent was paid after the same was due, then Tenant shall pay Landlord a late payment fee equal to five (5%) percent of the current overdue payment. In addition, if any installment of rent or other sum due Landlord is not paid when due, such installment shall bear interest from the date due until paid, at the rate of 12% per year not to exceed the highest rate permitted by law.
- 4.4 <u>Security and Restoration Deposit:</u> Upon Tenant's execution of this Lease, Tenant shall deposit with Landlord the Security and Restoration Deposit. Said deposit shall be held by Landlord as security for the faithful performance by Tenant of all the terms of this Lease by said Tenant to be observed and performed. The Security and Restoration Deposit shall not be mortgaged, assigned, transferred or encumbered by Tenant without the written

consent of Landlord and any such act on the part of Tenant shall be without force and effect and shall not be binding upon Landlord. If the Fixed Rent or Additional Rent or any other sum payable hereunder shall be overdue and unpaid or should Landlord make payments on behalf of the Tenant, or Tenant shall fail to perform any of the terms of this Lease, then Landlord may, at its option and without prejudice to any other remedy which Landlord may have on account thereof, appropriate and apply said entire deposit or so much thereof as may be necessary to compensate Landlord toward the payment of Fixed Rent, Additional Rent or other sums or loss or damage sustained by Landlord due to such breach on the part of Tenant; and Tenant shall forthwith within five (5) business days of demand restore said security to the original sum deposited. Said deposit shall be returned in full (or so much of the Security and Restoration Deposit as shall remain following any application thereof by Landlord) to Tenant within forty –five (45) days after the end of the Term (it being agreed that, if at the time Landlord is obligated to return such Security and Restoration Deposit, Landlord has given Tenant a notice of default for which the applicable cure period has yet to lapse, the forty-five (45) day period shall be extended to the end of the applicable cure period, and thereafter Landlord shall be entitled to draw upon the Security and Restoration Deposit to cure such default before returning what remains of the Security and Restoration Deposit to Tenant). In the event of bankruptcy or other creditor-debtor proceedings against Tenant, all securities shall be deemed to be applied first to the payment of rent and other charges due Landlord for all periods prior to the filing of such proceedings

ARTICLE 5 Landlord's Covenants

- 5.1 Affirmative Covenants. Landlord covenants with Tenant:
 - 5.1.1 Heat and Air-Conditioning. To furnish to the Premises heat and air-conditioning (reserving the right, at any time, to change energy or heat sources), sufficient to maintain the Premises at comfortable temperatures (subject to all federal, state, and local regulations relating to the provision of heat), during the normal business hours for the Building set forth in the Rules and Regulations attached hereto as Exhibit F ("Normal Building Hours"). Heat and air-conditioning will be available after Normal Building Hours, at Landlord's then standard rate for after-hours usage, including wear and tear on Building equipment, as reasonably determined by Landlord from time to time. If Tenant desires heat or air-conditioning after Normal Building Hours, Tenant must provide Building management with at least 24 hours prior notice.
 - 5.1.2 <u>Electricity.</u> To furnish to the Premises reasonable electricity for Tenant's Permitted Uses, separately check metered for Tenant's lights, plugs and supplemental hvac and at the direct expense of Tenant as provided in Section 4.2.5 above. If Tenant shall require electricity in excess of reasonable quantities for Tenant's Permitted Uses and if (i) in Landlord's reasonable judgment,

Landlord's facilities are inadequate for such excess requirements, or (ii) such excess use shall result in an additional burden on the Building utilities systems and additional cost to Landlord on account thereof, as the case may be, (a) Tenant shall, upon demand, reimburse Landlord for such additional cost, as aforesaid, or (b) Landlord, upon written request, and at the sole cost and expense of Tenant, will furnish and install such additional wire, conduits, feeders, switchboards and appurtenances as reasonably may be required to supply such additional requirements of Tenant (if electricity therefor is then available to Landlord), provided that the same shall be permitted by applicable laws and insurance regulations and shall not cause permanent damage or injury to the Building or cause or create a dangerous or hazardous condition or entail excessive or unreasonable alterations or repairs.

- 5.1.3 <u>Cleaning; Water.</u> To provide cleaning to the Premises on business days in accordance with cleaning and janitorial standards generally prevailing throughout the Term hereof in comparable office buildings within the municipality in which the Building is located; and to furnish water for drinking, lavatory and toilet purposes.
- 5.1.4 <u>Elevator; Fire Alarm.</u> To furnish elevator service from the lobby to the Premises; and to maintain fire alarm systems within the Building.
- 5.1.5 Repairs. Except as otherwise expressly provided herein, to make such repairs and replacements to the Common Areas, the roof, exterior walls and exterior windows, floor slabs and other structural components of the Building, and to the facilities and the base building plumbing, electrical, heating, ventilating and air-conditioning systems of the Building as may be necessary to keep them in good repair and condition (exclusive of equipment installed by Tenant and except for those repairs required to be made by Tenant pursuant to Section 6.1.3 hereof and repairs or replacements occasioned by any act or negligence of Tenant, its servants, agents, customers, contractors, employees, invitees, or licensees). The term "Common Areas" shall mean all areas within the Property which are not leased or held for the exclusive use of Tenant, including, but not limited to, parking areas, driveways, sidewalks, access roads, and landscaping. Landlord, from time to time, may change the size, location, nature and use of any of the Common Areas, construct additional parking facilities in the Common Areas and increase or decrease the Common Area land so long as Tenant's use of the Premises is not materially affected. At any time, Landlord may close any portion of the Common Areas to perform any acts that are, in Landlord's reasonable judgment, necessary or desirable to maintain, repair or improve the Property.
- 5.1.6 Access. To provide access to the Common Areas and the Building 24 hours per day, 7 days per week, subject to happenings beyond the control of Landlord,

and subject further to any security requirements implemented by Landlord for all tenants of the Building.

- 5.1.7 <u>Compliance with Law.</u> Landlord shall comply with all applicable laws relating to the Building and the Common Areas, provided that compliance with such applicable laws is not the responsibility of Tenant under this Lease or another tenant or occupant under its lease, and provided further that Landlord's failure to comply therewith would prohibit Tenant from obtaining or maintaining a certificate of occupancy for the Premises, or would unreasonably and materially affect the safety of Tenant, its employees or visitors, or otherwise materially interfere with or materially impact Tenant's Permitted Use of the Premises. Notwithstanding the foregoing, Landlord shall not be obligated to comply with any law to the extent it has a good faith objection to such compliance and is pursuing the same with applicable authorities.
- 5.2 <u>Interruption</u>. Landlord shall be under no responsibility or liability for failure or interruption of any of the abovedescribed utility services, or conditions arising in or about the Property caused by breakage, accident, strikes, repairs, inability to obtain supplies, labor or materials, or for any other causes beyond the control of the Landlord, and in no event for any indirect or consequential damages to Tenant; and failure or omission on the part of the Landlord to furnish any of the above services for any of the reasons set forth in this paragraph shall not be construed as an eviction of Tenant, actual or constructive, nor entitle Tenant to an abatement of rent, nor render the Landlord liable in damages, nor release Tenant from prompt fulfillment of any of its covenants under this Lease. However in each instance of failure or interruption Landlord shall use reasonable efforts to restore the unavailable service or remedy the condition following written notice from Tenant. Notwithstanding the foregoing, if an event or circumstance (an "Abatement Event") shall occur that causes an interruption or curtailment, suspension or stoppage of an Essential Service that reasonably prevents Tenant from using the entire Premises, or any material portion thereof, as a result of Landlord's failure to provide any Essential Service (defined below) required to be provided by Landlord under this Lease, provided that such failure or Landlord's inability to remedy such event or circumstance is not due to Force Majeure or a cause beyond Landlord's reasonable control generally affecting other buildings in the vicinity (such as a neighborhood power outage or other off-site occurrence) or the act or negligence of Tenant, its employees, vendors, or contractors, or any party claiming by, through or under Tenant then Tenant shall give Landlord notice (an "Abatement Notice") of any such Abatement Event. If such Abatement Event continues beyond the "Eligibility Period" (defined below), then the Monthly Fixed Rent and monthly charges on account of Operating Costs Excess and Tax Excess shall be abated entirely or proportionately, as the case may be, after the expiration of the Eligibility Period for such time that Tenant continues to be so prevented from using, and does not use, the Premises or a portion thereof, in the proportion that the rentable area of the portion of the Premises that Tenant is prevented from using, and does not use, bears to the total rentable area of the Premises. Tenant shall not be entitled to any abatement of Rent if Tenant is then in default beyond notice and cure periods of any of its obligations under this Lease (it being agreed that if Landlord has given notice of default for which the applicable cure period has yet to lapse,

Tenant must cure such default within such cure period for Tenant to be entitled to the abatement right set forth herein). The term "Eligibility Period" shall mean a period of five

- (5) consecutive business days after Landlord's receipt of any Abatement Notice(s). For purposes hereof, the term "Essential Services" shall mean the following services required to be provided by Landlord under this Lease: water and sewer service, HVAC, gas and electricity, and elevator service. The provisions of this paragraph shall not apply to any Abatement Event caused by fire or other damage or destruction to the Building, which shall be covered by Article 7 of this Lease, and shall be Tenant's sole remedy on account of an Abatement Event.
- 5.3 <u>Outside Services.</u> In the event Tenant wishes to provide outside services for the Premises over and above those services to be provided by Landlord as set forth herein, Tenant shall first obtain the prior written approval of Landlord for the installation and/or utilization of such services ("Outside services" shall include, but shall not be limited to, cleaning services, television, so-called "canned music" services, security services). In the event Landlord approves the installation and/or utilization of such services, such installation and utilization shall be at Tenant's sole cost, risk and expense.
- 5.4 <u>Landlord's Insurance</u>. During the term, Landlord shall keep in effect All Risk Property insurance with the usual extended coverage endorsements covering the lesser of (a) the full replacement cost of the Building (exclusive of excavation and foundation), including Landlord's Work (but not any of Tenant's personal property, nor any alteration or improvement outside of Landlord's Work installed by or on behalf of Tenant) and (b) such amounts required of Landlord by any mortgage holder of the Property.
- 5.5 <u>Indemnification.</u> Landlord shall save harmless, exonerate and indemnify Tenant, agents, partners, officers, directors, members, trustees, beneficiaries, shareholders, and employees (collectively "Tenant Related Parties") from and against any and all claims, liabilities, costs, expenses (including reasonable attorney's fees) or penalties asserted by or on behalf of any person, firm, corporation or public authority on account of injury, death, damage or loss to person or property in or upon the common area of the Building and Property arising out of the gross negligence or willful misconduct of Landlord, except if the same was caused by the negligence, fault or misconduct of Tenant or the Tenant Related Parties. In respect of all of the foregoing, Landlord shall indemnify Tenant and the Tenant Related Parties from and against all costs, expenses (including reasonable attorneys' fees), and liabilities incurred in or in connection with any such claim, action or proceeding brought thereon; and, in case of any action or proceeding brought against Tenant or the Tenant Related Parties by reason of any claim for which Landlord is obligated to indemnify Tenant and the Tenant Related Parties as set forth above, Landlord, upon notice from Tenant and at Landlord's expense, shall resist or defend such action or proceeding and employ counsel therefor reasonably satisfactory to Tenant provided that Tenant shall be deemed to have approved counsel provided by Landlord's liability insurer. The preceding indemnification shall expressly survive the expiration or earlier termination of this Lease.

ARTICLE 6 Tenant's Additional Covenants

- 6.1 <u>Affirmative Covenants.</u> Tenant covenants at all times during the Term and for such further time (prior or subsequent thereto) as Tenant occupies the Premises or any part thereof:
 - 6.1.1 <u>Perform Obligations.</u> To perform promptly all of the obligations of Tenant set forth in this Lease; and to pay when due the Fixed Rent and Additional Rent and all charges, rates and other sums which by the terms of this Lease are to be paid by Tenant.
 - 6.1.2 <u>Use.</u> To use the Premises only for the Permitted Uses, and from time to time to procure all licenses and permits necessary therefor, at Tenant's sole expense. With respect to any licenses or permits for which Tenant may apply, pursuant to this subsection 6.1.2 or any other provision hereof, Tenant shall furnish Landlord copies of applications therefor on or before their submission to the governmental authority.
 - Repair and Maintenance. To maintain the Premises in neat order and condition and to perform all routine and 6.1.3 ordinary repairs to the Premises (excepting only those repairs for which Landlord is responsible under the terms of this Lease) and to any plumbing, heating, electrical, ventilating and air- conditioning systems located within the Premises and installed by Tenant (or exclusively serving Tenant) such as are necessary to keep them in good working order, appearance and condition, as the case may require, reasonable use and wear thereof and damage by fire or by casualty excepted; to keep all glass in windows, doors and plate glass in the Premises (except glass in the exterior walls of the Building) whole and in good condition with glass of the same quality as that injured or broken; and to make as and when needed as a result of misuse by, or neglect or improper conduct of Tenant or Tenant's servants, employees, agents, invitees or licensees or otherwise, all repairs necessary, which repairs and replacements shall be in quality and class equal to the original work. Subject to the terms of Section 8.4 below, Landlord, upon default of Tenant hereunder beyond notice and cure periods (except in the case of potential injury and/or damage to persons and/or property, or where appearance and/or functionality is compromised in Landlord's good faith judgment) and upon prior notice to Tenant, may elect, at the expense of Tenant, to perform all such cleaning and maintenance and to make any such repairs or to repair any damage or injury to the Building or the Premises caused by moving property of Tenant in or out of the Building, or by installation or removal of furniture or other property, or by misuse by, or neglect, or improper conduct of, Tenant or Tenant's servants, employees, agents, contractors, customers, patrons, invitees, or licensees.
 - 6.1.4 Compliance with Law. To make all repairs, alterations, additions or replacements

to the Premises required by any law or ordinance or any order or regulation of any public authority; to keep the Premises equipped with all safety appliances so required; and to comply with the orders and regulations of all governmental authorities with respect to zoning, building, fire, health and other codes, regulations, ordinances or laws applicable to the Premises, except that Tenant may defer compliance so long as the validity of any such law, ordinance, order or regulations shall be contested by Tenant in good faith and by appropriate legal proceedings, if Tenant first gives Landlord appropriate assurance or security against any loss, cost or expense on account thereof.

- 6.1.5 *Indemnification.* To indemnify, defend and hold harmless Landlord, its agents (including, without limitation, Landlord's managing agent), partners, officers, directors, members, trustees, beneficiaries, shareholders, and employees (such parties being referred to collectively as the "Landlord Related Parties") from and against any and all claims, demands, liabilities, penalties, fines, settlements, damages, loss, costs or expenses resulting from, arising out of, or in any way related to injury, death, damage or loss to person or property in or upon the Premises and the Property arising out of each of the following (a) the use or occupancy of the Premises by Tenant or by any person claiming by, through or under Tenant (including, without limitation, all patrons, employees and customers of Tenant), and/or (b) the negligent acts or omissions or intentional misconduct of Tenant or any person claiming by, through or under Tenant, and/or (c) on account of any breach by Tenant of its obligations under this Lease, and/or (d) on account of or based upon anything whatsoever done on the Premises, except if the same was caused by the gross negligence or willful misconduct of Landlord or the Landlord Related Parties. In respect of all of the foregoing, Tenant shall indemnify Landlord and the Landlord Related Parties from and against all costs and expenses (including reasonable attorneys' fees), of whatever kind or nature incurred in or in connection with any such claim, action or proceeding brought thereon; and, in case of any action or proceeding brought against Landlord or the Landlord Related Parties by reason of any such claim, Tenant, upon notice from Landlord and at Tenant's expense, shall resist or defend such action or proceeding and employ counsel therefor reasonably satisfactory to Landlord. The preceding indemnification shall expressly survive the expiration or earlier termination of this Lease.
- 6.1.6 <u>Landlord's Right to Enter.</u> To permit Landlord and its agents to enter into and examine the Premises at reasonable times and upon reasonable prior notice (except in the case of emergency) and to show the Premises, and to make repairs to the Premises, and, during the last six (6) months prior to the expiration of this Lease, to keep affixed in suitable places notices of availability of the Premises.
- 6.1.7 <u>Personal Property at Tenant's Risk.</u> All of the furnishings, fixtures, equipment, effects and property of every kind, nature and description of Tenant and of all persons claiming by, through or under Tenant which, during the continuance of this Lease or any occupancy of the Premises by Tenant or anyone claiming under Tenant, may be on the Premises, shall be at the sole risk and hazard of Tenant and if the whole or any part thereof shall be destroyed or damaged by fire, water, rain or by the leakage or bursting or other defect of water pipes, steam pipes, or other pipes, sprinklers, lighting fixtures or other cause by theft, any acts or omissions of any other tenant of the Property, or from any other cause, no part of said loss or damage is to be charged to or to be borne by Landlord, except that Landlord shall in no event be indemnified or held harmless or exonerated from any liability to Tenant or to any other person, for any injury, loss, damage or liability to the extent prohibited by law.
- 6.1.8 <u>Payment of Landlord's Cost of Enforcement.</u> To pay on demand Landlord's expenses, including reasonable attorneys' fees, incurred in enforcing any obligation of Tenant under this Lease or in curing any default by Tenant under this Lease as provided in Section 8.4.
- 6.1.9 <u>Yield Up.</u> (A) At the expiration of the Term or earlier termination of this Lease: to surrender all keys to the Premises; to remove all of its trade fixtures and personal property in the Premises; to deliver to Landlord stamped architectural plans showing the Premises at yield up (which may be the

initial plans if Tenant has made no installations after the Commencement Date); to remove such alterations, improvements, or installations made by or on behalf of Tenant (including any of the alterations, improvements or installations made by or on behalf of Tenant pursuant to Article 3 above) as Landlord may request (including computer and telecommunications wiring and cabling, it being understood that if Tenant leaves such wiring and cabling in a useable condition, Landlord, although having the right to request removal thereof, is less likely to so request) and all Tenant's signs wherever located; to repair all damage caused by such removal and to yield up the Premises (including all installations and improvements made by Tenant except for trade fixtures and such of said installations or improvements as Landlord shall request Tenant to remove), broom-clean and in the same good order and repair in which Tenant is obliged to keep and maintain the Premises by the provisions of this Lease. Notwithstanding the preceding provisions of this Section 6.1.9 or anything to the contrary contained in this Lease, except for voice and data wiring and cabling installed by or on behalf of Tenant, Landlord shall, subject to the terms and conditions below, not require removal of any alterations or installations made by or for Tenant except for those items that are not customary office improvements ("Non-Standard Improvements"). For purposes of this paragraph, "customary office improvements" shall mean leasehold improvements similar to those customarily made in connection with general office uses. Non-Standard Improvements shall include, without limitation, raised floors, supplemental HVAC systems and equipment, any modification to any portion of the Building's systems, the installation of a vault or other similar device or system that is intended to secure the Premises or a portion thereof in a manner that exceeds the level of security used for ordinary office space, any plumbing connections in excess of those customarily made in connection with general office use (such as break stations and executive bathrooms), and any alterations altering or modifying the structural elements of the Building, including internal stairs. Tenant, at the time of requesting Landlord's approval of any installations or alterations (and also in connection with Landlord's Work), shall request in writing that Landlord designate whether the proposed installation or alteration is a Non- Standard Improvement that must be removed at the expiration or earlier termination of this Lease. After receipt of Tenant's request, Landlord shall notify Tenant in writing if any item is a Non-Standard Improvement that must be removed from the Premises at the expiration or earlier termination of this Lease. If Landlord does not indicate that a particular Non-Standard Improvement must be removed, then Landlord shall not later have the right to request its removal. Likewise, if Tenant, when making any installations or alterations (including but not limited to Landlord's Work), fails to so request in writing whether an item is a Non-Standard Improvement that Landlord will require removal of, then Landlord reserves the right to require Tenant to remove the installation or alteration at the end of the Term. For the avoidance of doubt, any portion of Landlord's Work that is a Non- Standard Improvement is subject to the removal requirement set forth above. Any property not so removed shall be deemed abandoned and, if Landlord so elects, deemed to be Landlord's property, and may be retained or removed and disposed of by Landlord in such manner as Landlord shall determine and Tenant shall pay Landlord the entire cost and expense incurred by it in effecting such removal and disposition and in making any incidental repairs and replacements to the Premises and for use and occupancy during the period after the expiration of the term and prior to its performance of its obligations under this subsection 6.1.9. Tenant shall further indemnify Landlord against all loss, cost and damage resulting from Tenant's failure and delay in surrendering the Premises as above

provided (it being agreed Tenant shall not be liable for consequential or indirect damages unless Tenant holds over for thirty (30) or more days).

- (B) If the Tenant remains in the Premises beyond the expiration or earlier termination of this Lease, such holding over shall be without right and shall not be deemed to create any tenancy, but the Tenant shall be a tenant at sufferance only at a daily holdover rate of rent equal to 200% the Monthly Fixed Rent Rate and 100% of Additional Rent on account of Operating Costs and Taxes last due as of the day prior to the date of expiration or earlier termination of this Lease, and shall otherwise be on the terms and conditions of this Lease as applicable, except that in no event shall any extension option, right of first offer or right of first refusal, or similar right or option be deemed applicable to such tenancy at sufferance.
- (C) Notwithstanding subparagraph (B) above, on the condition that (i) Tenant has given Landlord a notice of its intent to remain in the Premises beyond the expiration or termination of this Lease (a "Holdover Notice") no earlier than one and twenty (120) days and no later than ninety (90) days prior to the Expiration Date, and (ii) Landlord does not then have a signed letter of intent or lease with another tenant for the Premises, or any portion thereof, at the time Tenant gives the Holdover Notice, then if the Tenant remains in the Premises beyond the expiration or earlier termination of this Lease, such holding over shall be without right and shall not be deemed to create any tenancy and Tenant shall be a tenant at sufferance at a daily holdover rate of 150% of the Fixed Rent and 100% of Additional Rent on account of Operating Costs and Taxes last due at the end of the Term, but only for a period not to exceed thirty (30) days. If Tenant continues to holdover in the Premises following the end of that initial thirty (30) day period, then the holdover rate shall increase to 200% of the Fixed Rent and 100% of Additional Rent on account of Operating Costs and Taxes. In no way shall the terms of this subparagraph (C) be applicable in the event that Landlord exercises Landlord's Termination Right under Section 2.4 (it being agreed any holding over beyond the Early Termination Date shall be governed solely by the terms of subparagraphs (A) and (B) above).
- 6.1.10 <u>Rules and Regulations.</u> To comply with the Rules and Regulations set forth in Exhibit F, and with all reasonable Rules and Regulations of general applicability to all tenants of the Building hereafter made by Landlord, of which Tenant has been given notice; Landlord shall not be liable to Tenant for the failure of other tenants of the Building to conform to such Rules and Regulations. Tenant shall cause all parties performing work on behalf of Tenant in or about the Premises and Building to comply with the Construction Rules and Regulations attached hereto as Exhibit F-1.
- 6.1.11 <u>Estoppel Certificate.</u> Upon not less than ten (10) days' prior written request by Landlord, to execute, acknowledge and deliver to Landlord a statement in writing, which may be in the form attached hereto as Exhibit G or in another form reasonably similar thereto, or such other commercially reasonable form as Landlord may provide from time to time, certifying all or any of the following: (i) that this Lease is unmodified and in full force and effect (if such statement is true), (ii) whether the term has commenced and Fixed Rent and Additional Rent have become payable hereunder and, if so, the dates to which they have been paid, (iii) whether or not Landlord is in default in performance of any of the terms of this Lease, (iv) whether Tenant has accepted possession of the Premises, (v) whether Tenant has made any claim against Landlord under this Lease and, if so, the nature thereof and the dollar amount, if any, of such claim, (vi) whether there exist any offsets or defenses against enforcement of any of the terms of this Lease upon the part of Tenant to be performed, and (vii) such further information with respect to the Lease or the Premises as Landlord may

reasonably request. Any such statement delivered pursuant to this subsection 6.1.11 may be relied upon by any prospective purchaser or mortgagee of the Premises, or any prospective assignee of such mortgage. Tenant shall also deliver to Landlord such financial information as may be reasonably required by Landlord to be provided to any mortgagee or prospective purchaser of the Premises. If Tenant fails to deliver the estoppel certificate within the required time period, and such failure continues for an additional five (5) days following a second written request from Landlord, then Tenant shall be obligated to pay to Landlord, as Additional Rent within twenty (20) days of demand, a fee in the amount of \$500.00 per day for each day that Tenant fails to deliver the requested estoppel in the period beginning on the day after the expiration of the initial 10-day period, and ending on the day Tenant actually delivers the estoppel.

- 6.1.12 <u>Landlord's Expenses Re Consents.</u> To reimburse Landlord within ten (10) days of demand for all reasonable out-of-pocket legal expenses incurred by Landlord in connection with all requests by Tenant for consent or approval hereunder.
- 6.1.13 *Guaranty*. Upon Tenant's execution of this Lease, Tenant shall cause to be delivered to Landlord a guaranty agreement executed by Guarantor, on Landlord's standard form, to secure the performance of Tenant's obligations under this Lease.
- 6.2 <u>Negative Covenants.</u> Tenant covenants at all times during the Term and such further time (prior or subsequent thereto) as Tenant occupies the Premises or any part thereof:
 - 6.2.1 <u>Assignment and Subletting.</u> (a) Except in connection with a Permitted Transfer (defined in Section 6.2.1(e) below, not to assign, transfer, mortgage or pledge this Lease or to sublease (which term shall be deemed to include the granting of concessions and licenses and the like) all or any part of the Premises or permit this Lease or the leasehold estate hereby created or any other rights arising under this Lease to be assigned, transferred, pledged or encumbered, in whole or in part, whether voluntarily, involuntarily or by operation of law, or permit the occupancy of the Premises by anyone other than Tenant without the prior written consent of Landlord. In the event Tenant desires to assign this Lease or sublet any portion or all of the Premises, Tenant shall notify Landlord in writing of Tenant's intent to so assign this Lease or sublet the Premises and the proposed effective date of such subletting or assignment, and shall request in such notification that Landlord consent thereto which consent shall not be unreasonably withheld, conditioned or delayed to an assignment or to a subletting, provided that the following conditions are met:
 - (i) the assignee or subtenant shall use the Premises only for the Permitted Uses;
 - (ii) with respect to a subletting, that after such subletting the initial Tenant named herein occupies at least fifty (50%) percent of the Rentable Floor Area of the Premises;

- (iii) the proposed assignee or subtenant has a net worth and creditworthiness (aa) equal to or greater than Tenant's as of the Effective Date or (bb) otherwise reasonably acceptable to Landlord;
- (iv) the amount of the aggregate rent to be paid by the proposed subtenant is not less than [***] of the then current sublease market rate for the Premises; and
- (v) the proposed assignee or subtenant is not then a tenant in the Building or an entity with which Landlord is dealing or has dealt within the preceding six months regarding the possibility of leasing space in the Building.
- (vi) the proposed assignee or subtenant provides the representation and warranty required under Section 10.12 below (OFAC Compliance).

Tenant shall furnish Landlord with any information reasonably requested by Landlord to enable Landlord to determine whether the proposed assignment or subletting complies with the foregoing requirements, including without limitation, financial statements relating to the proposed assignee or subtenant, and, within thirty (30) days following receipt of all such items as set forth above and/or reasonably requested by Landlord, Landlord shall inform Tenant as to whether Landlord consents to, or denies consent to, such proposed sublet or assignment. With respect to any assignment or subletting during the Original Term of this Lease, such assignment shall not include the Extension Option granted to Tenant, and such sublease shall be for a term expiring no later than the Expiration Date.

- (b) Tenant shall, as Additional Rent, reimburse Landlord promptly for Landlord's reasonable out-of-pocket legal expenses incurred in connection with any request by Tenant for such consent. If Landlord consents thereto, no such subletting or assignment shall in any way impair the continuing primary liability of Tenant hereunder, and no consent to any subletting or assignment in a particular instance shall be deemed to be a waiver of the obligation to obtain the Landlord's written approval in the case of any other subletting or assignment.
- (c) If for any assignment or sublease consented to by Landlord hereunder Tenant receives rent or other consideration, either initially or over the term of the assignment or sublease, in excess of the rent called for hereunder, or in case of sublease of part, in excess of such rent fairly allocable to the part, after appropriate adjustments to assure that all other payments called for hereunder are appropriately taken into account and after first deducting the following costs (if any) incurred by Tenant in connection with such sublease or assignment: (i) any commercially reasonable alterations to the Premises made by Tenant, (ii) any commercially reasonable advertising or marketing expenses incurred by Tenant, (iii) any commercially reasonable improvement allowance or rent concessions provided to any transferee, and (iv) any commercially reasonable brokerage commissions, and
- (v) any commercially reasonable out-of-pocket attorney's fees (all of which shall be amortized on a straight line basis over the term of any assignment or sublease), to pay to Landlord as additional rent fifty (50%) percent of the excess of each such payment of rent or other consideration received by Tenant promptly after its receipt.
- (d) Any other provision of this Section 6.2.1 notwithstanding, Tenant shall have the right, without Landlord's prior written consent and without any right of Landlord to terminate this Lease, (i) to assign this Lease or sublease the Premises to an entity which is either a parent, subsidiary or affiliate of Tenant (an "Affiliate"), provided Tenant remains primarily liable under this Lease, notifies Landlord of the assignment or subletting and delivers reasonable proof of the

affiliation to Landlord, and (ii) to assign this Lease to an entity into or with which Tenant is merged or consolidated, whether or not Tenant is the surviving entity, or to an entity to which all or substantially all of Tenant's stock or assets are transferred, provided that in any of such events described in (i) and (ii) of this Section 6.2.1(d), the successor entity or the Affiliate, as applicable, has a net worth (computed in accordance with generally accepted accounting principles) immediately after the transfer, assignment or subletting (as applicable) at least equal to the net worth of Tenant either as of the Effective Date or immediately prior to such merger, consolidation, transfer, assignment or subletting, whichever is greater, and proof reasonably satisfactory to Landlord of such net worth shall have been delivered to Landlord at least ten (10) days prior to the effective date of any such transaction (except to the extent prohibited by applicable securities laws or regulations, in which case notice shall be given as soon as permissible under such laws or regulations). A transaction for which consent from the Landlord is not required pursuant to this paragraph (d) shall be referred to in this Lease as a "Permitted Transfer." In the event of any Permitted Transfer, Tenant shall remain primarily liable under this Lease and any assignee shall agree directly with Landlord, by written instrument in form satisfactory to Landlord, to assume and perform all the obligations of Tenant under this Lease.

- (e) If at any time during the Term of this Lease, there is a name change, Tenant shall so notify Landlord and deliver evidence reasonably satisfactory to Landlord documenting such name change. If, at any time during the Term of this Lease, there is a transfer of a controlling interest in the shares or stock which are not publicly traded upon a stock exchange, or the membership or general partnership or other ownership interests of Tenant, or a restructuring or reorganization of the Tenant entity, including any spin-off, Tenant shall so notify Landlord and (whether or not Tenant so notifies Landlord) such transfer, restructuring or reorganization shall be deemed an assignment of this Lease requiring Landlord's consent as provided in this Section 6.2.1. "Controlling interest" shall mean having ownership of fifty percent (50%) or more of the outstanding voting stock of a corporation or other majority equity and control interest if not a corporation and the possession of power to direct or cause the direction of the management of such corporation or other entity.
- (f) The following terms and conditions shall apply to any subletting by Tenant of all or any part of the Premises and shall be deemed included in all subleases under this Lease whether or not expressly incorporated therein:

- Tenant hereby assigns and transfers to Landlord all of Tenant's interest in all rentals and income arising (i) from any sublease of all or a portion of the Premises heretofore or hereafter made by Tenant, and Landlord may collect such rent and income and apply same toward Tenant's obligations under this Lease; provided, however, that until a default by Tenant hereunder that is not cured within applicable notice and cure periods shall occur under this Lease, Tenant may, except as otherwise provided in this Lease, receive, collect and enjoy the rents accruing under such sublease. Landlord shall not, by reason of this or any other assignment of such sublease to Landlord, nor by reason of the collection of the rents from a subtenant, be deemed liable to the subtenant for any failure of Tenant to perform and comply with any of Tenant's obligations to such subtenant under such sublease. Tenant hereby irrevocably authorizes and directs any such subtenant, upon receipt of a written notice from Landlord stating that a default hereunder exists under this Lease, to pay to Landlord the rents and other charges due and to become due under the sublease. The subtenant shall rely upon any such statement and request from Landlord and shall pay such rents and other charges to Landlord without any obligation or right to inquire as to whether such breach exists and notwithstanding any notice from or claim from Tenant to the contrary. Tenant shall have no right or claim against said subtenant, or, until the breach has been cured, against Landlord, for any such rents and other charges so paid by said subtenant to Landlord.
- (ii) In the event of a default by Tenant in the performance of its obligations under this Lease that results in a termination of this Lease, Landlord, at its option and without any obligation to do so, may require any subtenant to attorn to Landlord, in which event Landlord shall undertake the obligations of the sublandlord under such sublease from the time of the exercise of said option to the expiration of such sublease; provided, however, Landlord shall not be liable for any prepaid rents or security deposit paid by such subtenant to Tenant or for any other prior defaults or breaches of Tenant as sublandlord under such sublease.
- 6.2.2 <u>Nuisance.</u> Not to injure, deface or otherwise harm the Premises; nor commit any nuisance; nor permit in the Premises any vending machine (except such as is used for the sale of merchandise to employees of Tenant) or inflammable fluids or chemicals (except such as are customarily used in connection with standard office equipment); nor permit any cooking to such extent as requires special exhaust venting; nor permit the emission of any objectionable noise or odor; nor make, allow or suffer any waste; nor make any use of the Premises which is improper, offensive or contrary to any law or ordinance or which will invalidate any of Landlord's insurance; nor conduct any auction, fire, "going out of business" or bankruptcy sales.
- 6.2.3 <u>Hazardous Wastes and Materials</u>. Not to cause or permit any Hazardous Materials to be used, handled, generated, stored or disposed of by Tenant, or persons or entities acting by, through, or on behalf of Tenant, on, under or above, or transported to or from, the Premises and/or the Property (collectively, "Hazardous Materials Activities"). Nothing contained herein shall be deemed to prevent Tenant from using de minimus quantities of commercially available cleaners and office supplies which are customarily used in the ordinary course of first-class business office operations, which cleaners and/or office supplies contain Hazardous Materials; provided that, Tenant shall use such cleaners and/or office supplies in strict compliance (at Tenant's sole cost and expense) with all applicable laws, and shall use all necessary and appropriate precautions to prevent any spill, discharge,

release or exposure to persons or property. Landlord shall not be liable to Tenant for any loss, cost, expense, claim, damage or liability arising out of any Hazardous Materials Activities by Tenant, or by Tenant's employees, agents, contractors, licensees, customers or invitees, whether or not consented to by Landlord. Tenant shall indemnify, defend with counsel acceptable to and approved by Landlord, and hold Landlord and all Landlord Affiliates harmless from and against any and all losses, costs, expenses (including, without limitation, all reasonable attorneys' fees), claims, damages, obligations and liabilities arising out of: (i) any Hazardous Materials Activities on the Premises, whether or not consented to by Landlord; (ii) any Hazardous Materials Activities by Tenant, Tenant's employees, agents, contractors, licensees, customers or invitees or anyone claiming by, through or under Tenant, wherever occurring; and (iii) any contamination, claim of contamination, loss or damage, or the like arising out of or resulting from the foregoing. For purposes hereof, "Hazardous Materials" shall include but not be limited to substances defined as "hazardous substances," "toxic substances" or "hazardous wastes" or "oil" in any local, state or federal law, rule, regulation or ordinance (collectively, "Environmental Law(s)"). If Landlord consents to any Hazardous Materials Activities, prior to using, storing or maintaining any Hazardous Materials on or about the Premises, Tenant shall provide Landlord with a list of the types and quantities thereof, and shall update such list from time-to-time as necessary for continued accuracy. Tenant shall also provide Landlord with a copy of any Hazardous Materials inventory statement and any updates thereof required by any applicable Environmental Laws. If Tenant's activities violate or create a risk of violation of any Environmental Law or cause a spill, discharge, release or exposure to any persons or property, Tenant shall cease such activities immediately. Tenant shall immediately notify Landlord both by telephone and in writing of any spill, discharge, release or exposure of Hazardous Materials in or about the Premises, or of any condition in or about the Premises constituting an "imminent hazard" under any Environmental Laws. Landlord, Landlord's representatives and employees may enter the Premises during the Term to inspect Tenant's compliance herewith, and may disclose any spill, discharge, release, or exposure or any violation of any Environmental Laws to any applicable governmental agencies or authorities. Landlord represents to Tenant, to Landlord's actual knowledge as of the date of this Lease, that there are no hazardous materials in, on, about, under or emanating from the Premises in violation of any laws. The preceding indemnification contained in this Section 6.2.3 shall survive the expiration or earlier termination of this Lease.

- 6.2.4 Floor Load; Heavy Equipment. Not to place a load upon any floor of the Premises exceeding the floor load per square foot area which such floor was designed to carry and which is allowed by law. Landlord reserves the right to prescribe the weight and position of all heavy business machines and equipment, including safes, which shall be placed so as to distribute the weight. Business machines and mechanical equipment which cause vibration or noise shall be placed and maintained by Tenant at Tenant's expense in settings sufficient to absorb and prevent vibration, noise and annoyance. Tenant shall not move any safe, heavy machinery, heavy equipment, freight or fixtures into or out of the Premises except in such manner and at such time as Landlord shall in each instance authorize.
- 6.2.5 <u>Installation, Alterations or Additions.</u> (A) Not to make any installations, alterations or additions in, to or on the Premises nor to permit the making of any holes in the walls (except in the case of hanging artwork, monitors, screens or any other items provided that any of such items are of a size and

weight customarily found in a typical office setting), partitions, ceilings or floors nor the installation or modification of any locks or security devices without on each occasion obtaining the prior written consent of Landlord, and then only pursuant to plans and specifications approved by Landlord in advance in each instance. Notwithstanding the foregoing, Tenant, upon no less than ten (10) days prior notice to Landlord but without the prior consent of Landlord, may make cosmetic nonstructural, interior alterations or installations at a cost not to [***] per alteration/installation project in any 12-month period, only if such alterations or installations do not adversely affect the structural integrity of the Building, impact the Building systems, involve penetration of the roof or exterior walls, or require a building permit. In no event shall Landlord's approval of any plans, including plans for any initial tenant improvements in accordance with Article 3, be construed as a waiver of Landlord's right, pursuant to Section 6.1.9 above, to require removal of any Non-Standard Improvements at end of the Term. Tenant agrees to employ for any work one or more responsible contractors of whom Landlord has given prior approval, and whose labor will work without interference with other labor working on the Property, and to cause such contractors employed by Tenant to carry builder's risk insurance as set forth in Section 4.2.4.1, worker's compensation insurance in accordance with statutory requirements, Employers Liability Insurance at least equal to the limits set forth in Section 4.2.4.1, and commercial general liability insurance covering such contractors on or about the Premises in amounts at least equal to the limits set forth in Section 1.1. All contractors insurance shall name Landlord and its managing agent and any mortgagee as additional insureds on a primary and non-contributory basis, and indemnifying the parties so named against claims for death or injury to persons or damage to property claimed to have occurred in the Premises or on the property. Tenant shall cause its contractors to submit certificates evidencing such coverage to Landlord prior to the commencement of any such work. Tenant shall cause all contractors performing work on behalf of Tenant in or about the Premises and Building to comply with the Construction Rules and Regulations attached hereto as Exhibit F-1. Tenant shall pay promptly when due the entire cost of any work to the Premises undertaken by Tenant so that the Premises shall at all times be free of liens for labor and materials, and at Landlord's request Tenant shall furnish to Landlord a bond or other security acceptable to Landlord assuring that any work commenced by Tenant will be completed in accordance with the plans and specifications theretofore approved by Landlord and assuring that the Premises will remain free of any mechanics' lien or other encumbrance arising out of such work. In any event, Tenant shall within ten (10) days of notice from Landlord bond against or discharge any mechanics' liens or other encumbrances that may arise out of such work. If Tenant shall fail to cause any such lien to be discharged within such 10 day period, then in addition to any other available right or remedy, Landlord may discharge the same, either by paying the amount claimed to be due, or by bonding or otherwise. Any amount so paid, and all costs and expenses so incurred by Landlord in connection therewith, shall constitute Additional Rent hereunder. Tenant shall procure all necessary licenses and permits at Tenant's sole expense before undertaking such work. All such work shall be done in a good and workmanlike manner employing materials of good quality and so as to conform with all applicable zoning, building, fire, health and other codes, regulations, ordinances and laws. Tenant shall save Landlord harmless and indemnified from all injury, loss, claims or damage to any person or property occasioned by or growing out of such work, and any liability, loss, cost, damage and expense of every kind and nature incurred by reason of, or arising out of any and all

mechanic's and other liens filed in connection with any alterations or improvements.

- (B) Not to grant a security interest in, or to lease, any personal property or equipment being installed in the Premises, including, without limitation, demountable partitions (the "Collateral") without first obtaining an agreement for the benefit of Landlord in the form attached hereto as Exhibit G, from the secured party or lessor ("Secured Party") that stipulates in the event either the Lease is terminated or Tenant defaults in its obligations to Secured Party, then (i) Secured Party will remove the Collateral within ten (10) business days after notice from Landlord of the expiration or earlier termination of this Lease, or within ten (10) business days after Secured Party notifies Landlord that Secured Party has the right to remove the Collateral on account of Tenant's default in its obligations to Secured Party, (ii) Secured Party will restore the area affected by such removal, and (iii) that a failure to so remove the Collateral will subject such property to the provisions of subsection 6.1.9 of the Lease.
- 6.2.6 <u>Abandonment.</u> Not to abandon the Premises during the term, it being understood and agreed that vacancy of the Premises shall not be construed as abandonment so long as all of Tenant's other obligations under this Lease, including payment of rent and all other sums owing to Landlord, continue to be timely performed and reasonable and diligent measures are taken by Tenant to manage the vacant space..
- 6.2.7 <u>Signs.</u> Not without Landlord's prior written approval to paint or place any signs or place any curtains, blinds, shades, awnings, aerials, or the like, visible from outside the Premises. Landlord, at its sole expense, shall install Tenant's business name in the lobby of the South Building and on the monument sign located outside the main driveway entrance to the Property on Burlington Road, using Building standard signage. Tenant shall have the exclusive right to install, at its sole expense, its business identification sign (which may be lighted) on the exterior façade of the South Building in a mutually agreed to location. All exterior signage shall be subject to Landlord's prior approval of the specifications, location and manner of installation, and shall be in compliance with all requirements of the Town of Bedford. Tenant shall be solely responsible for obtaining, at its sole cost, all approvals and permits that may be required by the Town of Bedford. Tenant shall maintain its sign in good condition and repair, at its sole expense and shall remove the same at the end of the Term.
- 6.2.8 <u>Parking and Storage</u>. Not to permit any storage of materials outside of the Premises; nor to permit the use of the parking areas for either temporary or permanent storage of trucks; nor permit the use of the Premises for any use for which heavy trucking would be customary.

ARTICLE 7 Casualty or Taking

7.1 <u>Termination.</u> In the event that the Premises or the Building, or any material part thereof, shall be taken by any public authority or for any public use, or shall be destroyed or damaged by fire or casualty, or by the action of any public authority, then this Lease may be terminated at the election of Landlord. Such election, which may be made notwithstanding the fact that Landlord's entire interest may have been divested, shall be made by the giving of notice by Landlord to Tenant within sixty (60) days after the date of the taking or casualty. In the event that the Premises are substantially destroyed or damaged by fire or casualty, or by the action of public authority, and, in the reasonable opinion of an independent architect or engineer selected by Landlord, cannot be repaired or restored within two hundred and seventy (270) days from the date Landlord receives

the insurance proceeds for the repair or restoration work, then this Lease may be terminated at the election of Landlord or Tenant, which election shall be made by the giving of notice to the other party within thirty (30) days after the date the opinion of the architect or engineer is made available to the parties, which notice shall specify the effective date of termination, which shall not be less than thirty (30) nor more than sixty (60) days after the date of such parties' termination notice. Landlord shall endeavor to deliver the architect's opinion (the "Repair and Restoration Opinion") to Tenant within forty-five (45) days following the casualty or taking.

- 7.2 <u>Restoration.</u> If neither Landlord nor Tenant elects to terminate the Lease as provided for in Section 7.1 above, then this Lease shall continue in force and commencing on the date of such casualty and/or taking a just proportion of the rent reserved, according to the nature and extent of the damages sustained by the Premises, shall be suspended or abated until the Premises, or what may remain thereof, shall be put by Landlord in proper condition for use, which Landlord covenants to do with reasonable diligence to the extent permitted by the net proceeds of insurance recovered or damages awarded for such taking, destruction or damage and subject to zoning and building laws or ordinances then in existence. "Net proceeds of insurance recovered or damages awarded" refers to the gross amount of such insurance or damages less the reasonable expenses of Landlord incurred in connection with the collection of the same, including without limitation, fees and expenses for legal and appraisal services. Subject to the provisions of Section 10.5, if Landlord shall not have restored the Premises, or in the case of a taking, what may remain thereof (excluding any alterations or installations made by or on behalf of Tenant, including the initial tenant improvements) before the later to occur of: (i) two hundred and seventy (270) days from date Landlord receives the insurance proceeds for the repair or restoration work or (ii) the date corresponding to the estimated amount of time set forth in the Repair and Restoration Opinion for repair from the date Landlord receives the insurance proceeds (the later of (i) and (ii) the "Restoration Deadline"), Tenant shall have the right to terminate this Lease by giving notice of such termination to Landlord, effective at the expiration of thirty (30) days from the giving of such notice; provided however, that such termination will be rendered ineffective if, prior to the expiration of said 30-day period, Landlord shall have completed such restoration. The foregoing termination right shall be Tenant's sole remedy at law or in equity for Landlord's failure to restore the Premises on or before the Restoration Deadline.
- Award. Irrespective of the form in which recovery may be had by law, all rights to damages or compensation shall belong to Landlord in all cases. Tenant hereby grants to Landlord all of Tenant's rights to such damages and covenants to deliver such further assignments thereof as Landlord may from time to time request. Tenant shall have the right to any separate award made by a public authority as may be specifically awarded to Tenant for Tenant's personal property and relocation costs and moving fees in connection with a taking, but only to the extent (i) such amounts are awarded in a separate proceeding and
 - (ii) the award available to Landlord is not reduced or diminished in any way by such award being granted to Tenant.

ARTICLE 8 <u>Defaults</u>

8.1 <u>Events of Default.</u> (a) If Tenant shall default in the performance of any of its obligations to pay the Fixed Rent, Additional Rent or any other sum due Landlord hereunder, or if Tenant shall fail to maintain the insurance required under Section 4.2.4 of this Lease, and if such default or failure shall continue for ten (10) days after written notice from Landlord designating such default or failure or (b) if within thirty (30) days after written notice from

Landlord to Tenant specifying any other default or defaults Tenant has not commenced diligently to correct the default or defaults so specified or has not thereafter diligently pursued such correction to completion, or (c) if any assignment shall be made by Tenant in violation of the provisions of Section 6.2.1 of this Lease, or (d) if any assignment shall be made by Tenant or any guarantor of Tenant for the benefit of creditors, or (e) if Tenant's leasehold interest shall be taken on execution, or (f) if a lien or other involuntary encumbrance is filed against Tenant's leasehold interest or Tenant's other property, including said leasehold interest, and is not discharged within ten (10) days thereafter, or (g) if a petition is filed by Tenant or any guarantor of Tenant for liquidation, or for reorganization or an arrangement under any provision of any bankruptcy law or code as then in force and effect, or (h) if an involuntary petition under any of the provisions of any bankruptcy law or code is filed against Tenant or any guarantor of Tenant and such involuntary petition is not dismissed within sixty (60) days thereafter, then, and in any of such cases, Landlord and the agents and servants of Landlord lawfully may, in addition to and not in derogation of any remedies for any preceding breach of covenant, immediately or at any time thereafter without demand or notice and with process of law enter into and upon the Premises or any part thereof in the name of the whole and repossess the same as of Landlord's former estate and expel Tenant and those claiming through or under Tenant and remove its and their effects without being deemed guilty of any manner of trespass and without prejudice to any remedies which might otherwise be used for arrears of rent or prior breach of covenants, and/or Landlord may terminate this Lease by sending written notice of termination to Tenant and this Lease shall terminate and come to an end on the date of entry as aforesaid, or on the third (3rd) day following the giving of such notice as fully and completely as if such date were the date originally fixed for expiration of the Term of this Lease. Tenant will quit and surrender the Premises to Landlord, but shall remain liable as herein provided. Tenant hereby waives all statutory rights to the Premises (including without limitation rights of redemption, if any, granted under any present or future laws, to the extent such rights may be lawfully waived). Landlord, without notice to Tenant, may store Tenant's effects, and those of any person claiming through or under Tenant, at the expense and risk of Tenant, and, if Landlord so elects, may sell such effects at public auction or private sale and apply the net proceeds to the payment of all sums due to Landlord from Tenant, if any, and pay over the balance, if any, to Tenant. No termination or repossession provided for in this Section 8.1 shall relieve Tenant or any guarantor of Tenant of the liabilities and obligations of Tenant under this Lease, all of which shall survive any such termination or repossession.

8.2 <u>Remedies.</u> (a) In the event of termination or repossession, Tenant covenants to pay punctually to Landlord Fixed Rent, Additional Rent and all other sums for which Tenant is obligated in this Lease to pay and in the same manner and to the same extent and at the same time as if this Lease had not been terminated. In calculating the amounts to be paid by Tenant pursuant to the preceding sentence, Tenant shall be credited with any amount paid to Landlord as compensation as set forth below in this Section 8.2 and also with the net proceeds of any rent obtained by Landlord by reletting the Premises, after deducting all Landlord's expense in connection with such reletting, including, without limitation, all repossession costs, brokerage commissions, reasonable attorney's fees, and expenses of preparing the Premises for such reletting.

- (b) Landlord may elect to (i) relet the Premises or any part or parts thereof, for a term or terms which may be equal to or less than or exceed the period which would otherwise have constituted the balance of the Term and may grant such concessions and free rent as Landlord in its sole and good faith judgment considers advisable or necessary to relet the same and/or (ii) make such alterations, repairs and decorations in the Premises as Landlord in its sole and good faith judgment considers advisable or necessary to relet the same and no action of Landlord in accordance with the foregoing or failure to relet or to collect rent under reletting shall operate or be construed to release or reduce Tenant's liability as aforesaid. However, Landlord shall use reasonable efforts to re-let the Premises after Tenant vacates the Premises once this Lease is terminated on account of a default by Tenant. For the purposes of this paragraph, marketing of the Premises in a manner similar to the way Landlord markets its other premises in the suburban market shall be deemed to satisfy Landlord's obligation to use such "reasonable efforts." In no event shall Landlord obtains full and complete possession of the Premises including, without limitation, the undisputed right to re-let the Premises free of any claim of Tenant,
- (ii) to lease the Premises to a tenant whose proposed use, in Landlord's bona fide judgment, would violate any restrictions by which Landlord is bound, (iii) to re-let the Premises before leasing other comparable vacant space in the Building, (iv) to lease the Premises for a rental less than the current fair market rental then prevailing for similar office space in the Building, or (v) to enter into a lease with any proposed tenant that does not have, in Landlord's reasonable opinion, sufficient financial resources or operating experience to operate the Premises in a manner comparable to other tenants in the Building. In no event, however, shall Tenant's liability hereunder be diminished or reduced if or to the extent such reasonable efforts of Landlord to re-let are not successful.
- (c) If this Lease is terminated under any of the provisions contained in Section 8.1, at the election of Landlord, and in lieu of full recovery by Landlord of the sums payable under the foregoing provisions of this Section 8.2 (except for the amount of any rent of any kind accrued and unpaid at the time of termination), Landlord may by written notice to Tenant, elect to recover, and Tenant shall thereupon pay forthwith to Landlord, as compensation, an amount equal to the present value of the amount by which the payments of Fixed Rent and Additional Rent payable for balance of the Term would exceed the fair rental value of the Premises for the balance of the Term.

(d) INTENTIONALLY OMITTED

- (e) Nothing contained in this Lease shall, however, limit or prejudice the right of Landlord to prove for and obtain in proceedings for bankruptcy or insolvency by reason of the termination of this Lease, an amount equal to the maximum allowed by any statute or rule of law in effect at the time when, and governing the proceedings in which, the damages are to be proved, whether or not the amount be greater than, equal to, or less than the amount of the loss or damages referred to above.
- 8.3 <u>Remedies Cumulative.</u> Any and all rights and remedies which Landlord may have under this Lease, and at law and equity, shall be cumulative and shall not be deemed inconsistent with each other, and any two or more of all such rights and remedies may be exercised at the same time insofar as permitted by law.
- Landlord's Right to Cure Defaults. Landlord may, but shall not be obligated to, cure, at any time, without any additional notice, any default by Tenant under this Lease beyond applicable notice and cure periods (except in the case of emergency, where no notice and cure period shall apply); and whenever Landlord so elects, all reasonable costs and expenses incurred by Landlord, including reasonable attorneys' fees, in curing a default shall be paid, as Additional Rent, by Tenant to Landlord on demand, together with lawful interest thereon from the date of payment by Landlord to the date of payment by Tenant.

- 8.5 <u>Effect of Waivers of Default.</u> Any consent or permission by Landlord to any act or omission which otherwise would be a breach of any covenant or condition herein, shall not in any way be held or construed (unless expressly so declared) to operate so as to impair the continuing obligation of any covenant or condition herein, or otherwise, except as to the specific instance, operate to permit similar acts or omissions.
- 8.6 <u>No Waiver, etc.</u> The failure of Landlord to seek redress for violation of, or to insist upon the strict performance of, any covenant or condition of this Lease shall not be deemed a waiver of such violation nor prevent a subsequent act, which would have originally constituted a violation, from having all the force and effect of an original violation. The receipt by Landlord of rent with knowledge of the breach of any covenant of this Lease shall not be deemed to have been a waiver of such breach by Landlord. No consent or waiver, express or implied, by Landlord to or of any breach of any agreement or duty shall be construed as a waiver or consent to or of any other breach of the same or any other agreement or duty.
- 8.7 <u>No Accord and Satisfaction.</u> No acceptance by Landlord of a lesser sum than the Fixed Rent, Additional Rent or any other charge then due shall be deemed to be other than on account of the earliest installment of such rent or charge due, nor shall any endorsement or statement on any check or any letter accompanying any check or payment as rent or other charge be deemed an accord and satisfaction, and Landlord may accept such check or payment without prejudice to Landlord's right to recover the balance of such installment or pursue any other remedy in this Lease provided.

ARTICLE 9 Rights of Mortgage Holders

9.1 <u>Rights of Mortgage Holders.</u> The word "mortgage" as used herein includes mortgages, deeds of trust or other similar instruments evidencing other voluntary liens or encumbrances, and modifications, consolidations, extensions, renewals, replacements and substitutes thereof. The word "holder" shall mean a mortgagee, and any subsequent holder or holders of a mortgage. Until the holder of a mortgage shall enter and take possession of the Property for the purpose of foreclosure, such holder shall have only such rights of Landlord as are necessary to preserve the integrity of this Lease as security. Upon entry and taking possession of the Property for the purpose of foreclosure, such holder shall have all the rights of Landlord. No such holder of a mortgage shall be liable either as mortgagee or as assignee, to perform, or be liable in damages for failure to perform, any of the obligations of Landlord unless and until such holder shall enter and take possession of the Property for the purpose of foreclosure. Upon entry for the purpose of foreclosure, such holder shall be liable to perform all of the obligations of Landlord, subject to and with the benefit of the provisions of Section 10.4, provided that a discontinuance of any foreclosure proceeding shall be deemed a conveyance under said provisions to the owner of the equity of the Property.

The covenants and agreements contained in this Lease with respect to the rights, powers and benefits of a holder of a mortgage (particularly, without limitation thereby, the covenants and agreements contained in this Section 9.1) constitute a continuing offer to any person, corporation or other entity, which by accepting a mortgage subject to this Lease, assumes the obligations herein set forth with respect to such holder; such holder is hereby constituted a party of this Lease as an obligee hereunder to the same extent as though its name were written hereon as such; and such holder shall be entitled to enforce such provisions in its own name. Tenant agrees on request of Landlord to execute and deliver from time to time any commercially reasonable agreement which may be necessary to implement the provisions of this Section 9.1.

9.2 <u>Lease Superior or Subordinate to Mortgages.</u> It is agreed that the rights and interest of Tenant under this Lease shall be (i) subject or subordinate to any present or future mortgage or mortgages and to any and all advances to be made thereunder, and to the interest of the holder thereof in the Premises or any property of which the Premises are a

part if Landlord shall elect by notice to Tenant to subject or subordinate the rights and interest of Tenant under this Lease to such mortgage or (ii) prior to any present or future mortgage or mortgages, if Landlord shall elect, by notice to Tenant, to give the rights and interest of Tenant under this Lease priority to such mortgage; in the event of either of such elections and upon notification by Landlord to that effect, the rights and interest of Tenant under this Lease should be deemed to be subordinate to, or have priority over, as the case may be, said mortgage or mortgages, irrespective of the time of execution or time of recording of any such mortgage or mortgages (provided that, in the case of subordination of this Lease to any future mortgages, the holder thereof agrees not to disturb the possession of Tenant so long as Tenant is not in default hereunder). Tenant agrees it will, upon not less than fifteen (15) days' prior written request by Landlord, execute, acknowledge and deliver any and all commercially reasonable instruments deemed by Landlord necessary or desirable to give effect to or notice of such subordination or priority. Tenant also agrees that if it shall fail at any time to execute, acknowledge and deliver any such instrument requested by Landlord, Landlord may, in addition to any other remedies available to it, execute, acknowledge and deliver such instrument as the attorney-in-fact of Tenant and in Tenant's name; and Tenant does hereby make, constitute and irrevocably appoint Landlord as its attorney-in-fact, coupled with an interest with full power of substitution, and in its name, place and stead so to do. Any Mortgage to which this Lease shall be subordinated may contain such terms, provisions and conditions as the holder deems usual or customary. Landlord shall use reasonable efforts to obtain its current lender's (and, when applicable, future lender's) standard form of subordination, non-disturbance and attornment agreement ("SNDA"). In the event Tenant wishes to negotiate such SNDA, it shall do so directly with the lender and all costs of obtaining and negotiating the SNDA shall be paid by Tenant.

ARTICLE 10 Miscellaneous Provisions

- 10.1 <u>Notices from One Party to the Other.</u> All notices required or permitted hereunder shall be in writing and addressed, if to the Tenant, at the Original Notice Address of Tenant or such other address as Tenant shall have last designated by notice in writing to Landlord (with a copy of any notice of default being simultaneously sent to Guarantor pursuant to the terms of the Guaranty) and, if to Landlord, at the Original Notice Address of Landlord or such other address as Landlord shall have last designated by notice in writing to Tenant. Any notice shall be deemed duly given upon receipt or rejection when mailed to such address postage prepaid, by registered or certified mail, return receipt requested, or on the next business day after being deposited with a nationally recognized overnight courier service for delivery to such address, or when delivered to such address by hand.
- 10.2 <u>Quiet Enjoyment.</u> Landlord agrees that upon Tenant's paying the rent and performing and observing the agreements, conditions and other provisions on its part to be performed and observed, Tenant shall and may peaceably and quietly have, hold and enjoy the Premises during the term hereof without any manner of hindrance or molestation from Landlord or anyone claiming under Landlord, subject, however, to the terms of this Lease.
- 10.3 <u>Lease not to be Recorded.</u> Tenant agrees that it will not record this Lease. Both parties shall, upon the request of either, execute and deliver a notice or short form of this Lease in such form, if any, as may be permitted by applicable statute. Tenant hereby irrevocably appoints Landlord as Tenant's attorney-in-fact (which appointment shall survive termination of the term of this Lease) with full power of substitution to execute, acknowledge and deliver a notice of termination of lease in Tenant's name if Tenant fails, within ten (10) days after request therefor, to either execute, acknowledge or deliver such notice of termination or give Landlord written notice setting forth the reasons why Tenant is refusing to deliver such notice of termination.
- Limitation of Landlord's Liability. The term "Landlord" as used in this Lease, so far as covenants or obligations to be 10.4 performed by Landlord are concerned, shall be limited to mean and include only the owner or owners at the time in question of the Property, and in the event of any transfer or transfers of title to said property, the Landlord (and in case of any subsequent transfers or conveyances, the then grantor) shall be concurrently freed and relieved from and after the date of such transfer or conveyance, without any further instrument or agreement of all liability as respects the performance of any covenants or obligations on the part of the Landlord contained in this Lease thereafter to be performed, it being intended hereby that the covenants and obligations contained in this Lease on the part of Landlord, shall, subject as aforesaid, be binding on the Landlord, its successors and assigns, only during and in respect of their respective successive periods of ownership of said leasehold interest or fee, as the case may be. Tenant, its successors and assigns, shall not assert nor seek to enforce any claim for breach of this Lease against any of Landlord's assets other than Landlord's interest in the Property and in the rents, issues and profits thereof, and Tenant agrees to look solely to such interest for the satisfaction of any liability or claim against Landlord under this Lease, it being specifically agreed that in no event whatsoever shall Landlord (which term shall include, without limitation, any general or limited partner, trustees, beneficiaries, officers, directors, managers, members or stockholders of Landlord) ever be personally liable for any such liability.
- 10.5 <u>Force Majeure.</u> In any case where either party hereto is required to do any act, delays caused by or resulting from Acts of God, war, civil commotion, fire, flood or other casualty, labor difficulties, shortages of labor, materials or equipment, government regulations, unusually severe weather, quarantine, infectious disease, epidemic, pandemic, public health emergency or crisis or other causes beyond such party's reasonable control (any of the foregoing causes being referred to herein as "Force Majeure") shall not be counted in determining the time during which work shall be

completed and the period of time for the performance of such action shall be extended by the number of days that the performance is actually delayed by such Force Majeure delay, whether such time be designated by a fixed date, a fixed time or a "reasonable time," and such time shall be deemed to be extended by the period of such delay. During the continuance of any event of Force Majeure, Landlord may temporarily close the Building and/or cease to provide any service or amenity if Landlord deems it necessary, in its sole and good faith judgment, for the safety or well-being of the Building, the Property, its tenants and their employees, invitees and contractors. Tenant's inability to pay any sums due Landlord hereunder shall in no way be affected or excused by any of the foregoing causes and shall in no event be deemed a Force Majeure event.

- Landlord's Default. Landlord shall not be deemed to be in default in the performance of any of its obligations hereunder unless it shall fail to perform such obligations and such failure shall continue for a period of thirty (30) days or such additional time as is reasonably required to correct any such default after written notice has been given by Tenant to Landlord (and to all mortgagees of which Tenant has notice) specifying the nature of Landlord's alleged default. Subject to the terms and conditions set forth in this Lease, if Landlord shall default in the performance or observance of any agreement or obligation under this Lease, and such failure shall continue beyond expiration of the notice and cure period set forth above, then Tenant shall have all rights and remedies available to Tenant at law or in equity. Notwithstanding any provision contained herein, in no event shall Landlord ever be liable to Tenant, or any person claiming by, through or under Tenant, for any special, indirect, incidental or consequential damages, or for any lost profits. Tenant shall have no right to terminate this Lease for any default by Landlord hereunder and no right, for any such default, to offset or counterclaim against any rent due hereunder. Nothing contained herein shall be construed as waiving any right under law that Tenant may have to bring a legal action for constructive eviction.
- 10.7 <u>Brokerage</u>. Tenant warrants and represents that it has dealt with no broker in connection with the consummation of this Lease, other than [***](the "Brokers"), and in the event of any brokerage claims, other than by the Brokers against Landlord predicated upon prior dealings with Tenant, Tenant agrees to defend the same and indemnify and hold Landlord harmless against any such claim.
- 10.8 <u>Applicable Law and Construction; Merger; Jury Trial.</u> This Lease shall be governed by and construed in accordance with the laws of the Commonwealth of Massachusetts and, if any provisions of this Lease shall to any extent be invalid, the remainder of this Lease shall not be affected thereby. This Lease and the Exhibits attached hereto and forming a part hereof constitute all the covenants, promises, agreements, and understandings between Landlord and Tenant concerning the Premises and the Building and there are no covenants, promises, agreements or understandings, either oral or written, between them other than as are set forth in this Lease. Neither Landlord nor Landlord's agents shall be bound to any representations with respect to the Premises, the Building or the Property except as herein expressly set forth, and all representations, either oral or written, shall be deemed to be merged into this Lease. The headings and titles of the Articles and Sections contained herein are for convenience are referenced only and shall in no way be held to define, limit, amplify or describe or aid in the interpretation or construction of any provisions of this Lease. Tenant shall and does hereby waive trial by jury in any action, proceeding, or claim brought by or against Landlord regarding any matter arising out of or in any way connected with this Lease, the relationship of Landlord and Tenant or Tenant's use or occupancy of the Premises. Unless repugnant to the context, the words "Landlord" and "Tenant" appearing in this Lease shall be construed to mean those named above and their respective heirs, executors, administrators, successors and assigns, and those claiming through or under them respectively. If there be more than one tenant, the obligations imposed by this Lease upon Tenant shall be joint and several.
- 10.9 <u>Consents.</u> With respect to any provision of this Lease which either provides or is held to provide that Landlord shall not unreasonably withhold or unreasonably delay any consent or approval, Tenant shall not be entitled to make any claim for, and Tenant hereby

expressly waives, any claim for damages, it being understood and agreed that Tenant's sole remedy therefor shall be an action for specific performance.

- 10.10 <u>Authority.</u> In the event either party is a corporation, partnership or limited liability company, such party hereby represents and warrants that: it is a duly constituted corporation, partnership or limited liability company, as the case may be, qualified to do business in the Commonwealth of Massachusetts; that the person executing this Lease is duly authorized to execute and deliver this Lease on behalf of said corporation(s), partnership(s) or limited liability company(ies); and that the by-laws or operating agreement or partnership agreement of such party authorizes it to enter into this Lease.
- 10.11 Execution; Counterparts. This Lease shall not be valid and binding until executed and delivered by Landlord and may be executed in multiple counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument. Any facsimile or other electronic transmittal of original signature versions of this Lease shall be considered to have the same legal effect as execution and delivery of the original document and shall be treated in all manner and respects as the original document. Landlord and Tenant agree both parties shall execute this Lease, all amendments and other associated lease documents, by means of DocuSign (or another mutually acceptable electronic signature method), and that execution of this Lease, amendments and other lease documents by means of DocuSign (or another mutually acceptable electronic signature method) shall be valid and binding, and have the same legal effect as execution with original ink signatures.
- OFAC Compliance. Tenant hereby warrants and represents that: (a) neither Tenant nor any of its affiliates does business 10.12 with, sponsors, or provides assistance or support to, the government of, or any person located in, any country, or with any other person, targeted by any of the economic sanctions of the United States administered by The Office of Foreign Assets Control ("OFAC"); Tenant is not owned or controlled (within the meaning of the regulations promulgating such sanctions or the laws authorizing such promulgation) by any such government or person; and any payments and/or proceeds received by Tenant under the terms of this Lease will not be used to fund any operations in, finance any investments or activities in or make any payments to, any country, or to make any payments to any person, targeted by any of such sanctions; (b) no funds tendered to Landlord by Tenant under the terms of this Lease are or will be directly or indirectly derived from activities that may contravene U.S. federal, state or international laws and regulations, including anti-money laundering laws; (c) neither Tenant, nor any person controlling, controlled by, or under common control with, Tenant, nor any person having a beneficial interest in Tenant, nor any person for whom Tenant is acting as agent or nominee, nor any person providing funds to Tenant in connection with this Lease (i) is under investigation by any governmental authority for, or has been charged with, or convicted of, money laundering, drug trafficking, terrorist related activities, any crimes which in the United States would be predicate crimes to money laundering, or any violation of any Anti-Money Laundering Laws; (ii) has been assessed civil or criminal penalties under any Anti-Money Laundering Laws; (iii) has had any of its funds seized or forfeited in any action under any Anti-Money Laundering Laws; (iv) is a person or entity that resides or has a place of business in a country or territory which is designated as a Non-Cooperative Country or Territory by the Financial Action Task Force on Money Laundering, or whose subscription funds are transferred from or through such a jurisdiction; (v) is a "Foreign Shell Bank" within the meaning of the Patriot Act (i.e., a foreign bank that does not have a physical presence in any country and that is not affiliated with a bank that has a physical presence and an acceptable level of regulation and supervision); (vi) is a person or entity that resides in, or is organized under the laws of, a jurisdiction designated by the Secretary of the Treasury under Section 311 or 312 of the Patriot Act as warranting special measures due to money laundering concerns; (vii) is an entity that is designated by the Secretary of the Treasury as warranting such special measures due to money laundering concerns; or (viii) is a person or entity that otherwise appears on any US.-government provided list of known or suspected terrorists or terrorist

organizations. For purposes of this representation, the term "Anti-Money Laundering Laws" shall mean all laws, regulations and executive orders, state and federal, criminal and civil, that (1) limit the use of and/or seek the forfeiture of proceeds from illegal transactions; (2) limit commercial transactions with designated countries or individuals believed to be terrorists, narcotics dealers or otherwise engaged in activities contrary to the interests of the United States; (3) require identification and documentation of the parties with whom a financial institution conducts business; or

- (4) are designed to disrupt the flow of funds to terrorist organizations. Such laws, regulations, and sanctions shall include, without limitation, the USA PATRIOT Act of 2001, Pub. L. No. 107-56 (the "Patriot Act"), Executive Order 13224, the Bank Secrecy Act, 31 U.S.C. Section 531 et. seq., the Trading with the Enemy Act, 50 U.S.C. App. Section 1 et. seq., the International Emergency Economic Powers Act, 50 U.S.C. Section 1701 et. seq., the OFAC-administered economic sanctions, and laws relating to prevention and detection of money laundering in 18 U.S.C. Sections 1956 and 1957.
- 10.13 <u>Confidentiality; No Public Statement.</u> Tenant and Landlord agree that the execution of this Lease and the information contained herein shall be kept confidential and shall not be divulged to any third party without the consent of the other party; provided, without such consent, either party may provide such information to its accountants, attorneys, lenders, potential lenders, partners, potential partners, potential purchasers, employees, agents, consultants and any prospective assignee or subtenant, as may be reasonably necessary to perform their duties or in connection with the terms of any re-financing, sale, or assignment or subletting. Neither Landlord nor Tenant shall issue any public statement, announcement or press release regarding the Premises, this Lease or the terms of this Lease, without the approval of the other party. Landlord shall be entitled to withhold its approval of any statement, announcement or press release contemplated by Tenant that would coincide with or precede any public statement, announcement or press release by Landlord regarding this Lease. Notwithstanding the foregoing restrictions, Tenant shall be permitted to disclose this Lease or portions there to the extent required in order for shares of its stock (or equivalents thereof) to become or remain listed in a securities exchange or to the extent required by applicable law or governmental authority.

Signatures appear on the immediately following page

WITNESS the execution hereof under seal as of the Effective Date:

LANDLORD:

201 BURLINGTON ROAD OWNER, LLC

a Delaware limited liability company

By: 201 Burlington Road Venture, LLC, a Delaware limited liability company, its sole member

By: 201 Burlington Road NDC Promote LLC, a Massachusetts limited liability company, its Manager

By: Nordblom JV Manager, Inc., a Massachusetts corporation, its manager

By: /s/ <u>Crosby Nordblom</u> Name: Crosby Nordblom Title : VP

By: <u>/s/ Adele Olivier</u> Name: Adele Olivier Title: Senior Vice President

TENANT:

LANTHEUS MEDICAL IMAGING, INC.

By: <u>/s/ Mary Anne Heino</u> Name: Mary Anne Heino Title: Chief Executive Officer and President

EXHIBIT A

LEGAL DESCRIPTION OF THE PROPERTY

Three contiguous parcels of registered land in Bedford, Massachusetts, shown as Lots 37, 21A and 38 on Land Court Plan 29283G, respectively bounded and described as follows:

LOT 37

NORTHEASTERLY

EASTERLY

NORTHEASTERLY

SOUTHEASTERLY

NORTHEASTERLY

EASTERLY and SOUTHEASTERLY

SOUTHWESTERLY

WESTERLY

NORTHERLY

LOT 21A SOUTHERLY

SOUTHWESTERLY

NORTHWESTERLY

NORTHERLY

EASTERLY

LOT 38

SOUTHWESTERLY

NORTHWESTERLY and WESTERLY

NORTHERLY

EASTERLY

By Burlington Road, 282.27 feet;

410.02 feet;

109.81 feet, by Lot 28 as shown on said plan;

By Lot 21A on said plan, about 35 feet;

By the former location of **Vie** Brook;

By Lot 38 on said plan, about 671 feet;

383.95 feet

852.37 feet; and

199.44 feet, by the Northeasterly, Easterly and Southerly

lines of the State Highway (Route 3), non-access.

By Lot 15 as shown on plan hereinafter mentioned 213.97

By the former location of Vine Brook;

By Lot 37 on said plan, about 35 feet;

By Lot 28 on said plan, 344.35 feet; and

By Lot 22 on said plan, 20 feet.

By the Northeasterly line of the State Highway (Route 3),

non-access, 195 feet;

By Lot 37 as shown on said plan hereinafter mentioned, about 671 feet;

By the former location of Vine Brook; and

By land now or formerly of Warrenton Corporation, 762

feet.

All said boundaries are determined by the Court to be located as shown on a subdivision plan, as approved by the Court, filed in the Land Registration Office, a copy of which is filed in the Registry of Deeds for the South Registry District of Middlesex County in Registration Book 742, Page 69, with Certificate 122419.

EXHIBIT B PLAN SHOWING THE PREMISES

<u>[***]</u>

EXHIBIT C COMMENCEMENT DATE

NOTIFICATION

To:	
("Landlord") and	("Tenant") are parties to a lease
Commencement Date is; the First Rent A Rent Commencement Date is; the Expira	Fies Tenant that the Commencement Date is; the Initial Rent Adjustment Date is; the Second Rent Adjustment Date isand the Full ation Date is; and the first Lease Year commenced on eed for this notification to be effective, we would appreciate you confirming the
	(Landlord)
	By:
Confirmed:	
_	
(Tenant)	
By:	

EXHIBIT D PRELIMINARY FIT PLAN

<u>[***]</u>

EXHIBIT D-1 LANDLORD'S WORK LETTER

Landlord shall furnish and install the following materials and work in preparation for Tenant's occupancy, in accordance with the Final Plans created by [***] and using building standard materials and standards.

Demolition: Landlord will demo select walls and demise the Premises per the Plans.

Partitions: New partitions within premises to be 3 5/8" metal studs with one layer of gypsum board on each side. New

partitions will extend from floor to the acoustic tile ceiling, except for walls between offices which will

extend 6" above the ceiling grid.

Doors and Frames: Where possible, sliders/doors, frames and hardware removed during demolition will be repurposed

per the Plans. If needed, any new doors will be 3'0" x 7'0" x 1¾" solid-core. Door frames will be

pressed metal. Doors will be delivered in good functional condition.

Tenant may substitute wood framed glass door or full glass door for any existing door at an additional

charge.

Ceiling: A 2x2 acoustical ceiling tile and grid to be installed per the Plans.

Existing ceiling tile shall be delivered in good condition that matches the new tile and be reasonably

free of blemishes.

Open ceiling to remain in portions of the Break Room per the Plans.

Painting: All new wall surfaces shall receive two coats of latex paint with an eggshell finish using not more than three (3)

readily available colors. All doorframes and solid doors will match existing. All existing wall surfaces

will be touched up as needed.

Flooring: New carpet tiles to be installed in office area per the Plans. Landlord to provide Tenant an option of two (2)

different carpet tiles to select from.

Bathrooms: Bathrooms will be delivered in as-is condition with all fixtures and finishes to be in good order and

repair.

Break Room: Break Rooms on the 2nd and 3rd floor will be delivered in as-is condition with all fixtures and finishes to be

in good order and repair.

HVAC: HVAC distribution will be reconfigured as needed per the Plans and will remain sufficient for standard office use.

Any specialized HVAC requirements for Tenant's equipment will be at Tenant's sole expense.

Electrical: Lights

Existing fixtures to remain, and any new fixtures will match existing. Any specialty fixtures shall be at

the cost of the tenant.

Switches and Outlets

Install new switches and outlets as needed due to new partitions work, and provided in a similar standard as existing office conditions. The minimum requirement for offices is two duplex outlets and one ring/string for tel/data. Standard electrical distribution will also be provided for all conference

rooms, team rooms and huddle rooms.

Exit Signs Per code.

Fire Alarm

Adequate devices for typical office layout.

Window Blinds: Existing to remain, provided that any blinds that are not in good operating condition will be replaced by

Landlord.

Notes: Furniture, specialized HVAC or other equipment and any related infrastructure or structural work, power and

blocking needs related to Tenant AV installations, telephone and data wiring, signage (interior and exterior), appliances, and access control systems, and coffee stations are not included in Landlord's

Work.

Any existing millwork will be made available for reuse given the opportunity.

[***]

EXHIBIT E WORK CHANGE ORDER

Lease Date: Landlord: Tenant:	Date:
	Work Change Order No.:
Building Address: Premises:	
<u> </u>	
Tenant directs Landlord to make the following additions to Land	lord's work: Description of additional
work:	
Work Change Order Amount:	
Period of delay due to Work Change Order:	
<u> </u>	
Amount of Previous Work Change Orders:	
This Work Change Order:	
Total Amount of Work Change Orders:	
Landlord approves this Work Change Order and Tenant agrees to above in full upon Tenant's execution of this Work Change Order	
Tenant: Landlord:	
By: By:	
Title: Title:	

EXHIBIT F

RULES AND REGULATIONS

- 1. The sidewalks, entrances, passages, corridors, vestibules, halls, elevators, or stairways in or about the Building shall not be obstructed by Tenant.
- 2. Tenant shall not place objects against glass partitions, doors or windows which would be unsightly from the Building corridor or from the exterior of the Building.
- 3. Tenant shall not waste electricity or water in the Building premises and shall cooperate fully with Landlord to assure the most effective operation of the Building heating and air conditioning systems. All regulating and adjusting of heating and air-conditioning apparatus shall be done by the Landlord's agents or employees.
- 4. Tenant shall not use the Premises so as to cause any increase above normal insurance premiums on the Building.
- 5. No bicycles, vehicles, or animals of any kind shall be brought into or kept in or about the Premises. No space in the Building shall be used for manufacturing or for the sale of merchandise of any kind at auction or for storage thereof preliminary to such sale.
- 6. Tenant shall cooperate with Landlord in minimizing loss and risk thereof from fire and associated perils.
- 7. The water and wash closets and other plumbing fixtures shall not be used for any purposes other than those for which they were designed and constructed and no sweepings, rubbish, rags, acid or like substance shall be deposited therein. All damages resulting from any misuse of the fixtures shall be borne by the Tenant.
- 8. Landlord reserves the right to establish, modify, and enforce reasonable parking rules and regulations, provided such rules and obligations do not diminish Tenant's rights under the Lease.
- 9. Landlord reserves the right at any time to rescind, alter or waive any rule or regulation at any time prescribed for the Building and to impose additional reasonable rules and regulations when in its judgment deems it necessary, desirable or proper for its best interest and for the best interest of the tenants and no alteration or waiver of any rule or regulation in favor of one tenant shall operate as an alteration or waiver in favor of any other tenant, provided such rules and regulations do not diminish Tenant's rights under the Lease. Landlord shall not be responsible to any tenant for the nonobservance or violation by any other tenant however resulting of any rules or regulations at any time prescribed for the Building.
- 10. Tenant acknowledges that the Building has been designated a non-smoking building. At no time shall Tenant permit its agents, employees, contractors, guests or invitees to smoke in the Building or, except in specified locations, directly outside the Building.
- 11. The normal business hours for the Building are 7:00 A.M. to 6:00 P.M. on Mondays through Fridays, and 7:00 A.M. to 1:00 P.M. on Saturdays, excluding all legal holidays on which the Building is closed.
- 12. Landlord may from time to time adopt systems and procedures (including, but not limited to, temporary closures and/or suspension of services during a Force Majeure event) for the security and safety of the Building and Property, their occupants, entry, use and contents. Tenant, its agents, employees, contractors, guests and invitees shall comply with Landlord's systems and procedures.

EXHIBIT F-1 CONSTRUCTION RULES AND

REGULATIONS

- 1. All work shall be performed in accordance with all applicable state laws and in accordance with all requirements and codes of the Town of Bedford and guidelines of the Landlord's managing agent ("Building Management"). The Building is operational and extra care and precautions must be taken to avoid interruption of services.
- 2. Certificates of Insurance from Tenant's general contractor ("Contractor") and its subcontractors must be submitted seven (7) days prior to the commencement of any work.
- 3. At least three (3) weeks before construction, Contractor must schedule a pre-construction meeting with the Landlord's project management team. Meeting materials should include detailed schedules; addresses and telephone numbers of supervisors, contractors and subcontractors: copies of permits; proof of current insurance (including all subcontractors); and notice of any contractor's involvement in a labor dispute.
- 4. All Contractor personnel will be required to sign in and sign out at the security desk upon entering or exiting the premises. A Contractor ID badge may be issued during the work to facilitate this process.
- 5. Contractor personnel are only permitted within the Building during normal trade working hours plus 30 minutes time before and after normal trade hours for set up and pick up of tools, etc.
- 6. Testing of sprinkler system, fire protection system, demolition, coring, and any other similar type work must be coordinated through Building Management with five (5) business days prior notice. Normal business hours are 8:00am to 6:00pm Monday through Fridays and; 8:00am to 1:00pm Saturdays. Work on any system within the Building shall be coordinated with Building Management.
- 7. Contractor must use the freight elevator for deliveries of equipment and materials and for all construction personnel. Upon prior approval by Building Management under special circumstances (such as, for example, clean-item delivery, after hours delivery and if the freight elevator if out-of-service), Contractor shall be permitted to use one (1) passenger car for construction personnel only (no equipment or material). Elevators will be inspected by the Contractor and Building Management prior to use by Contractor. The Contractor is responsible for returning the elevators to the Landlord in the same good condition as received. Any cost for repair or servicing during the use by construction personnel will be at the cost of the Contractor.
- 8. If use of the passenger elevator is approved by Building Management, deliveries through the main lobby must be done in a professional manner. Floor protection is required and elevator wall mats must be used. No storage of any items allowed in main lobby or any common area. The Contractor is responsible for leaving the main lobby, elevator, and any other common area in "broom clean" condition. The Contractor will incur costs for the clean-up for areas left dirty. Rubbish cannot be stored in any common area.
- 9. The loading dock and freight elevator must be reserved in advance and are not exclusive to Contractor's sole use.
- 10. Public area corridors and carpet are to be protected by plastic runners or a series of walk- off mats of suitable appearance as required by Building Management.
- 11. Contractor shall provide heavy plastic screening for dust protection and/or temporary walls of suitable appearance as required by Building Management.

- 12. Walk-off mats are to be provided at entrance doors.
- 13. No utilities (electricity, water, and plumbing) or services to the tenants are to be cutoff or interrupted without first having requested, in writing, and secured, in writing, the permission of Building Management.
- 14. If taking the Building fire alarm points offline is required in connection with any work, the cost of such service will be billable to the Contractor.
- 15. Should any scope of work require entry to another tenant's space, Tenant and/or Contractor must notify Landlord. If approved by Landlord, the activity shall be under the supervision of Landlord's representative. The cost of said supervision will be billable to the Contractor.
- 16. Admittance to the roof of the Building is allowed only upon the prior written consent of Landlord.
- 17. There is a "No Smoking Policy" in effect for all areas of the Building.
- 18. Contractor will be responsible for daily removal of waste foods, milk and soft drink containers, etc. Building trash receptacles are not to be used.
- 19. Construction personnel are not to eat in the lobby or in front of the Building, nor are they to congregate in the lobby or in front of the Building.
- 20. Construction personnel are not to utilize any vacant space within the Building other than that space which is designated by the Landlord to the Contractor.
- 21. There will be no radios on the job site.
- 22. All workers are required to wear a shirt, shoes, and full length trousers.
- 23. Protection of hallway carpets, wall coverings, granite and marble, and elevator from damage with masonite board, carpet or pads is required.
- 24. Public spaces, corridors, elevator, bathrooms, lobby, etc., must be cleaned immediately after use. Restrooms for contractor use will be designated to the Contractor. No other restrooms are to be utilized by Contractor personnel.
- 25. There will be no smoking, eating, or open food containers in the elevator, carpeted areas, or public lobbies. There will be no yelling or boisterous activities; nor is alcohol or controlled substances allowed or tolerated. Individuals under the influence or in possession of such will be prosecuted.
- 26. Contractor shall post no signs without Building Manager's expressed prior approval which may be withheld for any reason.
- 27. Contractor shall supply Building Management with a copy of all permits prior to start of any work.
- 28. Contractor shall complete work without disruption from labor disputes and in harmony with other trades.
- 29. The Construction Rules and Regulations are subject to change in Landlord's sole discretion.

Landlord reserves the right to prohibit access to the Building by any contractors who fail to comply with these Construction Rules and Regulations.

EXHIBIT G

TENANT ESTOPPEL CERTIFICATE

- certain parcel of real property known as and numbered __ (the "Property"), a portion of which is presently occupied by the Tenant (the "Leased Premises") under a [NAME OF LEASE] dated __ by and between Tenant and __, a __ (the "Landlord") [, as affected by a certain [NAME OF AMENDMENT(S)] dated ____]([as so amended and]as may be [further] amended, restated or modified from time to time, the "Lease"). Incident to Lender's proposed loan and mortgage, and at Lender's request, Tenant hereby certifies as follows:
- 1. A true, correct and complete copy of the Lease (including, without limitation, all exhibits and schedules to the same and all amendments, restatements and modifications to the Lease) is attached hereto as Exhibit A.
- 2. Tenant is the holder of the lessee's interest under the Lease.
- 3. The Lease is in full force and effect. Tenant has accepted the Leased Premises without exception and all requirements for the commencement and validity of the Lease (including, without limitation, any and all tenant improvements and other construction work required under the terms of the Lease) have been satisfied. Tenant is presently (and has, since ____, been) in sole possession of the Leased Premises with ongoing full operation of its business at the Leased Premises, the conduct of such business falling within the uses permitted under the terms of the Lease. Tenant is presently paying the full amount of rent required to be paid under the terms of the Lease.
- 4. Tenant has not subleased all or any part of the Leased Premises or assigned, hypothecated or pledged all or any part of the Lease, or otherwise transferred its interest in the Lease or the Leased Premises other than to the Lender. The Lease has not been modified, changed, altered, amended or supplemented in any respect (except as indicated above) and is the only Lease or agreement between Tenant and Landlord affecting the Leased Premises.
- 5. The term of the Lease is currently scheduled to expire on ____, 20_. If there are any rights of extension or renewal remaining under the terms of the Lease, the same have not been exercised as of the date hereof.
- 6. Tenant has paid to Landlord a security deposit in the amount of \$___.

- 7. Tenant is current in payment of all fixed rent and other charges (including, without limitation, payment of all ____ which are to be paid by the Tenant pursuant to the terms of the Lease) which are due to be paid under the Lease, with minimum rent paid in full for the period ending _____. The monthly installment of fixed rent is \$___. No rent or other sum payable under the Lease is being paid in arrears. No rent or other sum payable under the Lease has been paid in advance of the due date thereof, and Tenant hereby agrees with Lender that it shall not pay any fixed rent or any other sum due or to be paid under the Lease more than thirty (30) days in advance of the due date thereof. Tenant has made no agreement with the Landlord (or its agents or employees) concerning free rent, partial rent, rent abatement, rebate or rental payments, or any other type of rental concession except as may be set forth in the Lease. Tenant has no defenses, set offs, or counterclaims to the payment of rent and all other amounts due from Tenant to the Landlord under the Lease.
- 8. All of the obligations on the part of Landlord under the Lease, including any obligation for the performance of any construction work or installation of any equipment, have been carried out, and Tenant has no claim or knowledge of any claim against the holder of Landlord's interest on account of any default or failure of performance by Landlord under the Lease. Tenant is not in default of any of its obligations to be paid or performed under the Lease. As of the date hereof, Tenant is entitled to no offset or deduction in rent and has no claim or defense to the performance of any obligation to be performed by it under the Lease.
- 9. Tenant has received or will receive payment or credit for tenant improvement work in accordance with the terms of the Lease.
- 10. Tenant is presently maintaining the liability and casualty insurance, if any, of the types and in the amounts required under the terms of the Lease. Tenant agrees that it shall designate the Lender as the loss payee and additional insured in all such policies now or hereafter obtained pursuant to the Lease.
- 11. The Lease does not contain and Tenant does not have and has not exercised any rights or options to terminate or cancel the Lease or to reduce or extend the terms thereof
- or to enlarge, reduce or otherwise change the Leased Premises. The Lease does not contain and Tenant has no outstanding options or rights of first refusal to purchase the Leased Premises or any part of the Property.
- 12. No actions or proceedings, whether voluntary or otherwise, are pending against Tenant under the bankruptcy or other insolvency laws of the United States or of any state thereof.
- 13. Tenant is not using the Leased Premises or any part of the Property in a manner which is not in compliance with all laws, or which would result in the presence of, use, generation, manufacture, storage, treatment, disposal, discharge or release on or from the Property of any substances or compounds prohibited or regulated under any Federal, state or municipal laws pertaining to health or the environment in violation of applicable laws. There are no regulatory actions or other claims pending or threatened against Tenant or against the Leased Premises arising out of the presence of such substances on the Leased Premises.
- 14. This certification is made to induce Lender to make certain fundings, knowing that Lender shall rely upon the truth of this certificate in disbursing said funds, and shall inure to the benefit of Lender, its successors and assigns, and shall be binding upon Tenant and Tenant's heirs, legal representatives, successors and assigns. This certification shall not be deemed to alter or modify any of the terms and conditions of the Lease except to the extent specifically set forth herein. The person signing this Tenant Estoppel Certificate on behalf of the Tenant is a duly authorized agent of the Tenant.

DO NOT EXECUTE THIS CERTIFICATE UNLESS A TRUE, CORRECT AND COMPLETE COPY OF YOUR LEASE AND ALL AMENDMENTS THERETO ARE ATTACHED. THE LENDER IS RELYING ON YOUR STATEMENTS AND AGREEMENTS CONTAINED IN THIS CERTIFICATE IN TAKING ACTION RESPECTING THE LANDLORD'S LOAN REQUEST.

Dated this day of , 201 .
TENANT:
by:
Name: Title:
Hereunto Duly Authorized
COMMONWEALTH OF MASSACHUSETTS)
) ss COUNTY OF)
On this day of, 201_, before me, the undersigned notary public, personally appeared, proved to me through satisfactory evidence of identification, which was, to be the person whose name is signed on the preceding or attached document, and acknowledged to me that he signed it as of, as voluntarily for its stated purpose.
NOTARY PUBLIC:
My Commission expires

EXHIBIT H

FORM OF LANDLORD'S WAIVER AND CONSENT

WHEREAS, (the "Tenant") has or is about to enter
into certain financing agreements with (the
"Bank") pursuant to which the Bank has been or may be granted a security interest in certain property of the Tenant;
and

WHEREAS, Tenant is the tenant, pursuant to a lease agreement by and between Tenant and the undersigned (the "Landlord") dated as of __ (the "Lease"), of certain demised premises contained in the building located at the following address:

and more particularly described in the Lease (the "Premises");

NOW, THEREFORE, for valuable consideration, the Landlord agrees, for as long as Tenant remains indebted to the Bank, as follows:

- 1. Landlord acknowledges and agrees that the personal property of Tenant (which for purposes hereof shall not include computer wiring, telephone wiring and systems, and demountable partitions) in which the Bank has been granted a security interest (the "Bank Collateral") may from time to time be located on the Premises;
- 2. Landlord subordinates, waives, releases and relinquishes unto the Bank, its successors or assigns, all right, title and interest, if any, which the Landlord may otherwise claim in and to the Bank Collateral, except as provided in subparagraph (d) hereinbelow;
- 3. Upon providing the Landlord with at least five (5) business days' prior written notice that Tenant is in default of its obligations to the Bank, the Bank shall then have the right to enter the Premises during business hours for the purpose of removing said Bank Collateral, provided (i) the Bank completes the removal of said Bank Collateral within ten (10) business days following said first written notice of default, and (ii) the Bank restores any part of the Premises which may be damaged by such removal to its condition prior to such removal in an expeditious manner not to exceed ten (10) business days following said first written notice of default;
- 4. Upon receipt of written notice from Landlord of the expiration or earlier termination of the Lease, the Bank shall have ten (10) business days to enter the Premises during business hours, remove said Bank Collateral, and restore any part of the Premises which may be damaged by such removal to its condition prior to such removal. If the Bank fails to so remove the Bank Collateral, the Bank agrees that the Bank Collateral shall thereupon be deemed subject to the yield up provisions of the Lease, so the Landlord may treat the Bank Collateral as abandoned, deem it Landlord's property, if Landlord so elects, and retain or remove and dispose of it, all as provided in the Lease;
- 5. All notices and other communications under this Landlord's Consent and Waiver shall be in writing, and shall be delivered by hand, by a nationally recognized commercial next day delivery service, or by certified or registered mail, return receipt requested, and sent to the following addresses:

if to the Bank:

Attention:

with a copy to:

if to the Landlord: c/o Nordblom Management Company, Inc.

71 Third Avenue

Burlington, MA 01803

Such notices shall be effective (a) in the case of hand deliveries, when received, (b) in the case of a next day delivery service, on the next business day after being placed in the possession of such delivery service with next day delivery charges prepaid, and (c) in the case of mail, five (5) days after deposit in the postal system, certified or registered mail, return receipt requested and postage prepaid. Either party may change its address and telecopy number by written notice to the other as provided above; and

6. The Bank shall indemnify and hold harmless the Landlord for any and all damage caused as a result of the exercise of the Bank's rights hereunder.

This Landlord's Consent and Waiver may not be changed or terminated orally and inures to the benefit of and is binding upon the Landlord and its successors and assigns, and inures to the benefit of and is binding upon the Bank and its successors and assigns.

IN WITNESS WHEREOF, Landlord and Bank have each executed this Landlord's Consent and Waiver or caused it to be executed by an officer thereunto duly authorized, and the appropriate seal to be hereunto affixed, this day of ___, 20_.

LANDLORD:

	By: (Name) (Title)
	BANK:
	By: (Name) (Title)
COMMONWEALTH OF MASSAC	HUSETTS
County, ss.	
proved to me by satisfactory eviden state or federal governmental docur known to me who knows the above s	re me, the undersigned Notary Public, personally appeared the above-named , ce of identification, being (check whichever applies): \square driver's license or other nent bearing a photographic image, \square oath or affirmation of a credible witness signatories, or \square my own personal knowledge of the identity of the signatory, to be ove, and acknowledged the foregoing to be signed by her/him voluntarily for its
Print Name: My commissi	ion expires:
STATE OF	
County, ss.	
proved to me by satisfactory eviden state or federal governmental docur known to me who knows the above s the person whose name is signed above, that l	re me, the undersigned Notary Public, personally appeared the above-named , ce of identification, being (check whichever applies): \square driver's license or other nent bearing a photographic image, \square oath or affirmation of a credible witness signatories, or \square my own personal knowledge of the identity of the signatory, to be ne/she signed it as for to be signed by her/him voluntarily for its stated purpose.
Print Name: My commission expi	res:

Certain identified information has been omitted from this exhibit because it is (i) not material and (ii) of the type that the registrant treats as private or confidential. [***] indicates the information has been omitted.

FINAL EXECUTION COPY

FIRST AMENDMENT TO LEASE

THIS FIRST AMENDMENT TO LEASE ("Amendment"), dated as of May 4, 2023 (the "Effective Date"), is entered into by and between **201 BURLINGTON ROAD OWNER, LLC** ("Landlord") and **LANTHEUS MEDICAL IMAGING, INC.** ("Tenant").

BACKGROUND

Landlord and Tenant are parties to a lease dated February 14, 2022 (the "Lease") for premises consisting of approximately 46,526 rentable square feet of floor area on the 1st, 2nd, and 3rd floors in the south portion of the building (the "Existing Premises") located on the parcel of land described in Exhibit A to the Lease (such parcel of land, the "Property") and commonly known as 201 Burlington Road, Bedford, Massachusetts (the "Building"). The south portion of the Building is known as the "South Building."

Tenant desires to lease from Landlord, and Landlord desires to lease to Tenant additional premises consisting of approximately 41,655 rentable square feet on the first floor of the north portion of the Building, substantially as shown on <u>Exhibit A</u> attached hereto (the "Expansion Premises") for its use as a research and development laboratory as further set forth in Section 10 hereof. The north portion of the Building is known as the "North Building."

Landlord and Tenant also desire to extend the Original Term and make certain changes to the Lease on account of the inclusion of the Expansion Premises, set forth herein.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and the mutual undertakings set forth below, Landlord and Tenant hereby agree as follows:

- 1. <u>Recitals.</u> The background recitals set forth above are incorporated herein as if fully set forth. Capitalized terms used herein without definition shall have the meanings given them in the Lease.
- 2. <u>Extension of Original Term.</u> The Original Term of the Lease is hereby extended with respect to the Existing Premises and shall run co-terminously with the Expansion Premises Term (hereinafter defined), expiring on the redefined Expiration Date set forth below. The extended Original Term shall be on all the same terms and conditions of the Lease, except as amended hereby. The Commencement Date of the Original Term shall be determined as set forth in Section 2.2 of the Lease.
- 3. <u>Expansion Premises</u>. (a) The Expansion Premises are hereby leased to Tenant on all the terms and conditions of the Lease, as hereby amended, for a term (the "<u>Expansion Premises Term</u>") commencing on the Expansion Premises Commencement Date (defined below) and ending on the date that is fifteen (15) Lease Years from the Expansion Premises Rent Commencement Date (hereinafter defined) plus the first four (4) full calendar months immediately following Lease Year 15. That 4-month period is referred to herein as the "<u>Partial Lease Year 16</u>". In implementation of the foregoing, the Expiration Date set forth in Section 1.1 of the Lease is hereby redefined to mean the last day of Partial Lease Year 16. That date, once determined shall be memorialized in the Commencement Date Notification required under Section 2.2 of the Lease. Landlord's Expansion Premises Work (hereinafter defined) shall be sufficiently completed by September 1, 2023, subject to Force Majeure events (defined in Section 10.5 of the Lease) and/or Tenant Delay (defined in Section 4 of Exhibit B attached hereto). The Expansion Premises are leased to Tenant without any representation or warranty

by Landlord as to their suitability for Tenant's use and occupancy, and without any requirement for Landlord to construct or prepare the Expansion Premises for Tenant's use other than the "Landlord's Expansion Premises Work" defined in and in accordance with Exhibit B attached hereto and incorporated by reference as if fully set forth herein.

- (b) The term "Lease Year" as used in this Amendment shall mean a period of twelve (12) consecutive full calendar months. The 1st Lease Year shall begin on the Expansion Premises Rent Commencement Date if that date is the first day of a full calendar month; if not, then the 1st Lease Year shall commence upon the first day of the full calendar month immediately following the Expansion Premises Rent Commencement Date and shall include the partial month in which the Expansion Premises Rent Commencement Date actually occurred. Each succeeding Lease Year shall commence upon the anniversary date of the 1st Lease Year. It is understood that the term "Lease Year" defined in Section 2.2 of the Lease applies only to the Existing Premises and shall have no applicability to the Expansion Premises.
- 4. <u>Tenant Improvement Access Date</u>. The "<u>Tenant Improvement Access Date</u>" shall be the date on which Landlord's Expansion Premises Work is sufficiently completed to the extent necessary to deliver the Expansion Premises to Tenant for commencement of its improvement work, and certified as sufficiently completed by Landlord's architect (provided such date shall not be earlier than September 1, 2023), subject to delay caused by Force Majeure events and/or any Tenant Delay. The term "sufficiently completed" as used in this Amendment shall mean that all work specified in the Landlord's approved construction schedule attached hereto as <u>Exhibit B-2</u> has been completed up to the date specified therein (or specified in the final approved construction schedule, if different) for delivery of the Expansion Premises to Tenant for commencement of Tenant's Work. Upon the Tenant Improvement Access Date, Tenant shall have access to the Expansion Premises to commence construction of "<u>Tenant's Work</u>" defined in and in accordance with said <u>Exhibit B</u>. Landlord and Tenant will work together to coordinate their respective construction schedules in an effort to achieve an earlier Tenant Improvement Access Date with respect to the office portion of the Expansion Premises. The Tenant Improvement Access Date shall also serve to mean the "<u>Expansion Premises Commencement Date</u>" for all purposes under this Amendment, including determination of the beginning of the Expansion Premises Term. As of the Expansion Premises Commencement Date all references in the Lease to the "Premises" shall mean the Existing Premises and the Expansion Premises.
- 5. <u>Early Office Use</u>. Tenant shall have the conditional right, but not the obligation, to commence beneficial use and occupation of designated office portions of the Expansion Premises prior to the Expansion Premises Commencement Date, so long as such early office use will not interfere with Landlord's core and shell work or delay completion of Landlord's Expansion Premises Work. Tenant's early office use shall be limited to that office space designated by Landlord based on the preliminary layout plan attached hereto as <u>Exhibit A-</u>
 1. Tenant's use of the office space designated by Landlord as available for Tenant's early use is not to be construed as causing any resulting "Tenant Delay" under Section 4 of <u>Exhibit B</u> attached hereto. However, Tenant's use of undesignated office space shall not be excused from any resulting Tenant Delay. Prior to commencing beneficial use of any designated office space, Tenant must provide Landlord with a copy of a partial certificate of occupancy for the office portion issued by the Town of Bedford. If Tenant commences beneficial use of any amount of office space prior to the Expansion Premises Commencement Date, then Tenant shall pay to Landlord pro-rated Fixed Rent (at the triple net rate for Lease Year 1 set forth in Section 6 below), <u>plus</u> Tenant's Percentage of Taxes and Operating Costs, all calculated on the basis of the rentable square feet of floor area used by Tenant, until the Expansion Premises Rent Commencement Date. On the Expansion Premises Rent Commencement Date the early use portion shall no longer be a separate and distinct area for which Tenant pays only pro-

rated Fixed Rent, it being agreed that on the Expansion Premises Rent Commencement Date Tenant shall have the obligation to pay full Expansion Premises Fixed Rent based on the Expansion Premises as a whole, in accordance with Section 6(i) below. For clarity, the time period of any beneficial office use pursuant to this Section 5 shall not be counted in determining

the

occurrence of the Expansion Premises Rent Commencement Date for any purpose under this Amendment. Notwithstanding Tenant's permitted early use, Tenant agrees to diligently pursue completion of its construction work in the remainder of the Expansion Premises and issuance of a partial temporary certificate of occupancy.

6. Expansion Premises Fixed Rent and Additional Rent. Tenant's obligation to pay

- (a) Fixed Rent for the Expansion Premises (the "Expansion Premises Fixed Rent") and (b) Additional Rent on account of Operating Costs and Taxes for the Expansion Premises shall commence on the date that is eleven (11) months from the Expansion Premises Commencement Date (such date is referred to as the "Expansion Premises Rent Commencement Date"). For the avoidance of doubt, the Net Lease Conversion Date (hereinafter defined), does not accelerate Tenant's obligation to pay Additional Rent on account of Operating Costs and Taxes for the Expansion Premises prior to the Expansion Premises Rent Commencement Date. Commencing as of the Expansion Premises Rent Commencement Date, Tenant shall pay to Landlord Expansion Premises Fixed Rent for the Expansion Premises at the same time and in the same manner set forth in Section 4.1 of the Lease, at the Annual Expansion Premises Fixed Rent Rate and the Monthly Expansion Premises Fixed Rent Rate as set forth in the rent schedule below. It is intended that Tenant's obligation to pay Expansion Premises Fixed Rent for the Expansion Premises shall be phased-in during the first sixteen (16) calendar months following the Expansion Premises Rent Commencement Date¹. The Expansion Premises Fixed Rent stated in the rent schedule in this Section 6 has been determined as follows:
 - (a) From the Expansion Premises Rent Commencement Date (plus any partial month in which such date occurs) through the twelve (12th) full calendar month thereafter, the Expansion Premises Fixed Rent is calculated based on approximately 27,000 rentable square feet of the Expansion Premises;
 - (b) From the first day of the 13th full calendar month from the Expansion Premises Rent Commencement Date through the end of the sixteenth (16th) full calendar month thereafter, the Expansion Premises Fixed Rent is \$0.00, it being agreed that Tenant shall pay no Fixed Rent for the Expansion Premises during the aforesaid 4-calendar month period; and
 - (c) From the first day of the 17th full calendar month from the Expansion Premises Rent Commencement Date) through the remainder of the Expansion Premises Term, the Expansion Premises Fixed Rent is calculated based on the entire 41,655 rentable square feet of the Expansion Premises as stated in the rent schedule below.

<u>Period</u>	Annual Fixed Rent Rate	Monthly Fixed Rent Rate
Lease Year 1	\$1,768,500.00	\$147,375.00
Calendar month 13 – Calendar month 16	\$0.00	\$0.00
Calendar month 17 – end of Lease Year 2	\$2,810,254.56 (annualized)	\$234,187.88
Lease Year 3	\$2,894,562.24	\$241,213.52
Lease Year 4	\$2,981,399.04	\$248,449.92
Lease Year 5	\$3,070,841.04	\$255,903.42
Lease Year 6	\$3,162,966.24	\$263,580.52
Lease Year 7	\$3,257,855.28	\$271,487.94
Lease Year 8	\$3,355,590.96	\$279,632.58
Lease Year 9	\$3,456,258.60	\$288,021.55

¹ It is understood that the "phase-in" schedule set forth in Section 1.1 and Section 4.1(a) of the Lease apply solely to the Existing Premises and shall have no applicability to the Expansion Premises. In Section 1.1, the terms "Initial Rent Commencement Date", "First Rent Adjustment Date", Second Rent Adjustment Date", and "Full Rent Commencement Date" are only applicable to the Existing Premises.

Lease Year 10	\$3,559,946.40	\$296,662.20
Lease Year 11	\$3,666,744.84	\$305,562.07
Lease Year 12	\$3,776,747.16	\$314,728.93
Lease Year 13	\$3,890,049.60	\$324,170.80
Lease Year 14	\$4,006,751.04	\$333,895.92
Lease Year 15	\$4,126,953.60	\$343,912.80
Partial Lease Year 16	\$4,250,762.16 (annualized)	\$354,230.18

- 7. <u>Existing Premises Extended Term Fixed Rent</u>. Commencing retroactively as of January 1, 2023, the Annual Fixed Rent Rate and the Monthly Fixed Rent Rate for the Existing Premises shall be determined pursuant to the provisions of Section 8(b) hereof, as shown on <u>Exhibit D</u> attached hereto and incorporated herein.
- 8. <u>Conversion to Net Fixed Rent for the Existing Premises</u>. (a) Following the end of calendar year 2023, the Existing Premises Fixed Rent presently set forth in the Lease shall be converted from a gross rent to a triple net rent effective retroactively as of January 1, 2023, and additionally Tenant shall then be responsible for its proportionate share of net rent Operating Costs and Taxes. For the avoidance of doubt, Tenant shall continue to pay Fixed Rent for the Existing Premises on a gross basis in the amounts set forth in Section 1.1 of the Lease without adjustment until the determination of the Rate Reduction (defined below) has been made in 2024, at which time the Rate Reduction will be applied retroactively to January 1, 2023 to adjust the Annual Fixed Rent paid by Tenant during calendar year 2023.
- (b) Effective retroactively as of January 1, 2023, the Annual Fixed Rent Rate per rentable square foot, shall be reduced, dollar for dollar, by the Rate Reduction. The term "Rate Reduction" shall mean an amount equal to (a) the 2023 calendar year Operating Costs (grossed up to reflect 100% occupancy) and Taxes (adjusted, if necessary, to reflect a fully assessed property) for fiscal tax year 2023 multiplied by (b) the Tenant's Percentage (as set forth in the original Lease), the product of which is then (c) divided by 46,526. A more illustrative example of the conversion from gross to net Fixed Rent is attached hereto as Exhibit D and is incorporated herein by reference. No later than ninety (90) days after the end of calendar year 2023, Landlord shall furnish Tenant with an itemized statement of the Operating Costs for calendar year 2023 and Taxes for fiscal year 2023 for the Existing Premises, increased to reflect

all additional costs and expenses that would have been incurred had the Building been 100% occupied, prepared, allocated and computed in accordance with generally accepted accounting principles, along with (a) Landlord's determination of the Rate Reduction, and (b) a revised schedule of Existing Premises Annual Fixed Rent for the remainder of the Term (the "Revised Rent Schedule").

- (c) Within fifteen (15) business days after Landlord gives Tenant Landlord's determination of the Rate Reduction, Tenant shall notify Landlord whether Tenant disputes or agrees with Landlord's calculation of the Rate Reduction. If Tenant disagrees with Landlord's determination, or requires additional information to confirm its agreement with Landlord's calculation of the Rate Reduction, then Tenant shall have the right to examine and review Landlord's books and records establishing Operating Costs solely for calendar year 2023 and Taxes solely for fiscal tax year 2023 in order to confirm Landlord's determination of the Rate Reduction. In so doing, Tenant shall follow the procedures set forth in Section 4.2.3 of the Lease for conducting an audit, including the dispute resolution procedure set forth therein. The time periods set forth in Section 4.2.3 for the parties to take action or give notice shall have no applicability to Tenant's audit under this Section 8(c). Instead, the time periods set forth below shall govern when counting the number of days in which a party must take action or give notice:
- (i) Tenant shall give Landlord at least five (5) business days prior written notice that it intends to conduct an audit; such audit must commence within ten (10) business days following Landlord's receipt of Tenant's audit notice and must be completed no later than thirty (30) days following the date the audit commenced.
- (ii) If Landlord disputes the results of Tenant's audit, Landlord shall have ten (10) business days after receipt of Tenant's audit results to obtain a certified public accountant to conduct Landlord's own audit.
- (iii) The parties shall have ten (10) business days after completion of Landlord's audit to resolve the dispute or select a mutually agreed upon independent certified public accountant who shall make a final determination, to be rendered no later than fifteen (15) business days after appointment (the third accountant's fees are to be shared equally between Landlord and Tenant).

The findings of such audit shall be conclusive as to the Operating Costs for calendar year 2023 and Taxes for fiscal tax year 2023 and shall be used in calculating the Rate Reduction in accordance with the methodology set forth in Section 8(b) above.

- (d) Promptly after Landlord and Tenant have agreed upon the Rate Reduction, Landlord shall furnish Tenant with the Revised Rent Schedule. Any overpayment of the Existing Premises Fixed Rent during all of calendar year 2023 and during the period from January 1, 2024 through the date of the Rate Reduction determination shall be promptly refunded to Tenant, or, at the election of Tenant, credited against the subsequent monthly installments of the net Fixed Rent for the Existing Premises. Any underpayments shall be paid by Tenant with the subsequent monthly installments of Fixed Rent for the Existing Premises.
 - 9. Effective as of January 1, 2024, (the "Net Lease Conversion Date"), the Lease shall be amended as follows:
 - (a) In Section 4.2, the first sentence thereof shall be deleted and the following substituted in its place: "Tenant covenants and agrees to pay its insurance costs and personal property taxes, and in order that the Fixed Rent be absolutely net to Landlord, Tenant agrees to pay to Landlord, as Additional Rent, utility charges, (exclusive of charges paid pursuant to Section 4.2.5(c)) and its proportionate share of Operating Costs and Taxes with respect to the Premises as provided in this Section 4.2, as follows:"

(b) In Section 4.2.1:

(i) The first paragraph shall be deleted in its entirety and the following substituted in its place:

"Real Estate Taxes. Tenant covenants to pay to Landlord, as Additional Rent, for each fiscal tax period (a "Tax Year") partially or wholly included in the Term, Tenant's Percentage of Taxes (as hereinafter defined). Tenant shall remit to Landlord, on the first day of each calendar month during the Term, estimated payments on account of Taxes, such monthly amounts to be sufficient to provide Landlord, by the time real estate tax payments are due and payable to any governmental authority responsible for collection of same, a sum equal to the Tenant's Percentage of Taxes, as reasonably estimated by Landlord from time to time on the basis of the most recent tax data available. The initial calculation of the monthly estimated payments shall be based upon Landlord's initial estimate of Tenant's Percentage of Taxes for the Tax Year and upon quarterly payments being due to the governmental authority on August 1, November 1, February 1 and May 1, and shall be made when the Commencement Date has been determined. If the total of such monthly remittances for any Tax Year is greater than the Tenant's Percentage of Taxes for such Tax Year, Landlord shall promptly pay to Tenant, or credit against the next accruing payments to be made by Tenant pursuant to this subsection 4.2.1, the difference; if the total of such remittances is less than the Tenant's Percentage of Taxes for such Tax Year, Tenant shall pay the difference to Landlord at least ten (10) days prior to the date or dates within such Tax Year that any Taxes become due and payable to the governmental authority (but in any event no earlier than ten (10) days following a written notice to Tenant, which notice shall set forth the manner of computation of Tenant's Percentage of Taxes). This section shall survive the expiration or earlier termination of this Lease.'

- (ii) the second, third and fourth paragraphs shall be revised by deleting the phrase "Tax Excess" in each instance where it appears and inserting the term "Taxes" in its place.
- (iii) the fifth paragraph shall be deleted in its entirety.

(c) <u>In Section 4.2.3</u>:

(i) The first paragraph shall be deleted in its entirety and the following substituted in its place:

"Tenant covenants to pay to Landlord the Tenant's Percentage of Operating Costs (as hereinafter defined) incurred by Landlord in any calendar year. Tenant shall remit to Landlord, on the first day of each calendar month during the Term, estimated payments on account of Tenant's Percentage of Operating Costs, such monthly amounts to be sufficient to provide Landlord, by the end of the calendar year, a sum equal to Tenant's Percentage of Operating Costs, as reasonably estimated by Landlord from time to time. The initial monthly estimated payments shall be in an amount equal to 1/12th of Landlord's initial estimate of Tenant's Percentage of Operating Costs for the calendar year. If, at the expiration of the calendar year in respect of which monthly installments of Operating Costs shall have been made as aforesaid, the total of such monthly remittances is greater than the actual Tenant's Percentage of Operating Costs for such year, Landlord shall promptly pay to Tenant, or credit against the next accruing payments to be made by Tenant pursuant to this subsection 4.2.3, the difference; if the total of such remittances is less than the Tenant's Percentage of Operating Costs for such calendar year, Tenant shall pay the difference to Landlord within twenty (20) days from the date Landlord shall furnish to Tenant an itemized statement of the Operating Costs, prepared, allocated and computed in accordance with generally accepted accounting principles. Any reimbursement for Operating Costs due and payable by Tenant with respect to periods of less than twelve (12) months shall be equitably prorated. This section shall survive the expiration or earlier termination of this Lease."

- (ii) The following shall be included in the definition of "Operating Costs": "all costs of maintaining and repairing any common lab- specific systems or equipment in the Building".
- (iii) The parenthetical text, "(including the calendar year for Base Operating Costs)" in the fourth paragraph (the first full paragraph on page 18 of the Lease) shall be deleted.
- (iv) The last 2 sentences in the fourth paragraph shall be deleted.
- (v) The word "Excess" shall be deleted from the fifth paragraph (audit right, being the last paragraph of Section 4.2.3) in each instance it appears.
- (vi) Any property management fee for the Expansion Premises that is included in Operating Costs shall not exceed market rates for comparable premises in comparable buildings in the Burlington/Bedford area.
- (vii) Anything to the contrary notwithstanding, the list of exclusions from Operating Costs remains unchanged and in full force and effect.
- (d) The terms "Base Operating Costs" and "Base Taxes" shall be deleted together with the definitions given them in Section 1.1.
 - 10. Amendment of Certain Defined Terms.
 - (a) As of the Effective Date, Section 1.1 of the Lease, Reference Data, shall be amended as set forth below (subject to Section 9(d) above):

Base Operating Costs Operating Costs for calendar year 2023

Base Taxes: Taxes for fiscal tax year 2023 (i.e., July 1, 2022 – June 30, 2023)

- (b) As of the Expansion Premises Commencement Date, Section 1.1 of the Lease, *Reference Data*, shall be amended as set forth below:
- (i) The definition of "*Premises*" shall be deleted and the following shall be substituted in its place:
 - "A portion of the first floor and the entire second and third floors of the South Building, substantially as shown on Exhibit B attached hereto, and a portion of the first floor of the North Building, as shown on Exhibit B-1 attached hereto."
- (ii) The definition of "*Rentable Floor Area of Premises*" shall be deleted and the following shall be substituted in its place:
 - "Approximately 46,526 rentable square feet consisting of (i) approximately 2,991 rentable square feet on the first floor, and (ii) 43,535 rentable square feet on the 2^{nd} and 3^{rd} floors in total in the South Building, and (iii) approximately 41,655 rentable square feet on the first floor of the North Building."

Prior to full execution of this Amendment, Landlord shall provide Tenant's architect with Landlord's calculations of the usable and rentable square foot measurement of the Expansion Premises, determined in accordance with the modified BOMA 2017 for Office Buildings: Standard Methods of Measurement (ANSI/BOMA Z65.1—2017).

Tenant shall have the right to use its own architect to verify Landlord's measurement, prior to execution of this Amendment.

- (iii) The definition of "*Tenant's Percentage*" shall be deleted and the following shall be substituted in its place:
 - "The ratio of the Rentable Floor Area of the Premises to the total rentable area of the Building (which is approximately 132,684 square feet), which shall initially be deemed to be 66.46%."
- (iv) The definition of "*Permitted Uses*" shall be deleted and the following shall be substituted in its place:
 - "Research and development laboratory, including, without limitation, the testing and quality control of radiopharmaceutical products using the chemicals and isotopes set forth in the Chemical Inventory and Isotope List (as defined in Section 24(g) of the First Amendment), as updated from time to time, office use and other related lawful uses supporting the foregoing."
- 11. <u>Landlord's Maintenance and Repair Obligations.</u> Effective as of the Net Lease Conversion Date, Section 5.1.5 shall be amended by inserting the following phrase "and any common lab-specific systems and equipment" in the 5th line after "air-conditioning systems".
- 12. <u>Tenant's Maintenance and Repair Obligations</u>. Effective as of the Net Lease Conversion Date, Section 6.1.3 of the Lease shall be amended as follows:
 - (a) By adding the following sentence to the end of that Section: "Notwithstanding Landlord's obligations under Section 5.1.5 above, if any repairs and maintenance are required to Tenant's alterations, installations, structural reinforcements or other items that impact the floor slab on account of Tenant's Expansion Premises Work, Tenant, not Landlord, shall be responsible for the repairing and maintaining the same, at its sole expense."
 - (b) In the 5^{th} line of the first paragraph, the words "located within the Premises and" shall be deleted.
- 13. Tenant's Additional Insurance Requirements for the Expansion Premises. Effective as of the date Tenant first occupies the Premises for lab purposes, Tenant, in addition to maintaining the insurance coverages required under Section 4.2.4 of the Lease, shall take out and maintain Pollution/Environmental Liability insurance with minimum limits of [***] for each policy. Tenant shall also take out and maintain Special Risk Property insurance (to include coverage for improvements and betterments) with the usual extended coverage endorsements covering all Tenant's furniture, furnishings, fixtures and equipment and all tenant improvements and alterations made by or on behalf of Tenant, including the Tenant's Work in the Expansion Premises, in an amount not less that their full replacement value, from time to time. Tenant represents that its Special Risk Property insurance includes builder's risk or similar coverage and shall provide Landlord with a certificate of insurance evidencing the same prior to commencing Tenant's Work. The proceeds of such insurance shall be used for the repair or replacement of the property insured, except that if the Lease is terminated as provided in Article 7 thereof, in the event of a loss, the proceeds attributable to the Tenant's Work and other tenant improvements and alterations shall be paid and belong to Landlord, and the proceeds attributable to Tenant's

personal property shall be paid and belong to Tenant. Tenant's insurance policy shall include business interruption insurance with limits not less than the equivalent of 12 months of rent, with extra expense coverage, soft costs at the full applicable limit per construction contract and budget. Such policy shall list Landlord as loss payee as their interests may appear. Section 4.2.4.2 of the Lease shall have no applicability to the Expansion Premises, however Tenant shall comply with all other provisions and requirements set forth therein as they relate to the Expansion Premises.

- 14. <u>Lease Exhibits</u>. <u>Exhibit A</u>, the Plan Showing the Expansion Premises, attached hereto, shall be deemed <u>Exhibit B-1</u> for the purposes of Section 1.1 of the Lease.
- 15. <u>Parking</u>. As of the Expansion Premises Commencement Date, the number of parking spaces Tenant is permitted to use under Section 2.1 of the Lease shall be increased from 158 unreserved parking spaces to 291 unreserved parking spaces.
- 16. <u>Utilities</u>. Tenant shall pay Landlord for its use of electricity, gas and water (including sewer charges) for ordinary office cleaning and lavatory use in the Expansion Premises through Operating Costs. All charges for the cost of electricity for Tenant's lights, outlets, VAV boxes/Water Source Heat Pumps, any special requirements such as supplemental HVAC units and for any above-normal office usage of electricity in the Expansion Premises, each as measured through a sub-meter (collectively, "Expansion Premises Tenant Electric") shall be paid by means of monthly escrow payments billed to Tenant in accordance with Section 4.2.5(b) of the Lease. Water and gas for Tenant's excess needs for its lab uses shall be sub-metered and all associated charges and costs shall be paid by means of monthly escrow payments billed to Tenant as Utilities Charges under said Section 4.2.5(b). All references to "Utilities Charges" in Section 4.2.5(b) of the Lease shall be deemed to mean and include all charges on account of Expansion Premises Tenant Electric and Tenant's excess water and gas usage in the lab areas.
 - 17. Extension Options. Section 2.3 of the Lease is hereby deleted and the following is substituted in its place:
 - "2.3 Extension Options.
 - (a) Tenant shall have two (2) options (each, an "Extension Option") to extend the Term of this Lease for an additional period of five (5) years each (each, an "Extended Term"). The first such 5-year period (the "First Extended Term") shall begin immediately upon the expiration of the Original Term and the second such 5-year period (the "Second Extended Term") shall begin immediately upon the expiration of the First Extended Term, provided that each of the following conditions has been satisfied:
 - (i) As of the date of the Extension Notice (defined below) and as of the commencement of the First Extended Term, and, if applicable, as of the commencement of the Second Extended Term, Tenant shall not (x) be in monetary default and/or (y) be in non-monetary default beyond notice and cure period and/or (z) have been in default of its obligations under this Lease beyond any applicable grace period more than once during the immediately preceding 12- month period in each instance, and is occupying 100% of the Premises;
 - (ii) Tenant and/or Guarantor shall have had a net income for the 12- month period immediately preceding the date of the Extension Notice and for the 12-month period immediately preceding the commencement of the First Extended Term, and if applicable, the Second Extended Term; and

(iii) Simultaneously with the delivery of an Extension Notice and also at the commencement of the First Extended Term, and, if applicable, the Second Extended Term, Tenant shall have delivered to Landlord (x) consolidated audited statements (collectively, an "Audited Statement") for Tenant and/or Guarantor for the preceding year, prepared using generally accepted accounting principles, and (y) unaudited consolidated statements for Tenant and/or Guarantor for the most recently completed quarter prepared using generally accepted accounting principles, all of which are evidencing such net income during each of the periods specified in clause (ii) hereinabove; it being agreed that if and when an additional Audited Statement is completed between the delivery of any Extension Notice and the commencement of the applicable Extended Term (an "Updated Audited Statement"), Tenant will deliver such Updated Audited Statement to Landlord (and such Updated Audited Statement must satisfy the requirements set forth herein in all respects).

If the above conditions have not been satisfied, then the Extension Option shall be void and the Extension Notice previously sent by Tenant shall be nullified and of no effect. Notwithstanding anything to the contrary contained herein, Landlord hereby reserves the right, in its sole and absolute discretion, to waive any condition set forth in clauses (i), (ii), and (iii) of this Section 2.3. No such waiver shall be effective unless communicated by Landlord to Tenant in writing and shall be used for the sole purpose of giving effect to an applicable Extension Option and no other purpose.

- (b) All of the terms, covenants and provisions of this Lease shall apply to each Extended Term, except that the Annual Fixed Rent Rate for the applicable Extended Term shall be the fair-market rental rate for comparable uses in comparable Class A buildings in the Burlington/Bedford area as of the commencement of such Extended Term ("Market Rate"), as designated by Landlord. If Tenant shall elect to exercise an Extension Option, it shall do so by giving Landlord written notice (the "Extension Notice") of its election not later than twelve (12) months prior to the expiration of the Original Term of this Lease, or the First Extended Term, as the case may be, time being of the essence thereof. If timely and properly Tenant gives such notice and satisfies the conditions specified above, the extension of this Lease shall be automatically effected without the necessity of the execution of any additional lease or amendment documents. Tenant's Extension Option is personal to the initial named Tenant, Lantheus Medical Imaging, Inc. (and any assignee or subtenant pursuant to a Permitted Transfer, as defined in Section 6.2.1 below), and may not otherwise be assigned under any circumstances. The Original Term and the Extended Term shall be collectively referred to as the "Term" or the "term".
- (c) Not later than thirty (30) days following the giving of Tenant's Extension Notice, Landlord shall notify Tenant of Landlord's determination of the Market Rate for the applicable Extended Term. Within fifteen (15) business days after Landlord gives Tenant Landlord's determination of the Market Rate, Tenant shall notify Landlord whether Tenant accepts or disputes such rate. If Tenant disagrees with Landlord's determination, then Landlord and Tenant shall commence negotiations to agree upon the Market Rate. In any event, the Annual Fixed Rent Rate for the First Extended Term shall not be less than the Annual Fixed Rent Rate in effect

immediately prior to the commencement of the First Extended Term, inclusive of any escalations (it being agreed that no such limitation shall apply to the Second Extended Term). If Landlord and Tenant are unable to reach agreement on the Market Rate within thirty (30) days after the date on which Landlord first gave Tenant Landlord's proposal for the Market Rate, then the Market Rate shall be determined as provided below.

- (d) If Landlord and Tenant are unable to agree on the Market Rate by the end of said thirty (30)-day period, then within five (5) business days thereafter, Landlord and Tenant shall each simultaneously submit to the other in a sealed envelope its good faith estimate of the Market Rate. If the higher of such estimates is not more than one hundred five percent (105%) of the other estimate, then the Market Rate shall be the average of the two estimates. If the matter is not resolved by the exchange of estimates, then Market Rate shall be determined by an independent arbitrator as set forth below.
- Within seven (7) days after the exchange of estimates, the parties shall select, as an arbitrator, a mutually acceptable commercial real estate broker or appraiser licensed in the Commonwealth of Massachusetts specializing in the field of commercial office leasing in the Burlington/Bedford market area, having no less than ten (10) years' experience (an "Approved Arbitrator"). If the parties cannot agree on such person, then within a second period of seven (7) days, each shall select one Approved Arbitrator and the two appointed Arbitrators shall, within five (5) days, select a third Approved Arbitrator who shall be the final decision-maker (the "Final Arbitrator"). If one party shall fail to timely make such appointment, then the person chosen by the other party shall be the sole arbitrator. Once the Final Arbitrator has been selected as provided for above, then, as soon thereafter as practicable, but in any case within fourteen (14) days after his or her appointment, the arbitrator shall determine the Market Rate by selecting either the Landlord's estimate of Market Rate or the Tenant's estimate of Market Rate. Such arbitrator must choose the proposed Market Rate that he/she determines is closest to the actual market rental rate for the Premises. There shall be no discovery or similar proceedings. The arbitrator's decision as to which estimate shall be the Market Rate for the Extended Term shall be rendered in writing to both Landlord and Tenant and shall be final and binding upon them and shall be the Annual Fixed Rent for the Extended Term. The costs of the Final Arbitrator will be equally divided between Landlord and Tenant. Any fees of any Approved Arbitrator or counsel engaged by Landlord or Tenant, however, shall be borne by the party that retained such Approved Arbitrator or counsel. If the dispute between the parties as to a market rate has not been resolved before the commencement of the Extended Term, then Tenant shall continue to pay Fixed Rent under the Lease based upon the Fixed Rent Rate in effect immediately prior to the commencement of the Extended Term until either the agreement of the parties as to the market rate, or the decision of the Final Arbitrator, as the case may be, at which time Tenant shall pay any underpayment of Fixed Rent to Landlord.
- (f) Once the Market Rate has been determined, the parties shall promptly execute an amendment to this Lease setting forth the Fixed Rent for the Premises during the applicable Extended Term.
- (g) With respect to any assignment or subletting during the Term of this Lease, such assignment shall not include the Extension Options granted to Tenant, and such sublease shall be for a term expiring no later than the Expiration Date."

- 18. <u>Landlord's Termination Right</u>. Section 2.4 of the Lease, *Landlord's Termination Right*, is hereby deleted and of no further effect.
- 19. <u>Right of First Offer</u>. The following new Section 2.4 is hereby added to the Lease:

"2.4 Right of First Offer to Lease.

- (a) During the Original Term of this Lease, Tenant shall have an ongoing right of first offer to lease the ROFO Space (hereinafter defined) as it becomes available for lease, subject to the terms and conditions contained below.
- (b) "ROFO Space" shall mean all or any tenant space in the North Building. Tenant's rights shall be subject to all rights of existing tenants to lease any ROFO Space, whether by expansion, extension, rights of first offer, or otherwise. [***] is the only tenant of the Building having rights to lease the ROFO Space.
- (c) If Landlord makes a bona fide written offer to lease ROFO Space to a prospective tenant (a "<u>Proposal</u>"), Landlord shall first notify Tenant in writing of the opportunity to lease the offered ROFO Space (the "<u>Opportunity Notice</u>"), on the condition that (i) the initial Tenant named herein, Lantheus Medical Imaging, Inc. or an affiliate (i.e., any person or entity which is controlled by or under common control with Tenant, or which controls Tenant), occupies the entire Premises and (ii) Tenant shall not (x) be in monetary default at the time Landlord delivers the Opportunity Notice and/or (y) be in non-monetary default beyond notice and cure period at the time Landlord delivers the Opportunity Notice, and/or (z) have been in default of its obligations under this Lease beyond any applicable grace period more than once during the 12-month period immediately preceding the time when Landlord delivers the Opportunity Notice. If the conditions in the preceding sentence are not fully satisfied at the time of the Opportunity Notice, Tenant's ROFO under this Section 2.4 shall not apply in that instance and Landlord shall be free to lease such ROFO Space to another tenant.
- (d) Within fifteen (15) business days after Tenant's receipt of Landlord's Opportunity Notice, Tenant shall provide written notice to Landlord indicating whether it is interested in the possibility of leasing the ROFO Space. If Tenant timely so indicates its interest in leasing the ROFO Space and satisfies the conditions set forth in subparagraph (c), then Landlord shall notify Tenant in writing ("Landlord's Offer Notice") specifying the terms on which Landlord intends to offer to lease the ROFO Space to the general market except that the term of the ROFO Space will be coterminous with the then remaining portion of the Term of this Lease. Landlord's Offer Notice shall identify the ROFO Space and the terms of the offer in reasonable detail and shall include Landlord's determination of the Market Rate (as such term is defined in Section 2.3) for the ROFO Space. Tenant shall within fifteen (15) business days after receipt of Landlord's Offer Notice either accept or decline such offer in writing, with Tenant's failure to respond within such fifteen (15)-business day period being deemed an election to decline the offer.
- (e) If Tenant has declined Landlord's offer for the ROFO Space, or is deemed to have declined the offer, or does not satisfy the conditions specified in subparagraph (c), Landlord shall be free to offer such ROFO Space to another tenant. However, Landlord shall deliver a new Landlord's Offer

Notice prior to entering into a lease for such ROFO Space with a third party that provides for an effective rent that is 90% or less than the net effective rent specified in the initial Landlord's Offer Notice for that specific ROFO Space. Landlord's new Offer Notice shall be on such revised terms, and Tenant shall have ten (10) business days after receipt of the new offer to accept or reject such new offer. Notwithstanding the foregoing, Tenant's failure to exercise its right to lease any ROFO Space pursuant to any Offer Notice shall not terminate Tenant's continuing ROFO over all subsequent Proposals for ROFO Space during the Original Term of this Lease, provided, however, once Landlord enters into a lease for ROFO Space with another tenant, Tenant's rights under this Section 2.4 shall be subject to the rights of that tenant to lease such ROFO Space in the future, whether by rights of extension or otherwise. If following Landlord's initial Offer Notice and Proposal, Landlord does not execute a lease with another tenant for the ROFO Space, or is not in active negotiations with another tenant, Landlord shall deliver to Tenant a subsequent Opportunity Notice the next time Landlord delivers a new Proposal to another tenant for such ROFO Space. If Tenant declines the opportunity to lease the ROFO Space, the process shall repeat itself each time Landlord intends to issue a Proposal until either Tenant or another tenant leases the ROFO Space, whichever occurs first. However, Tenant's ROFO shall only arise after the expiration of 90-days from the date of the previous Opportunity Notice, it being the intention of the parties that Landlord shall have no obligation to deliver an Opportunity Notice more than one time in any 90-day period even if Landlord issues a Proposal to another tenant during that 90-day period.

- (f) If Tenant has accepted Landlord's offer but believes that Landlord's Offer Notice does not reflect the Market Rate, Tenant shall so notify Landlord stating in detail Tenant's estimate of Market Rate. If the parties are unable to agree upon the Market Rate within thirty (30) days after the date of Landlord's Offer Notice, then the Market Rate shall be determined as set forth in subparagraph (g) below.
- (g) If Tenant timely elects to lease the ROFO Space, then the space shall, without further action by the parties, be leased by Tenant on the terms and conditions set forth in the Landlord's Offer Notice, subject to arbitration of the Market Rate if necessary, and otherwise on all of the other terms and conditions of this Lease. If Landlord and Tenant cannot agree on the Market Rate for any ROFO Space, their dispute shall be resolved by a procedure similar to that set forth in Section 2.3 (d), (e) and (f) of this Lease, and the result thereof shall be binding on Landlord and Tenant. When implementing the procedure set forth in said Section 2.3(d) (e) and (f), the phrase "Extended Term" used therein shall be deemed to mean the applicable ROFO term.
- (h) If there are 60 or fewer months remaining in the Term at the time of any Landlord's Offer Notice, Landlord shall have the option to require Tenant to concurrently exercise its next Extension Option, if available, as a prerequisite for accepting the ROFO Space. Tenant shall either (i) exercise an Extension Option, if available, in accordance with Section 2.3 of this Lease, or (ii) extend the Term for an additional period of time such that there will be a full sixty (60) calendar months remaining in the Term following the commencement of the ROFO term, such that in all cases the term of the Lease for the ROFO Space will be coterminous with the Term of the Lease. The Annual Fixed Rent Rate for the entire Premises, including the ROFO Space, during the Extended Term or the additional extended period pursuant to this Section 2.4(h), will be determined in the

same manner as set forth in Section 2.3 (b) for determining the Annual Fixed Rent Rate during the applicable Extended Term. If Tenant declines to exercise its Extension Option to extend the Lease for the applicable Extended Term, or if Tenant declines to extend the Term for the additional period contemplated in clause (ii) of this paragraph (h) then Tenant shall be deemed to have rejected Landlord's offer for the subject ROFO Space and Landlord will be free to offer such space to the general market.

(i) If Tenant timely elects to lease such ROFO Space, then the space shall, without further action by the parties, be leased by Tenant on the terms and conditions set forth in the Landlord's Offer Notice, and otherwise on all of the other terms and conditions of this Lease, and the parties shall memorialize such facts with a lease amendment.

This right of first offer is personal to Lantheus Medical Imaging, Inc. and may not be assigned to any other party except pursuant to a 'Permitted Transfer' under Section 6.2.1 of this Lease."

20. Tenant's Right of First Offer to Purchase.

- (a) Landlord agrees that if Landlord shall decide to offer for sale (to a party other than an affiliate of Landlord) all of the Landlord's interest in (i) the Building and the Property, (ii) the retained portion of the Property resulting from a subdivision of the Property, or (iii) any portion thereof comprising a condominium unit following Landlord's conversion of the Property to a condominium regime) (together, "Landlord's Interest in the Property") and on the condition (x) Tenant is not then in default of its obligations under the Lease, and (y) the original named Tenant, Lantheus Medical Imaging, Inc., occupies [***] of the Premises, Landlord shall first give notice thereof to Tenant. Such notice shall contain the terms and conditions upon which Landlord is proposing to offer to dispose of Landlord's Interests In Property and shall constitute an offer to Tenant under this Section 20 (herein called "Landlord's Submitted Offer"). For clarity, the subdivision of the Property or the submission of the Property to condominium status shall not be construed as a sale of all or any portion of Landlord's Interest In The Property.
- (b) Tenant shall have the right, to be exercised within [***] after its receipt of Landlord's Submitted Offer ("Tenant's Purchase ROFO Acceptance Period"), to give Landlord Tenant's acceptance of Landlord's Submitted Offer in writing, which shall be accompanied by a customary deposit, not to exceed [***], as established by Landlord in Landlord's Submitted Offer ("Tenant's Purchase ROFO Acceptance"). If Tenant shall duly and timely comply with the foregoing within Tenant's ROFO Acceptance Period, then not later than the [***] following the expiration of Tenant's Purchase ROFO Acceptance Period (the "Purchase ROFO Agreement Execution Period"), Landlord and Tenant, shall negotiate in good faith to finalize the terms and conditions of the sale and execute a binding purchase and sale agreement which shall include the terms set forth in Landlord's Submitted Offer, and such other customary terms and conditions that are usual and customary for a sale of similar buildings [***]. Thereafter, such disposition shall be carried out in accordance therewith. If both parties, having acted in good faith, fail to execute a binding purchase and sale agreement by the expiration of the [***] Purchase ROFO Agreement Execution Period, then the right of first offer to purchase granted to Tenant under this Section 20 shall automatically be suspended, and any deposit promptly refunded, and Landlord shall thereafter be free, expressly subject to the provisions of Section 20(c), to

enter into a binding purchase and sale agreement with any third party on terms substantially similar to the terms set forth in Landlord's Submitted Offer within the [***] period provided for in Section 20(c) below.

- (c) If Landlord has not entered into a binding purchase and sale agreement with a third party in accordance with the provisions of Section 20(b) within the [***] period following the expiration of Tenant's Purchase ROFO Agreement Execution Period, then Tenant's right of first offer under this Section 20 shall be reinstated in full force and effect in accordance with the provisions hereof. However, if Tenant's right of first offer has been terminated or extinguished as hereinafter set forth in this paragraph (c), there shall be no reinstatement of Tenant's right of first offer. If Landlord enters into a binding purchase and sale agreement with a third party on terms substantially similar to the terms set forth in Landlord's Submitted Offer within the aforesaid [***] period, the purchaser or purchasers shall not be bound by the provisions of this Section 20 and Tenant's right of first offer to purchase under this Section 20 shall terminate and be of no further force or effect. If Tenant has declined Landlord's Submitted Offer, Tenant's right of first offer to purchase shall be extinguished, provided, however, if Landlord thereafter offers Landlord's Interest in the Property to a third party upon material terms that are more favorable to such purchaser than the material terms contained in Landlord's Submitted Offer, or for a purchase price that is less than [***] of the purchase price that was offered to Tenant in Landlord's Submitted Offer, then Landlord shall re-offer the Property to Tenant on the terms offered to such third party ("Landlord's Revised Notice") and the same procedures shall apply with respect to Landlord's Revised Notice as set forth in paragraph (b) above, except that Tenant's Purchase ROFO Acceptance Period shall be reduced to [***].
- (d) Notwithstanding anything to the contrary contained in this Lease, Tenant shall not be entitled to exercise the rights provided to it under this Section 20 if at the time Tenant would otherwise by entitled to exercise such rights
 - (i) Tenant is then in default of its obligations under the Lease beyond the expiration of the applicable notice and cure periods under the Lease, or Landlord shall have given Tenant a notice of default which is not thereafter timely cured, (ii) the Lease shall no longer be in full force and effect, or (iii) Tenant (or a transferee pursuant to a Permitted Transfer under Section 6.2.1 of the Lease) itself is not occupying [***] of the Premises. Further, notwithstanding anything to the contrary contained in the Lease, Tenant's rights of first offer under this Section 20 shall be subject and subordinate to the provisions of any mortgage now or hereafter placed upon or affecting Landlord's Interest In The Property or any portion thereof and or interest therein and the rights of the holder thereof and to any renewals, extensions, replacements, and modifications of any such mortgage. Further, Tenant's right of first offer under this Section 20 shall not be applicable to any mortgagee of the Property. Landlord shall not be required to offer Landlord's Interest in the Property for sale to Tenant in the event of (i) a so-called "sale and leaseback financing" between Landlord, as borrower and any lender which involves a sale to such lender of all or any portion or portions of Landlord's Interest In The Property together with a simultaneous leasing back thereof by the lender as landlord to Landlord and/or affiliates of Landlord, as tenant, (ii) any transaction with a lender for the purposes of financing or refinancing the Property, Landlord's Interest In The Property (or any portion thereof) or Landlord's or any other owner's interest therein in which the lender receives or may receive any equity participation or other ownership in the

Property and/or Landlord's Interest In the Property (or any portion thereof), in Landlord's interest therein and/or in the building(s) built or to be built thereon and/or in rents, income and/or profits therefrom, (iii) any offer to sell or transfer the Property as part of a portfolio of any other one or more properties outside the Property owned by Landlord and/or affiliates of Landlord which Landlord intends to offer to the general market (a "Portfolio Sale"), and if a Portfolio Sale shall occur, Tenant's right of first offer to purchase Landlord's Interest in the Property under this Section 20 shall cease, terminate and be of no further force and effect. In no event shall Tenant's rights under this Section 20 be triggered by any transaction between Landlord and an affiliate of Landlord, or any change in or addition of any capital partners or investors of Landlord, from time to time. The parties hereto acknowledge and agree that the aforesaid right of first offer to purchase is personal to Lantheus Medical Imagining, Inc., and that in no event will any assignee, sublessee, or other successor have any rights under this Section 20, except pursuant to a "Permitted Transfer" under Section 6.2.1 of the Lease.

- 21. <u>Cleaning.</u> Tenant shall be responsible for keeping the Expansion Premises in clean condition. Tenant shall, at its sole expense, contract for professional cleaning services for the Expansion Premises on a regular basis in accordance with prevailing cleaning standards. Section 5.1.3 of the Lease shall have no applicability to the Expansion Premises.
- 22. <u>Signage</u>. In addition to the signage rights provided in Section 6.2.7 of the Lease, Tenant shall have the right to install, at its sole expense, its business identification sign on the façade of the North Building facing Burlington Road, subject to Landlord's approval of the specifications, locations and manner of installation (the "Building Sign"). Tenant's Building Sign shall comply with all applicable codes, ordinances and requirements of the Town of Bedford. Tenant shall maintain the Building Sign in good condition, at its sole expense, and shall remove it at the end of the Term. Landlord, at its sole expense, shall provide building standard signage in the North Building lobby.

23. Special Use Permits; Contingency.

(a) Tenant is required to obtain from the Town of Bedford one or more special use permits which are necessary to conduct Tenant's business operations and laboratory uses in the Expansion Premises (all such permits are collectively referred to as the "Special Use Permits"). It is agreed that the effectiveness of this Amendment is expressly conditioned upon Tenant obtaining the Special Use Permits (and only the Special Use Permits) on or before the date that is twelve (12) weeks after the Effective Date (such condition being referred to herein as the "Permitting Contingency" and such date being referred to as the "Permitting Contingency Date"). Promptly following the execution of this Amendment, Tenant, at its sole cost and expense, shall apply for and diligently take all necessary steps to obtain such Special Use Permits. The Special Use Permits shall be deemed obtained when the time for appeal of the issuance of the same has expired with no appeal pending and no such appeal having been instituted, or if any appeal had been instituted, with such appeal having been adjudicated in favor of the issuance of the Special Use Permits, and with no conditions placed upon the Special Use Permits except as are acceptable to Tenant in its good faith and commercially reasonable judgment. It is intended that the effectiveness of this Amendment shall not be dependent upon Tenant obtaining all other necessary governmental permits and approvals (such as a building permit for Tenant's Work) by the Permitting Contingency Date (but Tenant shall not be relieved of the obligation to actually obtain all necessary permits and approvals required by the applicable governmental authorities). Landlord shall assist and cooperate with Tenant in obtaining

all necessary permits and approvals as Tenant reasonably requests, at no cost to Landlord.

(b) If the Permitting Contingency has not been satisfied by the Permitting Contingency Date, then either Landlord or Tenant may elect to terminate this Amendment by giving written notice of termination to the other no later than five (5) days after the Permitting Contingency Date, effective as of the date such termination notice. However, the foregoing termination right shall not be available to Tenant if Tenant's actions or inactions, including not having used commercially reasonable efforts to obtain the Special Use Permits, are the sole reason the Permitting Contingency has not been satisfied, or (ii) to either party if either party is making significant progress in obtaining the Special Use Permits. As used herein, "significant progress" shall mean Landlord or Tenant (as applicable) has filed all required documentation, is attending all meetings and using commercially reasonable efforts to respond to follow-up questions and requests for information required by the Town in a timely manner. If either party timely elects to terminate, this Amendment shall be void ab initio and shall terminate as of the effective date of termination without the need for any further documentation. In no event shall the termination of this Amendment operate to terminate the Lease, it being agreed that the Lease shall continue in full force and effect in accordance with its terms immediately prior to the date of this Amendment, If neither Landlord nor Tenant elects to so terminate, the Permitting Contingency shall be deemed to have been waived by the parties without further action and this Amendment shall continue in full force and effect.

24. Hazardous Materials.

(a) Tenant covenants and agrees not to use, Release, dispose, manufacture, store, or transport any Hazardous Materials (hereinafter defined), and/or Radioactive Materials (hereinafter defined), at, on, under or from the Premises and the Property except for those Hazardous Materials and Radioactive Materials used in the ordinary course of Tenant's business [***]. Tenant covenants and agrees not to use, Release, dispose, manufacture, store, or transport any Hazardous Materials and/or Radioactive Materials at, on, under or from the Premises and the Property except in compliance with any and all laws, regulations, ordinances or orders promulgated, and as may be amended, by any local governmental authority, the Commonwealth of Massachusetts and/or the United States Government having jurisdiction over such Hazardous Materials and/or Radioactive Materials so brought upon or used or kept in or about the Property (collectively, "Legal Requirements"), and only in compliance with all such applicable Legal Requirements and any reasonable requirements of Landlord (such as zoning/permitting requirements for fencing or other locked enclosures). Tenant shall comply with all applicable governmental reporting requirements with respect to Hazardous Materials, Radioactive Materials, and all chemicals and flammable substances (in whatever form) used by Tenant in its business operations, and shall deliver to Landlord copies of all such reports. In the event that a Release or Threat of Release (hereinafter defined) of Hazardous Materials or Radioactive Materials occurs at, from or upon the Premises or Property during the term of this Lease for which Tenant is responsible, Tenant shall at its expense perform all actions required under any and all applicable Legal Requirements to assess, contain, remove or respond to such Release or Threat of Release; provided, however, that Tenant's work or actions hereunder shall be subject to Landlord's prior approval, which approval shall not be unreasonably withheld, conditioned or delayed. The term "Hazardous Materials" shall mean any and all materials defined or classified as "hazardous materials" "hazardous waste," "hazardous

substance," "toxic substance," "hazardous pollutant," "toxic pollutant" or "oil" or words of similar import under any applicable local, state or federal law and/or under the regulations adopted or publications promulgated pursuant thereto, including, but not limited to, 42 U.S.C. §9601 et. seq. (CERCLA), 42 U.S.C. §6901 et. seq. (RCRA), M.G.L. c. 21C or M.G.L. c.

21E, 310 CMR 40.0000 (MCP), or the Town of Bedford Bylaw Article 50, all as amended. The term "Radioactive Materials" shall mean any and all source, special nuclear, or byproducts of materials as defined by the Atomic Energy Act of 1954, and/or that are subject to regulation by the United States Nuclear Regulatory Commission (NRC) and/or Massachusetts Department of Public Health Radiation Control Program under 105 CMR 120.00. The term "Hazardous Materials" used under this Section 24 shall also mean and include "Radioactive Materials", unless the context requires otherwise. The terms "Threat of Release" and "Threats of Release" as used in this Amendment mean a substantial likelihood of a Release that requires action to prevent or mitigate damage to the environment or the Building that may result from such release. For purposes of this Section 24, a "Release" occurs when it meets or exceeds the Materiality Standards outlined in Exhibit C attached hereto.

Landlord and Tenant agree that Section 6.2.3 of the Lease, *Hazardous Wastes and Materials*, is hereby deleted in its entirety and that the terms and provisions of this Section 24 shall govern as of the Effective Date.

- (b) Landlord has a framework of Materiality Standards, set forth in <u>Exhibit C</u> attached hereto and incorporated herein, for determining certain spills, Releases or Threats of Release of Hazardous Materials or Radioactive Materials at the Premises, Building or Property requiring notification of Landlord. Tenant shall promptly notify Landlord in writing of all spills, Releases or Threat of Release of Hazardous Materials and/or Radioactive Materials at the Premises, the Building, or the Property which meet one or more of the Materiality Standards for notification of Landlord set forth in said <u>Exhibit C</u> caused by or involving Tenant or its business operations. Further, Tenant shall notify Landlord in writing of all notices, orders, fines or communications of any kind received by Tenant from any governmental authority or third party concerning the presence or suspected presence of Hazardous Materials on the Premises or the Property, the migration or suspected migration of Hazardous Materials from other property to other property, or the migration or suspected migration of Hazardous Materials from other property to the Premises or the Property.
- (c) Upon not less than fifteen (15) days advance written notice to Tenant, Landlord, its officers, employees, contractors and agents shall have the right, but not the duty, to cause its environmental LSP to inspect areas of the Premises to determine whether Tenant is complying with CERCLA, RCRA, Chapter 21C, Chapter 21E, the US NRC, 105 CMR 120.00, and other municipal, state and federal environmental laws, or regulations promulgated pursuant to any of the foregoing, as amended. Landlord shall use reasonable efforts to minimize interference with Tenant's business, including, but not limited to, conducting any inspection during regular business hours, or during other hours by agreement of the parties, but shall not be liable for any interference caused thereby, provided Landlord shall have used such reasonable efforts.
- (d) Tenant shall permit Landlord, its employees and its agents (including its environmental LSP), access to all areas of the Premises, from time to time during the Term, for the purposes of conducting an environmental assessment or inspection as set forth herein. Upon not less than fifteen (15) days advance written notice to Tenant, Landlord shall have the right to conduct an environmental assessment of the Premises as well as any other

areas in, on or about the Property that Landlord reasonably believes may have been affected adversely by Tenant's use of the Premises (collectively, the "Affected Areas") in order to confirm that the Premises and the Affected Areas do not contain any Hazardous Materials resulting from Tenant's use in violation of applicable Legal Requirements or under conditions constituting or likely to constitute a Release of Hazardous Materials. Such environmental assessment shall be such level of investigation necessary, as recommended by Landlord's environmental LSP. Such right to conduct such environmental assessment shall not be exercised more than once per calendar year, unless there is an environmental emergency or if Landlord reasonably believes that areas of the Premises or the Property may have been adversely affected by Tenant's use of the Premises. If contamination has occurred for which Tenant is liable for under this Lease, then Tenant shall pay all reasonable out-of-pocket costs incurred to conduct the assessment. If no such contamination is found, Landlord shall pay for the costs of the assessment. Landlord shall provide Tenant with a copy of all third-party reports and tests of the Premises and Affected Areas made by or on behalf of Landlord during the Term. In the event Landlord shall exercise its rights under this Section 24(d), Landlord shall use reasonable efforts to minimize interference with Tenant's business, including, but not limited to, conducting any assessment during regular business hours, or during other hours either by agreement of the parties or in the event of an environmental emergency, but shall not be liable for any interference caused thereby, provided Landlord shall have used such reasonable efforts.

- (e) Landlord shall have the right, from time to time, during the Term (provided that Landlord has a reasonable belief that Hazardous Materials are present on or at the Premises or the Property in violation of Legal Requirements and has delivered to Tenant evidence to support such belief), and upon the expiration of the Term of the Lease, to require that Tenant hire, and in such event, Tenant shall at its own expense hire, an environmental consultant reasonably satisfactory to Landlord to undertake an environmental assessment, inspection and/or sampling at the Premises and/or Property to determine whether Hazardous Materials have been Released during the Term of the Lease. Such right to require such environmental assessment shall not be exercised more than once per calendar year, unless there is an environmental emergency or if Landlord reasonably believes that areas of the Premises or the Property may have been adversely affected by Tenant's use of the Premises. If it is determined that no Release has occurred, Landlord shall pay for the costs of the assessment.
- (f) Should the assessment, inspection or sampling performed pursuant to Section 24(c) or Section 24(d) above, or any other assessment, inspection or sampling, reveal that there has been a Release or Threat of Release of Hazardous Materials by Tenant or its employees, agents or contractors, then Tenant shall,
 - (i) in the event of a Release, at its expense, undertake with an environmental LSP of its choice, all remediation and/or response action required by any governmental authority, or (ii) in the event of a Threat of Release, at its expense, undertake, with an environmental LSP of its choice, all remediation and/or response action reasonably required by Landlord but in no event in excess of any such response action that would be required by any applicable governmental authority and (iii) in either event, Tenant shall promptly thereafter restore any areas damaged or affected by such remediation and/or response action.
- (g) [***].
- (h) Tenant agrees to indemnify, defend with counsel reasonably acceptable to and approved by Landlord, and save Landlord harmless from all claims, liability,

loss, out-of-pocket costs or damage arising as a result of the use, Release, Threat of Release, holding, handling, transport, storage, or disposal of Hazardous Materials and/or Radioactive Materials by Tenant, its employees, agents, affiliates, assignees, subtenants, or contractors, or any party transporting Hazardous Materials and/or Radioactive Materials at, on, upon or from the Premises or Property from and after the Tenant Improvement Access Date or the date Tenant first occupied any portion of the Premises for lab use as set forth in Section 5 of this Amendment, including, without limitation, liability under any federal, state, or local laws, requirements and regulations, or damage to any of the systems of the Building or the Property, except to the extent caused by the negligence or misconduct of Landlord or any other occupants of the Building or other third parties not acting through or on behalf of Tenant. The provisions of this Section 24(h) shall survive the expiration or earlier termination of the Lease. Notwithstanding anything to the contrary contained in this subsection (h), Tenant shall not be responsible for, and the indemnification and hold harmless obligation set forth in this paragraph shall not apply to (i) contamination in the Premises which existed in the Premises prior to the Tenant Improvement Access Date or the date Tenant first occupied the Expansion Premises for lab use in accordance with Section 5 of this Amendment, or (ii) the presence of any Hazardous Materials and/or Radioactive Materials in the Premises which migrated from outside of the Premises into the Premises, unless and to the extent, in either case, the presence of such Hazardous Materials (x) is the result of a breach by Tenant of any of its obligations under this Lease, or (y) was caused, contributed to, in whole or in part, by Tenant or its agents, employees, affiliates, assignees, subtenants, or contractors.

(i) Tenant shall be solely responsible for conducting all facility decommissioning for the Expansion Premises. On or before the expiration or earlier termination of the Lease, Tenant shall surrender the Expansion Premises to Landlord in a condition deemed suitable for unrestricted use, as defined by the NRC, with all facility decommissioning work completed. Tenant's facility decommissioning work shall be deemed completed if Tenant complies, in all material respects with the American National Standards Institute's Laboratory Decommissioning guidelines (ANSI/AIHA Z9.11-2008) or any successor standards published by ANSI or any successor organization (or, if ANSI and its successors no longer exist, a similar entity publishing similar standards) and the Massachusetts Department of Public Health Regulation 105 CMR

120.00 (specifically regarding Radiological Criteria for License Termination (120.244-120.258), and the NRC Regulation Title 10 CFR (specifically Subpart E – Radiological Criteria for License Termination (collectively, the "Decommissioning Standards"). Landlord shall have the right (upon reasonable prior notice and during business hours) to have its environmental specialists, observe and inspect the progress of Tenant's decommissioning work. In addition, at least ten (10) days prior to Tenant's surrender of possession of any part of the Expansion Premises, Tenant shall (a) provide Landlord with written evidence of all appropriate governmental releases obtained by Tenant in accordance with applicable laws, (b) place Laboratory Equipment Decontamination Forms on all decommissioned equipment to assure safe occupancy by future users and (c) conduct a site inspection with Landlord. In addition, Tenant agrees to remain responsible after the surrender of the Expansion Premises for the remediation of any Hazardous Materials or Radioactive Materials, for which it is responsible, in, on, or at the Property due to a Release to the environment during the Term. Tenant's obligations under this Section 24 shall survive the expiration or earlier termination of the Lease.

Landlord represents and warrants to Tenant as of the Effective Date, that, except as identified in the Environmental Reports listed and defined in Exhibit H attached hereto and incorporated herein by reference, to Landlord's actual knowledge: (i) no Hazardous Materials or Radioactive Materials have been disposed of, or identified on, under, or at the Property in violation of applicable environmental laws; (ii) no written notice has been issued to Landlord from any government authorities, or any political or quasipolitical, subdivision, agency, authority, department, court, commission, board, bureau, or instrumentality of any of the foregoing asserting that the Property is or may be in violation of any applicable federal, state, or municipal law, ordinance, or regulation regarding Hazardous Materials; (iii) except as referenced in the Environmental Reports, no further Hazardous Materials were used in the construction of the Building; (iv) no Release of Hazardous Materials has occurred at, from, in, adjacent to, or on the Property, nor are there any Hazardous Materials migrating to the Property; (v) except as may be disclosed in the Environmental Reports, no incinerators, septic tanks, underground storage tanks, PCB-containing equipment, asbestoscontaining material, formaldehyde insulators, or cesspools are on the Property; (vi) all waste is discharged from the Property into a public sanitary sewer system in accordance with applicable legal requirements and no Hazardous Materials are discharged from the Property, directly or indirectly, into any body of water by Landlord.

Landlord has delivered to Tenant copies of all environmental reports, including, but not limited to, soil reports and groundwater reports, prepared by third parties that Landlord has in its possession. Tenant acknowledges receipt of the Environmental Reports.

Landlord agrees to indemnify, hold harmless, and defend Tenant from and against any and all claims, liabilities, losses, costs, damages, and expenses (including reasonable attorneys' fees, charges, and expenses in the enforcement of this indemnity) asserted against or suffered by Tenant arising out of, related to, or caused by the breach or inaccuracy of the representation and warranty made under this Section 24(j) by Landlord. The indemnification contained in this Section 24(j) shall survive the termination or earlier expiration of this Agreement or the Lease.

(k) Landlord shall indemnify, defend with counsel reasonably acceptable to Tenant and approved by Tenant, and save Tenant harmless from and against all claims, liability or loss, out of pocket costs or damage as a result of the existence of Hazardous Materials or Radioactive Materials in, at or upon the Expansion Premises prior to the Tenant Improvement Access Date or the date of Tenant's earlier occupancy of any portion of the Expansion Premises for lab use (whichever occurs first) provided, however, that this indemnity shall not apply to the extent Tenant or its agents, employees, affiliates, assignees, subtenants, or contractors causes or contributes to the Release or Threat of Release of Hazardous Materials and/or Radioactive Materials which are the subject of such claim, liability, loss and/or damage. With respect to claims against Tenant arising out of personal injury or damage suffered or alleged to have been suffered by Tenant's employees or former employees, subtenants or contractors relating to such pre-existing Hazardous Materials, Landlord shall have no duty to defend Tenant, however, Landlord shall indemnify Tenant (including legal costs incurred) in such cases only if such party's claim prevails, despite Tenant having used reasonable efforts to defend such claim, in a judicial proceeding in a court of competent jurisdiction through final judgement, with no further appeal. Landlord shall, at Landlord's sole cost and expense, be responsible for remediation of any Hazardous Material in, on, under or about the Expansion Premises prior to the Tenant Improvement

- Access Date or the date of Tenant's earlier occupancy of the Expansion Premises for lab purposes. The provisions of this Section 24(j) shall survive the expiration or earlier termination of the Lease.
- (l) It is a condition precedent to Tenant's obligation to perform under this Amendment that Tenant obtains an environmental insurance policy satisfactory to Tenant in its sole discretion. Landlord agrees to cooperate with Tenant to provide such information and documentation in its possession regarding the nature and extent of any Hazardous Materials at, on, under or from the Building and the Property as Tenant may reasonably request, and to allow such further testing as may be reasonably required and approved by Landlord (approval not to be unreasonably withheld) to enable Tenant to obtain the environmental insurance policy. If Tenant is unable to obtain a satisfactory policy prior to the Permitting Contingency Date, Tenant may elect, in its sole and absolute discretion, not to proceed with this Amendment by giving Landlord written notice of such election (the "Notice Not To Proceed") no later than ten (10) days after the Permitting Contingency Date. If Tenant gives Landlord the Notice Not To Proceed, this Amendment (but not the Lease) shall terminate, effective five (5) days after Landlord's receipt of such notice.

25. Rooftop Equipment.

Tenant shall have the right, at no charge to Tenant, to install and operate rooftop mechanical equipment to be used in connection with the Tenant's lab uses, to be installed in an area on the roof of the North Building to be designated for Tenant's use (such area is referred to as the "Rooftop Equipment Area"). Following the execution of this Amendment, Landlord and Tenant shall work together, each acting reasonably and in good faith, to agree on a rooftop plan which will (i) determine a mutually acceptable location for placement of the Rooftop Equipment (defined below) and (ii) reach an agreement as to the manner of installation of the Rooftop Equipment. While Landlord agrees to act reasonably and in good faith, Landlord shall not be deemed to be unreasonable if (x) it designates specific areas as suitable for placement of the Rooftop Equipment (it being understood that Landlord needs to take into account conditions for future rooftop installations), or (y) it places any condition(s) on any manner of equipment installation that would adversely affect the proper functioning of the Building systems. Tenant's rooftop equipment shall be subject to Landlord's approval, not to be unreasonably withheld, conditioned or delayed, regarding size and the manner of installation, including conformance with Landlord's reasonable design criteria and any other reasonable requirements. The approved rooftop equipment shall be referred to herein as the "Rooftop Equipment". Tenant shall ensure that installation of the Rooftop Equipment will not in any way affect the roof warranty or the structural integrity of the roof, and shall require its employees and contractors to exercise the standard of care required by the manufacturer of the roofing materials to prevent voiding roof warranties. In connection with the foregoing, Tenant shall submit to Landlord for its approval the plans and specifications for the Rooftop Equipment installation, which approval shall not be unreasonably withheld, conditioned or delayed, and shall comply with Landlord's requirements under Section 6.2.5 of the Lease, in all respects, provided, however, Landlord shall not be deemed unreasonable for conditioning its approval of any plans and specifications if the manner of installation will not meet, at a minimum, the same standard of care required by the manufacturer of the roofing materials to prevent voiding roof warranties. Tenant shall also have the right to use its proportionate share of common conduit and shaft space in the Building, if available, at no charge, to connect the Premises to the Rooftop Equipment. Tenant shall also pay

Landlord all costs of electricity in connection with the Rooftop Equipment, as reasonably determined by Landlord, such amounts to be included in Expansion Premises Tenant Electric payable by Tenant in accordance with Section 16 hereof and Section 4.2.5(b) of the Lease. Tenant agrees to maintain the Rooftop Equipment in good condition and repair throughout the Term. The installation, operation, maintenance, repair and removal of all Rooftop Equipment shall be at Tenant's sole cost and expense, utilizing contractors approved by Landlord, which may be Landlord's roofing contractor if required by Landlord, and shall be performed in accordance with all applicable laws and requirements of the Town of Bedford and any other governmental agencies and public authorities having jurisdiction.

- (b) Tenant's use of the Rooftop Equipment Area and the Rooftop Equipment shall not interfere with the use of the roof by other tenants (it being understood that the use of the Rooftop Equipment in the locations shown on a final rooftop plan approved by Landlord shall be deemed to not interfere). Tenant shall remove the Rooftop Equipment from the roof upon the expiration or earlier termination of the Lease, including capping off all shafts used in the operation of Tenant's mechanical equipment, and repair any damage to the roof and/or the Building caused by removal, unless at the time of review of Tenant's Plans Landlord advised Tenant that the Rooftop Equipment is not required to be removed at the end of the Term.
- (c) Tenant shall indemnify, defend and hold Landlord harmless from and against any and all liabilities, claims, damages, costs and expenses relating in any way to the Rooftop Equipment, including damage to the roof or the Building, unless caused by the negligence or willful misconduct of Landlord, its agents, employees or contractors, or other users of the roof. The foregoing indemnification shall survive the expiration or earlier termination of the Lease.
- (d) The right to install the Rooftop Equipment is personal to the initial named tenant, Lantheus Medical Imaging, Inc. and may not be assigned or subleased except in connection with a Permitted Transfer under Section 6.2.1(d) of the Lease (or any assignment of the Lease approved by Landlord pursuant to Section 6.2.1 of the Lease). Once installed, the Rooftop Equipment may be assigned or subleased only in connection with an assignment of the Lease or a subletting of the Premises. In that event, Tenant shall not be relieved of the primary responsibility for removal of any Rooftop Equipment under the terms of Section 25(b) above. Tenant shall be prohibited from granting any other party any right or license to use any portion of the Rooftop Equipment Area apart from the Lease.
- 26. <u>Subordination</u>. As a condition to the effectiveness of this Amendment, Landlord shall cause its current lender to execute and deliver to Tenant a Subordination, Attornment and Non-Disturbance Agreement in the form attached hereto as <u>Exhibit G</u> (the "SNDA"). If the SNDA is not delivered to Tenant within five (5) business days following full execution of this Amendment and execution of the SNDA by Tenant and Landlord, Tenant shall have the right to terminate this Amendment, <u>but not the Lease</u>, by giving written notice of termination to Landlord, effective upon Landlord's receipt of such notice, without liability or obligation on the part of either Landlord or Tenant.
- 27. <u>Lease not to be Recorded; Notice of Lease</u>. Tenant agrees it will not record the Lease or this Amendment. Contemporaneously with the execution of the Amendment, (a) Landlord agrees to execute, acknowledge and deliver a Notice of Lease in the form attached hereto as <u>Exhibit E</u>, which is to be held in escrow by Tenant until the Permitting Contingency has been satisfied or waived in writing at which point Tenant may record the same with the

Middlesex South Registry District of the Land Court, and (b) Tenant agrees to execute, acknowledge and deliver a Notice of Termination of Lease in Tenant's name in the form attached hereto as <u>Exhibit F</u>, which is to be held in escrow by Landlord until the expiration or earlier termination of the Lease.

- 28. Expansion Premises Yield-Up. Notwithstanding anything to the contrary contained in Section 6.1.9 of the Lease, Tenant shall have no obligation at the end of the Term to remove Landlord's Expansion Premises Work or Tenant's Work made in accordance with the Final Plans, except that Tenant shall remove the Rooftop Equipment (unless at the time of review of Tenant's Plans, Landlord advised Tenant that any Rooftop Equipment and/or ground-mounted is not required to be removed at the end of the Term). Any ground-mounted equipment approved by Landlord, such as a generator and associated fuel tank, shall be removed from the Property at the end of the Term (and any damage caused by removal shall be repaired by Tenant), unless Landlord informs Tenant at the time Landlord reviewed the plans therefor, that such equipment is not required to be removed at the end of the Term. However, Tenant shall be required to cap off all shafts and ductwork in the Expansion Premises. In addition to compliance with the yield-up requirements in Section 6.1.9 of the Lease, Tenant shall be obligated to perform all facility decommissioning work and surrender the Expansion Premises in accordance with the requirements set forth in Section 24(i) of this Amendment. Tenant's obligations under this Section 28 shall survive the expiration or earlier termination of the Lease. In no event shall this Section 28 be construed to diminish or relieve Tenant of its obligations to comply in all respects with the Decommissioning Standards.
- 29. <u>Guaranty.</u> Tenant's obligations under the Lease are secured by a Guaranty dated February 14, 2022 executed by Lantheus Holdings, Inc. ("Guarantor"). Guarantor joins in the execution of this Amendment for the purpose of confirming Guarantor's consent to the terms, agreements and obligations of Tenant set forth herein.
- 30. <u>Authority.</u> Landlord and Tenant each represent, as to itself, (a) that it is validly existing and in good standing in the state where it was organized, and in good standing in the Commonwealth of Massachusetts; (b) that it has the authority and capacity to enter into this Amendment and perform all of its obligations hereunder; (c) that all necessary action has been taken in order to authorize it to enter into and perform all of its obligations hereunder; (d) that the person executing this Amendment on its behalf is duly authorized to do so.
- 31. <u>Entire Agreement</u>. This Amendment contains the entire agreement of the parties regarding the subject matter hereof. There are no promises, agreements, conditions, undertakings, warranties or representations, oral or written, express or implied, among them, relating to this subject matter, other than as set forth herein. This Amendment shall be construed under the laws of the Commonwealth of Massachusetts and shall be binding upon and shall inure to the benefit of the parties hereto and their respective heirs, successors and assigns.
- 32. <u>Brokers</u>. Tenant warrants and represents that it has dealt with no broker in connection with the consummation of this Lease, other than [***] (the "Brokers"), and in the event of any brokerage claims, other than by the Brokers against Landlord predicated upon prior dealings with Tenant, Tenant agrees to defend the same and indemnify and hold Landlord harmless against any such claim.
- 33. <u>Execution</u>. This Amendment shall not be valid and binding until executed and delivered by Landlord, and may be executed in multiple counterparts, each of which shall be deemed an original and all of which, when taken together, shall constitute one and the same instrument. Any facsimile, PDF, or other electronic transmittal of original signature versions of this Amendment shall be considered to have the same legal effect as execution and delivery of the original document and shall be treated in all manner and respects as the original document. Execution of this Amendment by means of DocuSign is an acceptable form of execution, valid and binding and having the same legal effect as execution with wet ink signatures and shall be treated in all manner and respects as the original document.

Signatures appear on the following page

WITNESS the execution under seal this 4th day of May, 2023.

LANDLORD:

201 BURLINGTON ROAD OWNER, LLC a Delaware limited liability company

By: 201 Burlington Road Venture, LLC, a Delaware limited liability company, its sole member

By: 201 Burlington Road NDC Promote LLC, a Massachusetts limited liability company, its Manager

By: Nordblom JV Manager, Inc., a Massachusetts corporation, its manager

By: <u>/s/ Todd Nordblom</u> Todd Nordblom President

By: <u>/s/ Crosby Nordblom</u> Crosby Nordblom Senior Vice President

TENANT:

LANTHEUS MEDICAL IMAGING, INC.

By:/s/ Mary Anne Heino Chief Executive Officer

GUARANTOR:

LANTHEUS HOLDINGS, INC.

By: <u>/s/ Mary Anne Heino</u> Mary Anne Heino Chief Executive Officer Hereunto duly authorized

EXHIBIT A PLAN SHOWING THE EXPANSION PREMISES

[***]

EXHIBIT A-1

PRELIMINARY LAYOUT PLAN

EXHIBIT B

IMPROVEMENTS

EXHIBIT B-1

Tenant Landlord Matrix of Responsibility

EXHIBIT B-2 PRELIMINARY CONSTRUCTION SCHEDULE

EXHIBIT C

MATERIALITY STANDARDS

[***]

EXHIBIT D CONVERSION TO NET

FIXED RENT

<u>[***]</u>

EXHIBIT E

FORM NOTICE OF LEASE

Pursuant to General Laws of Massachusetts, Chapter 183, Section 4, notice is hereby given of the Lease described below:

Date of Lease Execution:

 $Landlord: \hspace{0.5cm} 201 \hspace{0.1cm} BURLINGTON \hspace{0.1cm} ROAD \hspace{0.1cm} OWNER, \hspace{0.1cm} LLC, \hspace{0.1cm} a \hspace{0.1cm} Delaware \hspace{0.1cm} limited \hspace{0.1cm} partnership, \hspace{0.1cm} having \hspace{0.1cm} an \hspace{0.1cm} address \hspace{0.1cm} of \hspace{0.1cm} c/o \hspace{0.1cm} a \hspace{0.1cm} partnership, \hspace{0.1cm} having \hspace{0.1cm} an \hspace{0.1cm} address \hspace{0.1cm} of \hspace{0.1cm} c/o \hspace{0.1cm} a \hspace{0.1cm} partnership, \hspace{0.1cm} having \hspace{0.1cm} an \hspace{0.1cm} address \hspace{0.1cm} of \hspace{0.1cm} c/o \hspace{0.1cm} a \hspace{0.1cm} partnership, \hspace{0.1cm} having \hspace{0.1cm} an \hspace{0.1cm} address \hspace{0.1cm} of \hspace{0.1cm} c/o \hspace{0.1cm} a \hspace{0.1cm} partnership, \hspace{0.1cm} having \hspace{0.1cm} an \hspace{0.1cm} address \hspace{0.1cm} of \hspace{0.1cm} c/o \hspace{0.1cm} a \hspace{0.1cm} partnership, \hspace{0.1cm} having \hspace{0.1cm} an \hspace{0.1cm} address \hspace{0.1cm} of \hspace{0.1cm} c/o \hspace{0.1cm} a \hspace{0.1cm} partnership, \hspace{0.1cm} having \hspace{0.1cm} an \hspace{0.1cm} address \hspace{0.1cm} of \hspace{0.1cm} c/o \hspace{0.1cm} a \hspace{0.1cm} partnership, \hspace{0.1cm} having \hspace{0.1cm} an \hspace{0.1cm} address \hspace$

ALAAC, LLC, 71 Third Avenue, Burlington, MA 01803

Tenant: LANTHEUS MEDICAL IMAGING, INC., a Delaware

corporation, having an address of 201 Burlington Road, Bedford, MA 01730

Premises: Approximately 46,526 rentable square feet of floor area on the 1st, 2nd, and 3rd floors in the south portion

of the building located on the parcel of land described in <u>Exhibit A</u> attached hereto (such parcel of land, the "Property") and commonly known as 201 Burlington Road, Bedford, Massachusetts (the "Building"), and approximately 41,655 rentable square feet on the

first floor of the north portion of the Building.

Term of the Lease: Fifteen (15) years and four (4) months as provided in the

Lease.

Rights to Extend: Tenant has two (2) options to extend the term of the Lease

for additional periods of five (5) years each, as further described in Section 2.3 of the

Lease.

Right of First Offer

to Lease: Tenant has a right of first offer to lease on all or any tenant space in the portion of the Building known as

the "North Building", subject to rights of existing tenants to lease any such space, as further described in Section 2.4 of the Lease and Section 19 of the First Amendment to

Lease.

Right of First Offer

To Purchase: Tenant has a right of first offer to purchase all or a portion

of Landlord's interest in the Building and the Property as provided in Section 20 of the

First Amendment to Lease.

This Notice of Lease has been executed solely to give record notice of the Lease, and all of the terms, conditions and covenants thereof, which are incorporated herein by reference. In the event of any discrepancy between the provisions of the Lease and this Notice of Lease the provisions of the Lease shall take precedence and prevail over the provisions of this. Notice of Lease. This Notice of Lease is not intended to modify or amend the terms, conditions and covenants of the Lease, nor shall it be deemed to vary or govern interpretation thereof.



WITNESS the execution hereof under seal as of the Effective Date:

LANDLORD:

201 BURLINGTON ROAD OWNER, L	LC
a Delaware limited liability company	

By: 201 Burlington Road Venture, LLC, a Delaware limited liability company, its sole member

By: 201 Burlington Road NDC Promote LLC, a Massachusetts limited liability company, its Manager

By: Nordblom JV Manager, Inc., a Massachusetts corporation, its Manager

COMMONWEALTH OF MASSACHUSETTS)) 600
COUNTY OF)) ss:

On this day of _____, 2023, before me, the undersigned notary public, personally appeared ____, proved to me through satisfactory evidence of identification, being (check whichever applies): \Box driver's license or other state or federal governmental document bearing a photographic image, \Box oath or affirmation of a credible witness known to me who knows the above signatory, or \Box my own personal knowledge of the identity of the signatory, to be the person whose name is signed on the preceding or attached document, and acknowledged to me that he/she signed it voluntarily in his/her capacity as the ___ of Nordblom JV Manager, Inc. for its stated purpose, as the voluntary act of said corporation.

Notary Public:
My commission expires:

[Affix Notarial Seal]

TENANT:	
LANTHEUS MEDICAL IMAGING, INC.	
By: Name: Title:	
COMMONWEALTH OF MASSACHUSETTS) COUNTY OF)) ss:
satisfactory evidence of identification, being (check vgovernmental document bearing a photographic image, \square the above signatory, or \square my own personal knowledge of	signed notary public, personally appeared, proved to me through whichever applies): driver's license or other state or federal oath or affirmation of a credible witness known to me who knows the identity of the signatory, to be the person whose name is signed to me that he/she signed it voluntarily in his/her capacity as the is the voluntary act of said corporation.
	y Public: ommission expires:

EXHIBIT A

LEGAL DESCRIPTION OF THE PROPERTY

Three contiguous parcels of registered land in Bedford, Massachusetts, shown as Lots 37, 21A and 38 on Land Court Plan 29283G, respectively bounded and described as follows:

LOT 37 NORTHEASTERLY
EASTERLY NORTHEASTERLY
SOUTHEASTERLY
NORTHEASTERLY
EASTERLY and SOUTHEASTERLY
SOUTHWESTERLY
WESTERLY
NORTHERLY

LOT 21A SOUTHERLY

SOUTHWESTERLY NORTHWESTERLY NORTHERLY EASTERLY

LOT 38

WESTERLY

SOUTHWESTERLY NORTHWESTERLY and

NORTHERLY EASTERLY

By Burlington Road, 282.27 feet; 410.02 feet; 109.81 feet, by Lot 28 as shown on said plan; By Lot 21A on said plan, about 35 feet; By the former location of **Vie** Brook; By Lot 38 on said plan, about 671 feet; 383.95 feet 852.37 feet; and 199.44 feet, by the Northeasterly, Easterly and Southerly lines of the State Highway (Route 3), non-access.

By Lot 15 as shown on plan hereinafter mentioned 213.97 feet; By the former location of Vine Brook; By Lot 37 on said plan, about 35 feet; By Lot 28 on said plan, 344.35 feet; and By Lot 22 on said plan, 20 feet.

By the Northeasterly line of the State Highway (Route 3), non-access, 195 feet; By Lot 37 as shown on said plan hereinafter mentioned, about 671 feet; By the former location of Vine Brook; and By land now or formerly of Warrenton Corporation, 762 feet.

All said boundaries are determined by the Court to be located as shown on a subdivision plan, as approved by the Court, filed in the Land Registration Office, a copy of which is filed in the Registry of Deeds for the South Registry District of Middlesex County in Registration Book 742, Page 69, with Certificate 122419.

EXHIBIT F

FORM NOTICE OF TERMINATION OF LEASE

eference is made to a certain Notice of Lease dated as of, 2023, recorded with Middlesex South Registry District of the and Court as Document No (the "Notice of Lease"), pursuant to which Landlord and Tenant gave notice of a certain case by and between 201 BURLINGTON ROAD OWNER, LLC ("Landlord") and LANTHEUS MEDICAL MAGING, INC. ("Tenant"), dated February 14, 2022 (as amended, the "Lease") of a certain premises located at 201 urlington Road, Bedford, Massachusetts, as more particularly described in the Notice of Lease (the "Premises").
Landlord and Tenant hereby give notice that the Lease and the Notice of Lease are no longer in force or effect.
This instrument is executed as of, 2023 as a notice of termination of the aforesaid Lease and is not intended, not nall it be deemed to vary or govern the interpretation of the terms and conditions thereof.
[REMAINDER OF PAGE LEFT INTENTIONALLY BLANK]

LANDLORD:

201 BURLINGTON ROAD OWNER, LLC a Delaware limited liability company

By: 201 Burlington Road Venture, LLC, a Delaware limited liability company, its sole member

By: 201 Burlington Road NDC Promote LLC, a Massachusetts limited liability company, its Manager

By: Nordblom JV Manager, Inc., a Massachusetts corporation, its Manager $\,$

By: ____ Name:
Title:

COMMONWEALTH OF MASSACHUSETTS)

COUNTY OF ___) ss:

On this day of _____, 2023, before me, the undersigned notary public, personally appeared ____, proved to me through satisfactory evidence of identification, being (check whichever applies): \Box driver's license or other state or federal governmental document bearing a photographic image, \Box oath or affirmation of a credible witness known to me who knows the above signatory, or \Box my own personal knowledge of the identity of the signatory, to be the person whose name is signed on the preceding or attached document, and acknowledged to me that he/she signed it voluntarily in his/her capacity as the ___ of Nordblom JV Manager, Inc. for its stated purpose, as the voluntary act of said corporation.

Notary Public: My commission expires:

[Affix Notarial Seal]

TENANT:
LANTHEUS MEDICAL IMAGING, INC.
By: Name: Title:
COMMONWEALTH OF MASSACHUSETTS)) ss:
COUNTY OF)
On this day of, 2023, before me, the undersigned notary public, personally appeared, proved to me through satisfactory evidence of identification, being (check whichever applies): driver's license or other state or federal governmental document bearing a photographic image, oath or affirmation of a credible witness known to me who knows the above signatory, or my own personal knowledge of the identity of the signatory, to be the person whose name is signed on the preceding or attached document, and acknowledged to me that he/she signed it voluntarily in his/her capacity as the of Lantheus Medical Imaging, Inc. for its stated purpose, as the voluntary act of said corporation.
Notary Public: My commission expires: [Affix Notarial Seal]

EXHIBIT G

LEASE SUBORDINATION, ATTORNMENT AND $\underline{\text{NON-DISTURBANCE}}$ $\underline{\text{AGREEMENT}}$

[***]

EXHIBIT H LIST OF ENVIRONMENTAL REPORTS

<u>[***]</u>

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Mary Anne Heino, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Lantheus Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 3, 2023

/s/ MARY ANNE HEINO

Name: Mary Anne Heino
Title: Chief Executive Officer
(Principal Executive Officer)

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Robert J. Marshall, Jr., certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Lantheus Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 3, 2023

/s/ ROBERT J. MARSHALL, JR.

Name: Robert J. Marshall, Jr.

Title: Chief Financial Officer and Treasurer

(Principal Financial Officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

Pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, Mary Anne Heino, the Chief Executive Officer, and Robert J. Marshall, Jr., the Chief Financial Officer, of Lantheus Holdings, Inc. (the "Company"), hereby certify, that, to their knowledge:

- 1. The Quarterly Report on Form 10-Q for the period ended June 30, 2023 (the "Report") of the Company fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 3, 2023

/s/ MARY ANNE HEINO

Name: Mary Anne Heino
Title: Chief Executive Officer

(Discised Energy of College)

(Principal Executive Officer)

Date: August 3, 2023

/s/ ROBERT J. MARSHALL, JR.

Name: Robert J. Marshall, Jr.

Title: Chief Financial Officer and Treasurer

(Principal Financial Officer)

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.